

Cornescu, Andrei Cătălin (2019) *High efficiency and high frequency resonant tunneling diode sources*. PhD thesis.

http://theses.gla.ac.uk/76766/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

# High Efficiency and High Frequency Resonant Tunneling Diode Sources



## Andrei Cătălin Cornescu

School of Engineering

University of Glasgow

A thesis submitted in fulfillment of the requirements for the degree of

**Doctor of Philosophy** 

December 2019

© Andrei Cătălin Cornescu, 2019



## **University of Glasgow**

School of Engineering

### Statement of Originality to Accompany Thesis Submission

Name: Andrei Catalin Cornescu

### **Registration Number:**

I certify that the thesis presented here for examination for a PhD degree of the University of Glasgow is solely my own work other than where I have clearly indicated that it is the work of others (in which case the extent of any work carried out jointly by me and any other person is clearly identified in it) and that the thesis has not been edited by a third party beyond what is permitted by the University's PGR Code of Practice.

The copyright of this thesis rests with the author. No quotation from it is permitted without full acknowledgement.

I declare that the thesis does not include work forming part of a thesis presented successfully for another degree.

I declare that this thesis has been produced in accordance with the University of Glasgow's Code of Good Practice in Research.

I acknowledge that if any issues are raised regarding good research practice based on review of the thesis, the examination may be postponed pending the outcome of any investigation of the issues.

### Statement of conjoint work

I confirm that Chapter 6 was jointly authored with Razvan Morariu and I contributed 50% of this work.

Signature: .....

Date: .....

## Acknowledgements

I would like to express gratitude to my supervisor, Dr. Edward Wasige, for his unconditional support and guidance through my years as an undergraduate and during the course of this PhD program. His patience, support and valuable advice, both in matters of research and life throughout my study program made it a valuable and enjoyable experience.

I would like to thank Dr. Jue Wang, for his support during my first years in the cleanroom and for the patience and knowledge shared with regards to design, fabrication and measurement of resonant tunneling diodes. I would like to extend my appreciation and gratefulness to the James Watt Nanofabrication Centre (JWNC) staff for their help in achieving a successful fabrication process.

I would also like to thank my colleagues: Răzvan Morariu and Dr. Afesomeh Ofiare for all their help, both in matters of work and life, which were invaluable and helped greatly towards the completion of my research. I gratefully acknowledge financial support from the Engineering and Physical Sciences Research Council (EPSRC) for my studentship to undertake research into terahertz technology.

Finally, special thanks go to my parents Andrieana and Traian Cornescu, family and friends for their consistent support.

### Abstract

Terahertz (THz) technology has been generating a lot of interest due to the numerous potential applications for systems working in this previously unexplored frequency range. THz radiation has unique properties suited for high capacity communication systems and non-invasive, non-ionizing properties that when coupled with a fairly good spatial resolution are unparalleled in its sensing capabilities for use in biomedical, industrial and security fields. However, in order to achieve this potential, effective and efficient ways of generating THz radiation are required. Devices which exhibit negative differential resistance (NDR) in their current-voltage (I – V) characteristics can be used for the generation of these radio frequency (RF) signals. Among them, the resonant tunneling diode (RTD) is considered to be one of the most promising solid-state sources for millimeter and submillimeter wave radiation, which can operate at room temperature.

However, the main limitations of RTD oscillators are producing high output power and increasing the DC-to-RF conversion efficiency. Although oscillation frequencies of up to 1.98 THz have been already reported, the output power is in the range of micro-Watts and conversion efficiencies are under 1 %. This thesis describes the systematic work done on the design, fabrication, and characterization of RTD-based oscillators in monolithic microwave/millimeter-wave integrated circuits (MMIC) that can produce high output power and have a high conversion efficiency at the same time.

At the device level, parasitic oscillations caused by the biasing line inductance when the diode is biased in the NDR region prevents accurate characterization and compromises the maximum RF power output. In order to stabilise the NDR devices, a common method is the use of a suitable resistor connected across the device, to make the differential resistance in the NDR region positive. However, this approach severely hinders the diode's performance in terms of DC-to-RF conversion efficiency. In this work, a new DC bias decoupling circuit topology has been developed to enable accurate, direct measurements of the device's NDR characteristic and when implemented in an oscillator design provides over a 10-fold improvement in DC-to-RF conversion efficiency. The proposed method can be adapted for higher frequency and higher power devices and could have a major impact with regards to the adoption of RTD technology, especially for portable devices where power consumption must be taken into consideration.

RF and DC characterization of the device were used in the realization on an accurate large-signal model of the RTD. S-parameter measurements were used to determine an accurate small-signal model for the device's capacitance and inductance, while the extracted DC characteristics where used to replicate the *I-V* characteristics. The model is able to replicate the non-stable behavior of RTD devices when biased in the NDR region and the RF characteristics seen in oscillator circuits. It is expected that the developed model will serve in future optimization processes of RTD devices in millimeter and submillimeter wave applications.

Finally, a wireless data transmission link operating in the Ka-band (26.5 GHz - 40 GHz) using two RTDs operating as a transmitter and receiver is presented in this thesis. Wireless error-free data transfer of up to 2 gigabits per second (Gbit/s) was achieved at a transmission distance of 15 cm.

In summary, this work makes important contributions to the accurate characterization, and modeling of RTDs and demonstrates the feasibility of this technology for use in future portable wireless communication systems and imaging setups.

## **Publications**

- [1] Cornescu, A. C., Morariu, R., Ofiare, A., Al-Khalidi, A., Wang, J., Figueiredo, J. M. L. and Wasige, E., "High efficiency bias stabilisation for resonant tunneling diode oscillators," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 8, pp. 3449-3454, Aug. 2019.
- [2] Cornescu, A. C., Wang, J., Al-Khalidi, A., Morariu, R. and Wasige, E., "I-V characteristics of a stabilized resonant tunneling diode," *Compound Semiconductor Week (CSW) 2017*, Berlin, Germany, 14 – 18 May 2017.
- [3] Morariu, R., Wang, J., Cornescu, A. C., Al-Khalidi, A., Ofiare, A., Figueiredo, J. M. L. and Wasige, E., "Accurate small-signal equivalent circuit modeling of resonant tunneling diodes to 110 GHz," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 11, pp. 4332-4340, Nov. 2019.
- [4] Wang, J., Al-Khalidi, A., Cornescu, A.C., Morariu, R., Ofiare, A. and Wasige, E., "Design, fabrication and characterization of RTD terahertz oscillators," in *European Microwave Conference in Central Europe* (*EuMCE*), Prague, Czech Republic, 2019.
- [5] Wang, J., Al-Khalidi, A., Zhang, C., Cornescu, A. C., Morariu, R., and Wasige, E., "Resonant tunneling diode oscillator source for terahertz applications," *Oxford Circuits and Systems Conference (OXCAS 2017)*, Oxford, UK, 19 Sep 2017.

[6] Wang, J., Al-Khalidi, A., Cornescu, A. C., Morariu, R., Khalid, A. -u. -H., Cumming, D. and Wasige, E., "Loading effect of W-band resonant tunneling diode oscillator by using load-pull measurement, " *Compound Semiconductor Week (CSW) 2017*, Berlin, Germany, 14 – 18 May 2017.

## **Table of Contents**

List of Figures	xi
List of Tables	XX
List of Symbols, Acronyms and Constantsx	xi
Chapter 1. Millimetre and Terahertz Wave Radiation	.1
1.1 Introduction	. 1
1.2 THz Applications	.4
1.3 THz Sources	. 8
1.3.1 Electronic Sources	.9
1.3.2 Photonic Sources	12
1.4 The Resonant Tunneling Diode	14
1.4.1 Historical Perspective	14
1.4.2 Operational Principle of RTD	16
1.4.3 RTD Material Systems	18
1.4.4 State of the Art	20
1.5 Thesis Aim and Organization	25
References	27
Chapter 2. RTD MMIC Fabrication Technology	37
2.1 Introduction	37
2.2 RTD Device Fabrication Flow	38
2.2.1 Layer Structure	38
2.2.2 Sample Preparation	39
2.2.3 Fabrication Process Flow	40
2.3 Fabrication Processes	42
2.3.1 Lithography	42
2.3.2 Metallisation	47
2.3.3 Dielectric Layer Deposition	
	47
2.3.4 Etching	
2.3.4 Etching	48
C C	48 52
2.4 Summary	48 52 53
2.4 Summary	48 52 53 <b>55</b>

	3.2.1Thin-film and Semiconductor Resistors	55
	3.2.2 Metal-Insulator-Metal Capacitors	57
	3.2.3 Coplanar Waveguide	58
3.3	Characterization of Passive Components	60
	3.3.1 Coplanar Waveguide	61
	3.3.2 MIM Capacitor	
	3.3.3 Thin-film and Semiconductor Resistors	65
3.4	Ohmic Contacts on InGaAs	66
3.5	Summary	70
Ref	erences	71
Chapter 4.	Resonant Tunneling Diode Bias Circuit Instabilities	74
4.1	Introduction	74
4.2	RTD Layer Structure	75
4.3	Resonant Tunneling Diode Models	78
	4.3.1 RC Circuit Model	78
	4.3.2 RLC Circuit Model	79
4.4	Bias Circuit Instabilities	
	4.4.1 Shunt Resistor Stabilisation	
	4.4.2 Shunt Capacitor and Resistor Stabilisation	
4.5	Measurement Results and Simulations	
	4.5.1 DC Measurements	86
	4.5.2 RF Measurements	
	4.5.3 Simulation Results	91
	4.5.4 Experimental Results	
4.6	Summary	
Ref	erences	96
Chapter 5.	Resonant Tunneling Diode Based Oscillators	
5.1	Introduction	
5.2	Characteristics of RTD oscillator	
	5.2.1 Maximum RF Power	
	5.2.2 Effect of Parasitics on Maximum RF Power	
	5.2.3 Oscillation and Cut-off Frequency	
	5.2.4 Phase Noise	110
	5.2.5 Figure of Merit for VCOs	

5.3 RTD Oscillator Topologies	
5.3.1 Single RTD Oscillators	
5.3.2 High-Efficiency RTD Oscillators	
5.3.3 Discussion	
5.4 Summary	
References	
Chapter 6. Wireless Data Transmission Link	
6.1 Introduction	
6.2 Experimental Setup	
6.2.1 RTD Transmitter	
6.2.2 RTD Receiver	
6.3 Measurement Results	
6.4 Summary	
References	
Chapter 7. Conclusions and Future Work	
7.1 Conclusions	
7.2 Future Work	140
7.2.1 RTD Modulation Bandwidth	
References	
Appendix A. Fabrication Process	145

## **List of Figures**

1.1	Electromagnetic spectrum showing the millimetre and sub-millimetre radiation location between microwaves and infrared spectrum
1.2	Atmospheric attenuation and expected data transfer rate of different propagation frequencies
1.3	Semiconductor devices showing the RF power output decreasing towards the THz-gap
1.4	Electromagnetic spectrum from microwave to the end of the THz region, showing the allocations for 5G, 6G and spectrum unification for wireless data centres
1.5	Hybrid wired / wireless data centre with wireless nodes connection between the server racks
1.6	Transmission image of a tooth showing the contrast at several frequencies in the mm-wave and terahertz range
1.7	Passive full body millimetre-wave security screening portal developed by Trex Enterprises working between 75.5 GHz to 93.5 GHz7
1.8	Image fusion from Safe VISITOR project. (a) visible image, (b) THz image at 350 GHz, (c) Infrared image, (d) final fused image
1.9	Operating principle of Gunn diodes (a) Gunn diode structure, (b) conduction and valence band profile of Gunn diode
1.10	Dual cavity power combining circuit using Gunn diodes mounted on diamond heatsinks
1.11	IMPATT diode structure. When a sufficiently high negative voltage is applied, avalanche breakdown occurs in the p+- n region resulting in a large number of carriers. The phase lag is controlled by the width of the drift region 11
1.12	Photoconductive antenna operating principle. A femtosecond laser is focused at the gap between the two electrodes generating free carriers, which are then accelerated by the bias voltage across the gap

#### 

### 

- 1.19 RTD oscillator integrated with Vivaldi antenna operating at 1.1 THz...... 23

- 2.2 RTD device fabrication steps: (a) top contact metal deposition. (b) wet etch to the bottom emitter/collector layer. (c). bottom contact metal deposition. (d) wet

	etch to substrate. (e) PI-2545 deposition and via opening using dry etch. (f) bond pad metal deposition
2.3	Illustration of the photolithography process. The substrate is first coated with a thin layer of positive photoresist. Using the Karl Suss MA6 mask aligner, the photomask is placed in contact with the sample. After exposure the sample is developed, whereby the resist that was not shielded by the photomask is removed leaving an accurate reproduction of the mask pattern
2.4	S1800 series photoresist spin speed vs thickness
2.5	Illustration of the lift-off process using toluene. (a) clean, spin and pre-bake the photoresist (b) dip in toluene for about 1-minute (c) expose and develop the photoresist, (d) metallization, and (e) metal lift-off
2.6	Illustration of the bilayer lift-off process (a) clean, spin and pre-bake the LOR layer (b) spin and deposit S1800 resist (c) expose and develop the photoresist, (d) metallization, and (e) lift-off
2.7	Etching process. The substrate is first coated with a layer of photoresist which is used as an etch mask to protect the desired areas. A wet or dry etching is carried out. The resist is then removed
2.8	SEM picture of an RTD device showing the undercut profile caused by wet etching. The effective area of the RTD is reduced due to lateral etching 50
2.9	Interferogram obtained during the dry etch process of the polyimide layer. The etch duration is approximately 5 min and 30 seconds, total etch depth is $1.2 \ \mu m$ for the polyimide layer
2.10	SEM picture of an RTD device showing the polyamide cushion that isolates the top emitter and bottom collector metals and prevents the metal from breaking
3.1	NiCr thin-film resistor design test structure: (a) top view, (b) cross section view
3.2	Metal-insulator-metal capacitor design: (a) top view, (b) cross section view

3.3 Coplanar waveguide structure on a substrate with dielectric constant εr an thickness h. The centre conductor line width is denoted as w, the width of the ground planes is g, the gap between the signal line and ground plane is s, the	ne
thickness of the conductor is t and the lengths are denoted with l. (a) top view (b) cross section view	
3.4 Schematic of a coplanar waveguide stub: (a) simple short circuit stub, (b) MIM-short stub 3D model	
<ul> <li>3.5 Micrograph of CPW test structure. The centre strip width is 60 μm, the ga width is 40 μm, while the ground plane width is 200 μm. The CPW test structure length is 500 μm.</li> </ul>	st
<ul> <li>3.6 Measured S-parameters (S<sub>11</sub>, S<sub>22</sub>) of the CPW line test structure. The return loss indicates that most of the power transmitted is not reflected back to the same port</li></ul>	ne
3.7 Measured S-parameters $(S_{12}, S_{21})$ of the CPW line test structure. The insertion loss indicates that most of the power is transmitted to the second port	
3.8 Measured S-parameter $(S_{11}, S_{12})$ of the parallel capacitor test structure 6	53
3.9 Measured S-parameter $(S_{11}, S_{22})$ of the series capacitor test structure	<b>j</b> 4
3.10 Measured S-parameters (S11) of the series and parallel capacitor test structure plotted on a Smith chart	
3.11 SEM picture of a fabricated NiCr resistor test structure. Two resistors and placed in parallel to achieve the designed value	
3.12 LTLM structure with various separation distances. Four probes are placed of two adjacent contact pads, where a constant current is applied, and the voltage drop measured	ge
3.13 Example of LTLM resistance as a function of gap distance. The intersection between the line and the resistance axis is double the contact resistance while the intersection with the distance axis represents double the transfer length	e, ne

3.14	SEM of a fabricated LTML test structure with a gap spacing of 2 $\mu m.$ The
	position of the current probes is marked with I1 and I2, while the sense probes
	are marked Vsense
3.15	Measured plot of resistance as a function of contact separation distance for
	TLM structure
4.1	DC measurement of a 4x4 $\mu m^2$ RTD device. The negative differential
	resistance (NDR) region is located between 0.6 V and 1.2 V. The diode shows
	a signature plateau-region within the NDR due to parasitic oscillations75
4.2	RTD I-V characteristic of the two wafer structures used in this project: (a) layer
	structure #1, (b) layer structure #2
4.3	RC small-signal circuit model for the RTD. The capacitance $C_n$ represents the
	geometrical capacitance of the device. The conductance G <sub>n</sub> is obtained by
	differentiating the stable I-V curve of the device
4.4	RLC small-signal equivalent circuit model for RTD. The capacitance $C_n$
	represents the geometrical capacitance of the device. The conductance $G_n$ is
	obtained by differentiating the stable I-V curve of the device
4.5	(a) DC measurement setup of an RTD device including the bias circuit and
	device parasitics, (b) simplified equivalent circuit
4.6	Circuit for stabilising the RTD using a parallel resistance $(R_{\mbox{\scriptsize shunt}})$ across the
	device. L denotes the parasitic inductance. $C_n$ is the diode capacitance and $-G_n$
	is the negative differential conductance. The device contact resistance has been
	omitted
4.7	Circuit for stabilising the RTD using a parallel resistance (R <sub>B</sub> ) and capacitance
	(C <sub>B</sub> ) across the device
4.8	(a) Simplified low frequency equivalent circuit, (b) simplified high frequency
	equivalent circuit
4.9	DC measurement set-up for an RTD device. A spectrum analyser/oscilloscope
	is used to measure the parasitic oscillations

4.10 Measured I-V characteristic of an RTD device (black), large-signal model of
the I-V characteristic with the NDR approximated with a smooth trace (blue), calculated value of conductance of model I-V (red)
4.11 Low frequency oscillations of the RTD measured by a spectrum analyser 88
4.12 Micrograph of open and short dummy pad structures used for de-embedding of the parasitic elements originating from the CPW pads
4.13 Measured and simulated real and imaginary parts of $S_{11}$ after de-embedding the parasitics at 0.1 V bias voltage. No data could be extracted above 15 GHz as the RTD and the pads were in resonance
4.14 Extracted values for the device inductance (red), model of the device inductance obtained (blue)
4.15 Extracted values for the device capacitance (red), model of the device capacitance obtained (blue)
4.16 MATLAB Simulink large-signal model of RTD measurement setup. The negative differential resistance has been modelled as a variable resistance with the values extracted from the DC large-signal model of the RTD (Fig. 4.10). The inductance and capacitance have been modelled using custom block with their values extracted from Fig.4.14 and Fig.4.15, respectively
<ul> <li>4.17 Measured I-V characteristic of an RTD device (black), Simulated MATLAB model of DC measurement setup with added extrinsic parasitic elements (red)</li></ul>
4.18 Micrograph of a fabricated $4x4 \ \mu m^2$ RTD device with integrated bias stabilisation. The capacitor C <sub>B</sub> and resistor R <sub>B</sub> are comprised of two parts connected in parallel with the RTD device
4.19 Frequency spectrum of RTDs with different stabilisation networks
4.20 RTD I-V characteristics: (a) $4x4 \ \mu m^2$ RTD I-V with and without bias

5.1	Schematic block diagram used for on-wafer frequency measurements using a
	spectrum analyser. Biasing of the device is done through a ground-signal-
	ground (GSG) probe, an external DC block capacitor is used when not
	incorporated on-chip

- 5.2 RF large-signal equivalent circuit of an RTD based oscillator with the RTD represented by a voltage controlled current source I = f(V) and self-capacitance  $C_n$ . L represents the resonating inductor and Rload is the load resistance .. 102

- 5.7 RTD RF small-signal equivalent circuit with parasitic resistance R<sub>s</sub>...... 109

5.11 3D illustration of a single RTD device oscillator. The shunt resistance R <sub>B</sub> is fabricated from two parts placed on either side of the DC line
<ul><li>5.12 Measured single side-band phase noise of 30 GHz single RTD oscillator at 10 kHz offset from carrier to 100 MHz offset from carrier.</li></ul>
<ul> <li>5.13 Simulated and measured spectrum of RTD oscillator. The output power of the simulated RTD is about 3 mW with an oscillation frequency of 30.9 GHz.</li> </ul>
5.14 Proposed RTD device oscillator topology with RC stabilisation network 119
5.15 Micrograph of the fabricated RTD oscillator with an integrated stabilising network. For measurement, a GSG probe is used. The capacitor $C_B$ and resistance $R_B$ are split in two and placed in parallel with the RTD. The capacitor $C_E$ acts as a short to ground for the RF signal
5.16 Measured high efficiency RTD spectrum at $V_{bias} = 0.94$ V and $I_{bias} = 18$ mA The frequency of oscillation is 34.1GHz with an output power of 3.95 dBm
5.17 Measured oscillator output power and frequency as a function of $V_{bias}$ 121
<ul> <li>5.18 Measured phase noise for the high efficiency RTD oscillator. The phase noise is -89 dBc/Hz at 100 kHz offset, -137 dBc/Hz at 1 MHz offset and -133 dBc/Hz at 10 MHz offset from the carrier</li></ul>
6.1 Schematic block diagram of wireless data transmission setup using RTDs as transmitters and receivers
6.2 Operation of the RTD as a transmitter using OOK modulation scheme. The BE signal is used to move the operating point of the RTD (on/off)
6.3 Amplitude modulation of a carrier signal by a much lower frequency modulating signal at 0.5 and 1 modulation index value. For OOK modulation the modulation index is 1 (100 %)
6.4 RF spectrum of transmitter RTD when biased in the NDR and the BB signal was used to move the biasing point in the second PDR region. The modulating

	sidebands can be seen at a 1 GHz distance from the carrier (1-0-1-0 pattern @
	2 GHz clock frequency)
6.5	Operating point of an RTD receiver. The non-linear characteristics of the IV
	are shown in the inset
6.6	Eye diagrams of RTD wireless link (a) 1 Gbit/s data rate , 7.2mV $V_{\text{pp}},\ S/N =$
	15 (b) 2 Gbit/s data rate, 2.2 mV $V_{pp}$ , $S/N = 5$
6.7	Picture of the measurement setup used in the wireless data transmission
	link
7.1	Layout of CPW balun, the phase difference between the output at port 2 and 3
	is 180°
7.2	Schematic block diagram of a BPSK modulator. The switches represent the
	PIN diodes, which are used to turn on/off the carrier input141
A.1	Fabrication process: (a) top contact metal deposition, (b) etch to bottom
	InGaAs layer, (c) bottom contact metal deposition, (d) etch to InP substrate,
	(e) deposit polyamide and open VIA for bond pad contact
A.2	Fabrication process: (a) thin-film resistor and 1st bond pad metal deposition,
	(b) dielectric silicon nitride deposition (Si3N4), (c) 2nd bond pad metal
	deposition150

## List of Tables

1.1	III-V RTD material parameters at room temperature	19
2.1	RTD epi-layer design structure used in this project	39
3.1	Thin-film NiCr and InGaAs resistor test structure measurements	65
3.2	TLM measurement data	70
3.3	Extracted TLM parameters	70
4.1	RTD epi-layer design structures used in this project	76
5.1	Measurement results of the single RTD oscillators 1	18
5.2	Comparison between different VCOs in the Ka-band	23

## List of Symbols, Acronyms and Constants

#### Acronyms

- AC Alternating Current
- ADS Advanced Design System
- CAD Computer Aid Design
- CMOS Complementary Metal Oxide Semiconductor
- CPW Coplanar Waveguide
- DBQW Double Barrier Quantum Well
- DC Direct Current
- DI De-Ionised
- DUT Device Under Test
- FIR Far-InfraRed
- FOM Figure of Merit
- Gbitps Gigabits per second
- GHz Gigahertz
- GSG Ground Signal Ground
- HEMT High Electron mobility Transistor
- ICP-CVD Inductively Coupled Plasma Chemical Vapour Deposition
- IF Intermediate Frequency
- IMPATT Impact Ionization Transit Time
- IOT Internet of Things
- IP Internet Protocol
- ITU International Telecommunications Union
- JWNC James Watt Nanofabrication Centre
- LO Local Oscillator
- LOR Lift-off Resit
- MBE Molecular Beam Epitaxy
- MMIC Monolithic Microwave Integrated Circuit
- MIM Metal-Insulator-Metal

- NDR Negative Differential Resistance
- PCA Photoconductive Antenna
- PDR Positive Differential Resistance
- PVCR Peak Valley Current Ration
- QCL Quantum Cascade Laser
- RF Radio Frequency
- RIE Reactive Ion Etching
- RTD Resonant Tunneling Diode
- SEM Scanning Electron Microscope
- SOLT Short-Open-Load-Thru
- S-parameters Scattering Parameters
- TCR Temperature Coefficient of Resistance
- THz Terahertz
- TLM Transmission Line Model / Method
- TUNNET Tunnel Injection Transit Time
- VCO Voltage Controlled Oscillator
- VIA Vertical Interconnect Access
- VNA Vector Network Analyser

### **Physical Constants**

- $c_0 = 3 \times 10^8 \text{ m/s}$  Speed of Light in Free Space
- $\mu_0 = 4\pi \times 10^{-7} \text{ H} / \text{m}$  Vacuum Permeability
- $\epsilon_0 = 8.85 \times 10^{-12} \text{ F} / \text{m}$  Vacuum Permittivity
- h  $= 4.136 \times 10^{-15}$  eVs Planck's Constant
- $\hbar$  = 6.582 × 10<sup>-15</sup> eVs Reduced Planck's Constant
- e  $= 1.6 \times 10^{-19}$  C Elementary Charge
- $m_0 = 9.11 \times 10^{-31}$  kg Electron Rest Mass

### **Chemical Symbols**

- (CH<sub>3</sub>)<sub>2</sub>CO Acetone
- Al Aluminium

- AlAs Aluminium Arsenide
- AlSb Aluminium Antimonide
- As Arsenic
- Au Gold
- C<sub>3</sub>H<sub>8</sub>O Propanol
- C<sub>6</sub>H<sub>6</sub>Cl Chlorobenzene
- C<sub>7</sub>H<sub>8</sub> Toluene
- CH<sub>3</sub>OH Methanol
- CO<sub>2</sub> Carbon Dioxide
- Ga Gallium
- GaAs Gallium Arsenide
- Ge Germanium
- H<sub>2</sub>O<sub>2</sub> Hydrogen Peroxide
- H<sub>3</sub>PO<sub>4</sub> Phosphoric Acid
- HCl Hydrogen Chloride
- In Indium
- InAlAs Indium Aluminium Arsenide
- InAs Indium Arsenide
- InGaAs Indium Gallium Arsenide
- InP Indium Phosphide
- Mo Molybdenum
- NH<sub>4</sub>OH Ammonium Hydroxide
- NiCr Nickel Chrome
- Pd Palladium
- Si Silicon
- Si<sub>3</sub>N<sub>4</sub> Silicon Nitride
- SiGe Silicon Germanium
- SiO<sub>2</sub> Silicon Dioxide
- Ti Titanium

### Symbols

- C<sub>n</sub> RTD Self-capacitance
- E<sub>g</sub> Band Gap
- I<sub>p</sub> Peak Current
- I<sub>v</sub> Valley Current
- J Current Density
- m<sup>\*</sup> Electron Effective Mass
- V<sub>p</sub> Peak Voltage
- V<sub>v</sub> Valley Voltage
- $\Delta E_c$  Conduction Band Offset
- ε<sub>r</sub> Dielectric Constant
- $\lambda$  Wavelength

# Chapter 1. Millimetre and Terahertz Wave Radiation

### **1.1 Introduction**

Millimetre and terahertz wave technologies are fields that lie in between microwave electronics and photonics. Millimetre waves (mm waves) are located within the 30 GHz – 300 GHz frequency band, with wavelengths,  $\lambda$ , corresponding to 10 mm to 1 mm wavelength. Electromagnetic waves with  $\lambda$  between 1 mm and 100  $\mu$ m are usually referred to as terahertz (THz) or sub-millimetre wave (sub-mm-wave) radiation. This corresponds to the frequency range between 300 GHz and 3 THz [1]-[3]. The electromagnetic spectrum is shown in Fig. 1.1.



*Figure 1.1:* Electromagnetic spectrum showing the millimetre and submillimetre radiation location between microwaves and infrared spectrum.

In nature, mm-wave and sub-mm-wave emissions occur from black body radiations with temperatures between 14 K - 140 K, well below the ambient background temperature on Earth [1]. The results from the NASA Diffuse Infrared Background Experiment (DIRBE), show that approximately one-half of the total luminosity and 98% of the number of photons emitted since the Big Bang fall into the millimetre and sub-millimetre frequency range [4].

The amount of energy carried by a single photon is directly proportional to the photon's electromagnetic frequency, and equivalently, inversely proportional to the wavelength. The photon energy can be calculated using Eq. 1.1

$$E = \frac{hc}{\lambda} \tag{1.1}$$

where *h* is Planks constant, *c* represents the speed of light and  $\lambda$  is the wavelength of the radiation. For the frequency span 30 GHz – 3 THz, the corresponding photon energy level ranges from 0.12 meV – 12.4 meV. This energy level is well below the range of ionising radiation level (10 eV), as they do not contain the amount of energy needed to displace an electron from an atom or molecule, in contrast with X-rays which have enough energy to strip electrons from an atom and therefore causing radiation damage.

The mm waves, more relevant to the current work, present different atmosphere attenuation rates at different frequencies. As shown in Fig. 1.2, the electromagnetic waves with frequencies up to and including microwave are generally unaffected by atmospheric effects, making them suitable for long-range radio communications. However, millimetre and terahertz waves suffer from attenuation caused by rain and gas molecules, making them unsuitable for long-distance transmission [5].



*Figure 1.2:* Atmospheric attenuation and expected data transfer rate of different propagation frequencies . Adapted from [5].

Radiation in these frequency bands is of great interest as it offers many different applications such as ultra-fast wireless communication for short-range, high-capacity, line-of-sight data transfer [6], security imaging, which exploits the properties of terahertz wave to penetrate many non-conducting materials to identify hidden objects [7][8], and non-invasive highly sensitive medical imaging due to the non-ionizing nature of the radiation [9].

The current cellular network technologies operate at frequencies between 800 MHz to 2.6 GHz. The International Telecommunication Union (ITU), the organization that coordinates the shared global use of the radio spectrum, published the WRC-19 document [10], which states that 24-86 GHz frequency bands will be used for the next 5G technologies, with a particular focus on the spectrum range 26.5 to 29.5 GHz, which many countries have already adapted while waiting for the specifications to be completed.

Despite of on-going research in millimetre and terahertz waves, the lack of reliable, compact, high power sources, which can operate at room temperature, has resulted in this being one of the least used regions of the electromagnetic spectrum. This region is also referred in literature as the "THz gap" [11]-[13] because there had been no effective way of generation and detection of THz signals. This is illustrated in the output power versus frequency graph of various electronic and photonic sources shown in Fig. 1.3.



*Figure 1.3:* Semiconductor devices showing the RF power output decreasing towards the THz-gap. Adapted from [11].

### **1.2 THz Applications**

New advances in different technologies have made the previously unused portion of the electromagnetic spectrum accessible to a wide variety of applications. This section therefore tries to provide a brief overview of the possible applications of millimetre and terahertz waves. The first major commercial application area is in millimetre and sub-millimetre wave communications. It is foreseen that the global internet protocol (IP) traffic will increase threefold in 2022 when compared to that of 2017 and reach about 400 exabytes (EB) per month [14]. Most notably the traffic from wireless and mobile devices account for 71% of the total traffic, driven largely by the popularity of content-based applications like YouTube and Netflix, as well as the expansion of Internet of Things (IoT) devices, allowing in principle any object to be networked wirelessly [15]-[17].

The evolution of the radio frequency (RF) spectrum used in mobile networks (from 1G analog cellular networks to high speed, internet protocol based 4G networks) has allowed for larger communication bandwidths and/or tailoring the employed technology to the specific link requirements. As such as higher frequencies are opening up, the capacity of wireless networks will extend beyond the few gigabits per second (Gbit/s) into the range of hundreds of Gbit/s [18][19]. The RF spectrum and mobile network band allocations are presented in Fig. 1.4 [20].



*Figure 1.4:* Electromagnetic spectrum from microwave to the end of the THz region, showing the allocations for 5G, 6G and spectrum unification for wireless data centres. Adapted from [20].

As data volumes grow substantially so too data centres need to evolve and develop new hybrid architectures based on wireless and mobile technology to attain higher bandwidth, lower latency and higher fault tolerance. The total energy consumption of data centres worldwide was estimated at about 270 TWh in 2012, which represented about 2% of the global power consumption [21], with a significant portion of the power consumed attributed to the air-conditioning of the equipment. In the front section of a data centre cabinet there are several tens of metres of category 6 ethernet cables connecting multiple servers together [22]. This cabling substantially impedes the proper airflow and increases the cooling energy cost. Furthermore, the wired networks entail high wiring cost, suffer from performance bottlenecks and have a low fault tolerance. Mobile data centres can solve these issues by replacing hundreds of meters of cable connections with mm-wave wireless connections both intra- and inter-rack. Fig. 1.5 depicts a hybrid data centre with ad-hoc point to point wireless connectivity to help solve hotspots/bottlenecks and allow reconfigurability.



*Figure 1.5: Hybrid wired / wireless data centre with wireless nodes connection between the server racks. Adapted from [20].* 

Moreover, spectrum regulations can be somewhat relaxed when compared to outdoor applications allowing for spectrum unification, with more than three decades of continuous electromagnetic spectrum (60 GHz - 3 THz) for usage [20]. Key elements for opening the new opportunities in communication are technical performance and cost. Therefore, a high performance, reliable and high yield monolithic microwave integrated circuit (MMIC) process needs to be available.

Another possible application stems from the fact that THz radiation is characterized by its low photon energy, which would not cause detrimental photoionization in organic tissue. As such the technology opens a multitude of non-invasive applications in the biomedical field. In [23] the authors make direct measurements of the water content of the cornea, made possible by exploiting the high sensitivity of terahertz frequencies to water content, which can be used to assess the onset and evolution of diseases which affect the hydration of the eye such as dystrophies, degenerations, injuries and corneal graft rejection and can be helpful in evaluating contact lenses and eye drops in order to preserve the health hydration balance in the cornea.

In oral healthcare X-rays only reveal tooth decay at relatively late stages, when drilling and filling is the only method available to stop the process. Terahertz imaging can distinguish between the different types of tissue in a human tooth; detect carriers at an early stage in the enamel and dentine layers and monitor demineralization spots in the tooth. In [24] the authors used frequencies between 200 GHz and 1.5 THz where the contrast was associated with changes of refractive indexes. If the decay can be detected early enough it is possible to reverse the process without the need for drilling by using either fissure sealing or remineralization [25].



*Figure 1.6: Transmission image of a tooth showing the contrast at several frequencies in the mm-wave and terahertz range. Adapted from [24].* 

Intensive research is currently being carried out on intrinsic contrast reflection imaging techniques to differentiate cancerous and healthy tissue using THz radiation [26]-[29]. The main difficulty that limits its application is the penetration depth of THz radiation into water-abundant specimens, however it can be overcome by using several techniques such as freezing [26][27] and the use of penetration enhancing gels [30].

Another important property of THz radiation is that it can penetrate many dielectric materials such as plastic, wood, paper and semiconductors, which makes THz waves a valuable tool in non-destructive detection fields such as security where they can be used in the detection and identification of concealed prohibited items and/or explosives.

International customs and border security agencies have traditionally used screening technologies based on X-rays to inspect baggage, vehicles, mail and cargo for illicit goods. However, pedestrian based smuggling of items such as narcotics and weapons has largely been neglected due to the lack of viable solutions. Among promising candidates is a full body passive millimetre-wave security screening portal such as the one developed by Trex Enterprises working at frequencies between 75.5 GHz to 93.5 GHz [31]. The system requires 48 seconds of scan time due to its limited number of detectors and use of mechanical scanning. An image collected with the system showing concealed objects (black contrast) against the emitted body heat (white) is shown in Fig. 1.7.



*Figure 1.7:* Passive full body millimetre-wave security screening portal developed by Trex Enterprises working between 75.5 GHz to 93.5 GHz. Adapted from [31].

Since of materials the properties are frequency dependent, the transmission/absorptivity, emissivity, reflectivity and scattering of the object of interest all change with respect to frequency. In [32] the authors showed how multispectral THz imaging can be used to combine images taken at different frequencies in order to enhance the contrast and spatial resolution of the final processed image. The Safe VISITOR: Visible, Infrared and Terahertz Object Recognition project uses the 350 GHz frequency range to define and detect objects from 5 m distance and resolve them to 256 shades of grey [33].



*Figure 1.8:* Image fusion from Safe VISITOR project. (a) visible image, (b) THz image at 350 GHz, (c) Infrared image, (d) final fused image. Adapted from [33].

### **1.3 THz Sources**

The two main sources for terahertz radiation are electronic, which operate at the lower frequency end of the spectrum and photonic, operating at the higher frequency end. There are many candidates for electronic THz sources such as tunnel transit-time (TUNNET) diodes, impact ionization avalanche transit-time (IMPATT) diodes, Gunn diodes, resonant tunneling diodes (RTDs) and CMOS or III-V semiconductor frequency multipliers. Among them, the RTD has the highest reported oscillation frequency of 1.98 THz [34][35]. The most common photonic sources are terahertz photoconductive antennas (THz-PCAs) and terahertz semiconductor lasers, such as far-infrared (FIR) and quantum cascade lasers (QCL).

### **1.3.1 Electronic Sources**

Gunn diodes are two-terminal devices which exhibit negative differential resistance (NDR), and when coupled to a suitable resonator, are able to generate RF power. Typically, the Gunn diode is made of a uniformly doped n-type III-V material, such as gallium arsenide (GaAs) or indium phosphide (InP) sandwiched between heavily doped regions at each end (Fig. 1.9 (a)). In these types of devices, the lowest conduction band profile presents distinct valleys, labelled  $\Gamma$  and L, as shown in Fig. 1.9 (b) [36].



*Figure 1.9:* Operating principle of Gunn diodes (a) Gunn diode structure, (b) conduction and valence band profile of Gunn diode. Adapted from [36].

When the electrons reside in the lower  $\Gamma$ -valley, they exhibit very high mobility and a small effective mass, whereas in the L-valley the effective mass is large, and the mobility is low. The valleys are separated by a small energy gap  $\Delta E$ . When sufficient bias voltage is applied, electrons gain enough energy to be transferred from the  $\Gamma$ -valley to the L-valley, where due to their increased effective mass the electrons have a lower drift velocity and the current decreases with bias. Gunn diodes are conventionally implemented in vertical device structures and are available in a number of commercial and industrial applications such as radar [37] and imaging [38].

A planar Gunn diode operating at 78.9 GHz was reported in [39] with 24.7 mW output power and 2.5% conversion efficiency. Fundamental oscillation frequencies of 100

GHz - 150 GHz with very high output power (300 mW - 100 mW) were achieved with devices mounted on diamond heat sinks [40] in a dual cavity power combining circuit as shown in Fig. 1.10. The authors later reported second harmonic oscillators with 4.8 mW output power at 282 GHz [41]. The efficiency of such oscillators is quite low at around 0.2%.



*Figure 1.10:* Dual cavity power combining circuit using Gunn diodes mounted on diamond heatsinks. Adapted from [40].

Impact ionization avalanche transit time diodes or IMPATT diodes in their basic form are based on a reverse-biased p-n junction and an intrinsic (high-resistivity) drift layer. When a sufficiently high reverse bias voltage is applied across the junction, avalanche breakdown occurs, resulting in a large number of carriers in the avalanche region. The negative differential resistance (NDR) characteristic of an IMPATT diode is produced by a combination of impact avalanche breakdown and charge-carrier transit time effects. Inside the diode the current lags the voltage by 90 degrees, known as the injection delay. The length of the diode can then be tailored so that the transit time delay results in a further 90 degrees phase lag and therefore the NDR is formed. An external resonant circuit can then be connected to the diode in order to sustain the oscillations. The basic structure of an IMPATT diode is shown in Fig. 1.11.



*Figure 1.11: IMPATT diode structure. When a sufficiently high negative voltage is applied, avalanche breakdown occurs in the*  $p^+$ *- n region resulting in a large number of carriers. The phase lag is controlled by the width of the drift region.* 

The main advantage of the IMPATT diode is its high RF power capability. IMPATT diode oscillators have been reported with fundamental frequencies up to 217 GHz in continuous wave mode, with an output power of 1 W implemented in rectangular waveguide technology [42]. Monolithically integrated IMPATT diodes operating at 82 GHz with 14.6 mW output power have been reported in [43], with 4.3% conversion efficiency.

The main disadvantage of the IMPATT diode is that the avalanche breakdown process is noisy, making the oscillators susceptible to high phase noise [36]. A variation of the IMPATT diode is the tunnel injection transit time or TUNNETT diode, where the avalanche region of the IMPATT diode is changed to a highly doped, narrow,  $p^{++}n^+$ junction that changes the breakdown mechanism from avalanche to tunnel injection. Compared with the IMPATT diode, the TUNNETT exhibits very low noise and high operation frequency at the cost of output power. TUNNET diodes have been developed with fundamental frequencies at 655 GHz with 0.8 nW output power and 0.2 nW at 701 GHz [44][45].

Frequency multiplication is the most common approach for generating THz radiation. A non-linear element, which is usually a Schottky diode because of their low selfcapacitance and very high switching rate, implemented in CMOS or III-V semiconductor technologies is used to generate harmonics of an input signal.
Multiplier chains with cascaded frequency doublers and triplers are used to upconvert signals from the GHz frequency range up to THz frequencies.

However, the main disadvantage of multipliers chains is that they require high power input signals, have low power generation capability and conversion efficiency. To date the highest reported oscillation frequency is 2.55 THz with an output power of 0.1  $\mu$ W and 0.002 % conversion efficiency [46].

### **1.3.2 Photonic Sources**

The photoconductive antenna (PCA), consists of a low temperature grown GaAs or InGaAs film covered by two dc-biased electrodes that form an antenna. An ultrashort optical pulse from a femtosecond laser is focused on the gap between the electrodes. The optical pulses generate free electrons, which are accelerated by the electric field formed at the interface between the photoconductor and the antenna electrodes before becoming trapped in the semiconductor. If the process occurs at sub-picosecond timescale the resulting impulse current is at THz frequencies [47]. The operating principle is shown in Fig. 1.12.

However, since the output of such a system is not a continuous wave THz radiation, they are most commonly used in spectroscopy applications for material characterization. The PCA has been extensively studied and many different structures have been presented, such as the THz dipole antenna [48], Yagi–Uda antenna [49], spiral-type antenna [50], and butterfly-shaped antenna [51], with various degrees of radiation efficiency.



*Figure 1.12:* Photoconductive antenna operating principle. A femtosecond laser is focused at the gap between the two electrodes generating free carriers, which are then accelerated by the bias voltage across the gap. Adapted from [47].

While most lasers are located at the high end of the THz range, several laser devices have been developed to bridge the THz gap, such as far-infrared gas lasers (FIR) and quantum cascade lasers (QCL). In FIR lasers the THz emission originates from transitions between rotational states of gas molecules such as methanol (CH<sub>3</sub>OH), which are stimulated by an external carbon dioxide (CO<sub>2</sub>) laser [52][53]. High output power of 100 mW has been reported at a frequency of 2.5 THz for such a system [54]. QCL lasers are semiconductor heterostructures capable of emitting continuous wave THz radiation by exploiting transitions between conduction band states [55]. The lowest published frequency for QCL is 2.1 THz with 1.2 mW power in continuous-wave mode [56]. However, both systems are bulky and QCLs require cryogenic cooling to operate, making them unusable outside the laboratory environment.

## **1.4 The Resonant Tunneling Diode**

## **1.4.1 Historical Perspective**

In 1958, a Japanese PhD research student named Leo Esaki reported at the International Conference on Solid State Physics in Brussels a new type of diode based on the quantum-mechanical tunneling effect, which exhibited a region of negative differential resistance [57]. In 1973 Esaki received the Nobel Prize in physics for the pioneering work done in the first experimental demonstration of electron tunneling in semiconductor materials [58]. The tunnel diode was made with a very highly doped (10<sup>19</sup> cm<sup>-3</sup>) germanium p-n junction [59]. The operation principle of the tunnel diode is shown in Fig. 1.13.

At zero and low forward bias condition, shown in Fig. 1.13 (a) the states in the conduction band of the n-type doped side are filled with electrons resulting in the Fermi level ( $E_{Fn}$ ) to move up and into the conduction band. Likewise, the states in the valence band on the p-type material are not filled with electrons, and as a result the Fermi level ( $E_{Fp}$ ) goes down into the valence band. When a forward bias is applied, a number of electrons will have sufficient energy to tunnel through the barrier between the conduction band of the n-type to the valence band of the p-type material. The current will continue to increase with forward bias until the Fermi level  $E_{Fn}$  aligns with the p-type material valence band energy level shown in Fig. 1.13 (b), where a peak current value ( $I_p$ ) is reached, at which the bias voltage is called peak voltage ( $V_p$ ). When the bias voltage is increased beyond  $V_p$ , i.e. Fig. 1.13 (c), the tunneling current begins to drop rapidly as the bands between both sides become mis-aligned and the tunneling probability decreases until the valley voltage is reached ( $V_v$ ) with valley current ( $I_v$ ). A further increase in forward bias beyond the  $V_v$  point, as shown in Fig. 1.13 (d), results into conduction of carriers through diffusion as a normal p-n diode.



*Figure 1.13:* Energy band diagram of a tunnel diode and its I-V characteristic. (a) Tunnel diode at zero bias, (b) band diagram at the peak voltage, (c) band diagram at the valley voltage, (d) band diagram at the second positive differential region, (e) I-V characteristic of tunnel diode at each biasing point.

The Esaki tunnel diode was employed in high frequency amplifiers, logic circuits and oscillator applications, as early transistors did not operate well at frequencies of several GHz [60]. However, their inherently high specific capacitance restricted their use at higher frequencies, i.e. millimetre wave frequencies and above.

A new tunneling phenomenon was reported by Tsu and Esaki in 1973 [61] while they were exploring the properties of superlattices in terms of electrical transportation. This theoretically predicted behaviour showed electrons tunneling inside a double barrier quantum well (DBQW) in the form of a thin gallium arsenide (GaAs) well sandwiched between two aluminium gallium arsenide (AlGaAs) barriers. A year later, the team demonstrated a practical quantum barrier device, currently known as the resonant tunneling diode or RTD [62].

### **1.4.2 Operational Principle of RTD**

The operating principle of the resonant tunneling diode (RTD) can be illustrated with the help of the conduction band diagram of the double barrier structure at different bias voltages with the corresponding *I-V* characteristics shown in Fig. 1.14. The quantum well in its most basic form consists of a layer of narrow-gap material sandwiched between two large-gap layers. The resulting conduction band profile resembles a square well whose width is dictated by the thickness of the narrow gap material and depth is controlled by the conduction band discontinuity between the barrier and well material. For electrons trapped inside the well, the motion of electrons is quantized in discrete energy states called resonant energy levels. If no voltage is applied, then the fermi level of the emitter  $E_{FE}$  and the fermi level of the collector are aligned below the resonant energy levels  $E_{rI}$  and  $E_{r2}$  as shown in Fig. 1.14 (a). In this case, no current will flow from the emitter to the collector due to thermal equilibrium.

When the bias voltage increases from zero, the conduction band profile of the DBQW structure shifts as shown in Fig. 1.14 (b). The bias voltage induces an electric field from the emitter to the collector and electrons obtain kinetic energy, and thus the probability of electrons to tunnel through the barrier increases in accordance with the bias level. This corresponds with the first positive differential resistance (PDR) region on the device's *I-V* characteristic. By further increasing the bias voltage the emitter conduction band reaches the resonant energy state, shown in Fig. 1.14 (c), where the probability of tunneling through the barrier is unity. This corresponds to the voltage peak point ( $V_p$ ) and current peak ( $I_p$ ) in the *I-V* characteristic.

Increasing the potential beyond  $V_p$  the resonant energy level and the conduction band energy level become mis-aligned and the probability of electrons to tunnel through drops rapidly as shown in Fig. 1.14 (d). This will correspond to the negative differential region of the diode, where  $V_v$  is the valley voltage and  $I_v$  is the valley current. Further increasing the applied bias voltage, the electrons acquire enough kinetic and thermal energy to go over the barrier and the current rises again with bias voltage, Fig. 1.14 (e).



**Figure 1.14:** Energy band diagram of a resonant tunneling diode (RTD) and its I-V characteristic. (a) RTD at zero bias, no current flows due to thermal equilibrium, (b) a small bias voltage is applied increasing the probability of electrons to tunnel through the barrier, (c) band diagram when the conduction band aligns with the resonant energy state  $E_{r1}$  and the transmission probability is at unity, (d) band diagram when the conduction band and the resonant energy state become misaligned and the current drops rapidly, (e) at a high enough bias voltage the current starts increasing again due to thermal emission, having enough energy to go over the barrier, (f) the I-V characteristic of the RTD at each biasing condition.

From the *I-V* characteristics of an RTD device, different key parameters can be extracted to estimate the device performance. The figures of merit (FOM) of an RTD include the peak-to-valley voltage difference ( $\Delta V = V_v - V_p$ ), the peak-to-valley current difference ( $\Delta I = I_p - I_v$ ), and the peak-to-valley current ratio (PVCR =  $\frac{I_p}{I_v}$ ). The parameters  $\Delta V$  and  $\Delta I$  are used to estimate the maximum RF power that an RTD-based oscillator can deliver to the load, which is given by  $P_{max} = \frac{3}{16} \Delta I \Delta V$  [63]. A large PVCR maximises the dynamic range of the device. The valley current can often be associated with thermionic emission through higher resonant states or above the barrier, scattering effects and tunneling via X and L states [64].

#### **1.4.3 RTD Material Systems**

RTD devices are usually realized using III-V semiconductor materials [65]-[67], that possess a small effective electron mass ( $m^*$ ) and have a high electron mobility, which leads to a high current density ( $J_p$ ). Furthermore, the materials have a high conduction band offset which improves the peak voltage-current ratio (PVCR) by suppressing thermal current [68]. The typical III-V materials used to realise RTDs are gallium arsenide / aluminium gallium arsenide (GaAs/AlGaAs), indium arsenide / aluminium antimonide (InAs/AlSb) and indium gallium arsenide / aluminium arsenide (InGaAs/AlAs). In addition, silicon-based RTDs such as silicon / silicon germanium (Si/SiGe) are also being investigated due to the potential integration of RTDs with silicon technologies.

The parameters of the RTD material systems are summarized in Table 1.1, where the effective electron mass is denoted with  $m^*$ ,  $m_o$  represents the electron rest mass  $(m_o = 9.11 \times 10^{-31} \text{ kg})$ , the band gap is denoted with  $E_g$ , relative dielectric constant ( $\epsilon_r$ ) and conduction band offset ( $\Delta E_c$ ) [69].

Material	m*	Eg (eV)	٤r	$\Delta E_{c} (eV)$
GaAs AlAs	$\begin{array}{c} 0.067m_o\\ 0.1m_o \end{array}$	1.42 2.16	12.9 10.1	0.28
In <sub>0.53</sub> Ga <sub>0.47</sub> As AlAs	0.042m <sub>o</sub> 0.1m <sub>o</sub>	0.71 2.16	12.9 10.1	0.65
InAs AlSb	0.027m <sub>o</sub> 0.12m <sub>o</sub>	0.36 1.61	14.6 12.04	1.35
In <sub>x</sub> Ga <sub>1-x</sub> As	(0.067+0.083x) m <sub>o</sub>	1.424+1.247 <i>x</i>	12.9-2.84 <i>x</i>	N/A
In <sub>0.48</sub> In <sub>0.52</sub> As	0.084m <sub>o</sub>	In <sub>0.53</sub> Ga <sub>0.47</sub> As	12.45	N/A

Table 1.1: III-V RTD material parameters at room temperature

The GaAs quantum well with  $Al_xGa_{1-x}As$  barriers was the first demonstrated resonant tunneling material system [70] and, remains an important system for high-speed RTDs due to the simple growing procedure [71]. By changing the composition (mole fraction *x*) a PVCR = 3.9 at room temperature with  $J_p = 7.7 \text{ kA/cm}^2$  was reported in [65].

The InAs/AlSb material system offers several advantages when compared to GaAs/AlGaAs, such as low effective mass and high conduction band offset. Secondly, using InAs instead of GaAs lowers the series contact resistance, as nearly ideal ohmic contacts can be formed to InAs, where the Fermi level is pinned in the conduction band [72]. However, the system can suffer from impact ionization due to the low bang gap of InAs ( $E_g = 0.36 \text{ eV}$ ). When a high electric field is applied, an electron with energy slightly larger than the band gap could collide with an electron in the valence band and knock it out into the conduction band. The number of current carriers will therefore multiply, and the current will begin to increase rapidly due to carrier multiplication, a process referred to as avalanche breakdown [73].

The most popular, and the one used in this work is the InGaAs/AlAs material system, as it can offer a very high PVCR and high peak to valley current density ( $\Delta J = J_p - J_v$ ). In [74] a PVCR of 30 was reported where an InAs sub-well and AlAs barriers was used. The InAs sub-well enabled a low resonance energy, provided a large separation between the resonant levels, and greatly reduced alloy scattering when compared to an InGaAs well. A comparison between the different RTD material systems can be seen in Fig. 1.15.



*Figure 1.15:* Current density versus voltage curves for different resonanttunneling diode materials. The dashed lines represent interpolations through the respective NDR regions. Adapted from [71].

Several attempts have been made to integrate an RTD into a complementary metal oxide semiconductor (CMOS) process [75] [76]. Memory and logic circuits employing RTD devices would provide a significant advantage to CMOS technology by reducing circuit complexity, power consumption, and enhancing high-speed operation [77]. Furthermore, series connected vertically integrated resonant tunneling diodes (VIRTD) with multiple negative resistance (NDR) regions would allow the realisation of multiple-valued memory (MVM) [78] [79]. Traditionally, the memory cell contains only one bit of information (1 or 0), however MVM offers the possibility of storing and transferring more information with fewer interconnects between devices, thereby providing a more compact solution to data storage.

#### 1.4.4 State of the Art

The first RTD oscillator was demonstrated by Sollner et. al. in 1984, with a maximum observed oscillation frequency of 18 GHz and 50  $\mu$ W output power [80]. For the first two decades of RTD device development, the oscillators were realized using a quasi-

optical resonator [81]-[83] as shown in Fig. 1.16, where a semi-confocal open-cavity resonator was used to stabilise the RTD waveguide oscillator.



*Figure 1.16:* Schematic cross-section of a quasi-optical RTD oscillator. Adapted from [81].

Due to the large contact inductance, introduced by the whisker contacts and the high series resistance a large portion of the RF power was dissipated due to spurious bias oscillations and by the high series resistance. The highest obtained oscillation power obtained was 50  $\mu$ W at a fundamental frequency of 210 GHz [83], and the highest frequency of the quasi-optical RTD was 720 GHz with 0.3  $\mu$ W output power [84].

Planar RTD technology has since become more popular due to advancements in manufacturing technology as well as fewer constraints imposed on the bias stability circuit. In 1994, Smith et. al. proposed a technique with the top collector Ohmic contact replaced by a Schottky contact in order to reduce parasitic resistance and consequently increase the maximum oscillation frequency [85]. A 64-element array was first demonstrated by Reddy et. al. using the Schottky contact technique (SRTD), shown in Fig. 1.17, in which a fundamental frequency of 650 GHz was achieved [86]. Furthermore, a Si lens was used to couple the power from the oscillator array from the substrate into free space.



*Figure 1.17:* Planar RTD oscillator with 64 elements (a) circuit schematic of a monolithic slot-antenna coupled SRTD oscillator, (b) oscillator array mounted on a silicon lens, (c) SEM image of  $0.1 - \mu m$  contact stripe, (d) oscillator array section, (e) photograph of a 64-element SRTD oscillator. Adapted from [86].

In 1995, a planar-slot antenna integrated RTD was first reported [87] with a fundamental and third harmonic frequency of 342 GHz and 1.02 THz, respectively. The devices were fabricated using e-beam lithography and metal-insulator-metal reflectors were integrated for the slot antenna. The output power at these frequencies was 23  $\mu$ W and 0.6  $\mu$ W, respectively. A bismuth sheet resistor was deposited in



parallel to the RTD in order to suppress the parasitic oscillations from the biasing circuit.

*Figure 1.18: Structure of RTD oscillator with integrated slot antenna. Adapted from [89].* 

Fundamental oscillation frequencies of 831 GHz, and 1.04 THz were later reported by the same research group in 2009 [88], 2010 [89], respectively through epi-layer design optimizations.

In 2011, Feigninov et. al. reported on an RTD oscillator operating at a fundamental frequency of 1.1 THz and an estimated 0.5  $\mu$ W output power [90]. The RTD mesa size was 0.35  $\mu$ m<sup>2</sup> and had a peak current density of 14 mA/ $\mu$ m<sup>2</sup>. The RTD oscillator used a planar Vivaldi antenna placed on top of a dielectric membrane that was used to extract the radiation. The RTD Vivaldi oscillator is shown in Fig. 1.19.



*Figure 1.19: RTD* oscillator integrated with Vivaldi antenna operating at 1.1 THz. Adapted from [90].

A new RTD circuit structure was proposed by Suzuki and Asada in 2007 based on an offset slot antenna, where the position of the RTD is shifted within the slot antenna [91]. Using this technique, the radiation conductance of the antenna and oscillation frequency can be independently determined by the lengths of the short and long parts of the slot, respectively. An RTD oscillator was fabricated with a fundamental oscillation frequency of 1.92 THz with 0.4  $\mu$ W output power [34].

High performance double RTD oscillators implemented in coplanar waveguide (CPW) technology were presented in [92] [93], operating in the J frequency band (220 GHz – 325 GHz) and with about 1mW output power. A picture of the double RTD oscillator design is shown in Fig. 1.20. It is expected that output powers of 10 mW and 1mW at 500 GHz and 1 THz, respectively, can be obtained from coupled RTD technology, making the RTD a suitable candidate for the next-generation millimetre-wave and submillimetre-wave applications [94].



*Figure 1.20:* A double RTD oscillator implemented in CPW technology operating in the J-band with around 1 mW RF output power. Adapted from [92].

# **1.5 Thesis Aim and Organization**

The aim of this PhD thesis was to realize and characterize RTD oscillators in monolithic microwave/millimetre-wave integrated circuit (MMIC) form for high power and frequency applications. The oscillators described in this thesis are designed to operate in the Ka-band (26.5 GHz – 40 GHz) with output power higher than 1 mW. A new circuit for RTD oscillators to suppress parasitic oscillations and increase the DC-to-RF conversion efficiency was developed and will be described.

The thesis is organized as follows: Chapter 1 introduces and reviews millimetre-wave and submillimetre-wave radiation and its characteristics. The main sources being developed are described in this chapter with discussion on possible applications. The operating principle of negative differential resistance devices is also presented alongside the current state of the art in RTD technology.

In Chapter 2 the main fabrication process required to realize RTD-based circuits using MMIC technology is discussed including photolithography, dry/wet etching, metallization, lift-off and vertical interconnect access (VIA) opening through polyimide.

Chapter 3 describes the fabrication process required for passive components that include metal-insulator-metal (MIM) capacitors, coplanar waveguide (CPW) transmission line, shorted CPW lines for inductors and thin-film resistors. The realization of low ohmic contact resistance to InGaAs is also investigated.

Chapter 4 discusses the bias circuit instabilities, their effects on performance and methods to suppress these parasitic effects. A method to accurately measure the DC current-voltage (I-V) characteristics and a model for the RTD is discussed.

Chapter 5 shows the design procedure of MMIC RTD oscillators, as well as defining the important parameters required to characterize voltage-controlled oscillators (VCOs). A new circuit topology for RTD oscillators is proposed, fabricated and compared with the previous generation of RTD oscillators in terms of DC-to-RF conversion efficiency. The RTD oscillators presented in this work operated at frequencies between 17 GHz - 36 GHz with most of them exceeding the 1 mW output power target.

Chapter 6 presents a wireless link application in the Ka-band (26.5 GHz - 40 GHz) using an RTD modulated by a pseudo-random binary sequence (PRBRS) generator as a transmitter (Tx), and an RTD receiver (Rx) to demodulate the signal. A measured 2 Gb/s wireless data transfer speed was achieved with this setup.

Finally, conclusions and future work are discussed in Chapter 7.

# References

- P. H. Siegel, "Terahertz technology," *IEEE Transaction on Microwave Theory* and Techniques, vol. 50, no. 3, pp. 910-928, 2002.
- [2] A. Rostami, H. Rasooli and H. Baghban, Terahertz Technology: Fundamentals and Applications, Sprigner, 2011.
- [3] B. Ferguson and X. C. Zhang, "Materials for terahertz science and technology," *Nature materials*, vol. 1, no. 1, pp. 26-33, 2002.
- [4] D. Leisawitz, "Scientific motivation and technology requrments for the SPIRIT and SPECS far-infrared/submilimeter space interferometers," in *Proc. SPIE, UV, Optical, and IR Space Telescopes and Instruments*, Munich, Germany, 2000.
- [5] N. Kukutsu and Y. Kado, "Overview of millimeter and terahertz wave application research," *NTT Technical Review*, vol. 7, no. 3, 2009.
- [6] A. Hirata, T. Kosugi, H. Takahashi, R. Yamaguchi, F. Nakajima, T. Furata, H. Ito, Y. Sato and T. Nagatsuma, "120-GHz-band milimeter-wave photonic wireless link for 10-Gb/s data transmission," *IEEE Transactions on Microwave Theory and Techniques*, vol. 54, no. 5, pp. 1937-1944, 2006.
- [7] F. Xin, H. Su and Y. Xiao, "Terahertz imaging system for remote sensing and security applications," in *Proceedings of 2014 3rd Asia-Pacific Conference on Antennas and Propagation*, Habin, 2014.
- [8] M. O. AlNabooda, R. M. Shubair, N. R. Rishani and G. Aldabbagh, "Terahertz spectroscopy and imaging for the detection and identification of illicit drugs," in 2017 Sensors Networks Smart and Emerging Technologies (SENSET), Beirut, 2017.
- [9] J.-H. Son, S. J. Oh and H. Cheon, "Potential clinical applications of terahertz radiation," *Journal of Applied Physics*, vol. 125, no. 19, 2019.

- [10] ITU, "World Radiocommunication Conference 2019," August 2017. [Online].
   Available: https://www.itu.int/dms\_pub/itu-r/oth/14/02/R1402000010001PDFE.pdf.
- [11] C. Sirtori, "Bridge for the Thz gap," Nature, vol. 417, pp. 132-133, 2002.
- [12] M. Cooke, "Filling the THz gap with new applications," *Semiconductor Today*, vol. 2, no. 1, pp. 39-43, 2007.
- [13] M. Tonouchi, "Cutting edge terahertz technology," *Nature Photonics*, vol. 1, pp. 97-106, 2007.
- [14] L. Zhand, Y. Liang and D. Niyato, "6G Visions: mobile ultra-broadband, super internet-of-things, and artificial intelligence," *China Communications*, vol. 16, no. 8, pp. 1-14, 2019.
- [15] M. R. Palattella, "Internet of Things in the 5G era: enablers, architecture, and business models," *IEEE journal on Selected Areas in Communications*, vol. 34, no. 3, pp. 510-527, 2016.
- [16] L. Zhang, Y. Liang and M. Xiao, "Spectrum sharing for Internet of Things: a survey," *IEEE Wireless Communications*, vol. 26, no. 3, pp. 132-139, 2019.
- [17] L. Zhang, M. Xiao, G. Wu, M. Alam, Y. Liang and S. Li, "A survey of advance techniques for spectrum sharing in 5G networks," *IEEE Wireless Communications*, vol. 24, no. 5, pp. 44-51, 2017.
- [18] J. Yu, X. Li, J. Zhang and J. Xiao, "432-Gb/s PDM-16 QAM signal wireless delivery at W-band using optical and antenna polarization multiplexing," in 2014 The European Conference on Optical Communication (ECOC), Cannes, 2014.
- [19] R. Puerta, S. Rommel, J. Olmos and I. Monroy, "Optically generated single sideband radio-over-fiber transmission of 60 Gbit/s over 50m at W-band," in *Optical Fiber Communications Conference and Exhibition (OFC)*, Los Angeles, USA, 2017.
- [20] S. Rommel, T. Raddo, U. Johannsen, C. Okonkwo and I. Monroy, "Beyond 5G
   wireless data center connectivity," in *Proceedings Volume 10945, Broadband* Access Communication Technologies XIII, San Francisco, USA, 2019.

- [21] J. Webber, H. Kamoda, N. Kukutsu and T. Kumagai, "Millimeter-wave wireless communication in a data center cabinet with adaptive control of propagation," in 21st Asia-Pacific Conference on Communications (APCC), Kyoto, Japan, 2015.
- [22] M. F. Bari, "Data center network virtualization: a survey," IEEE Communications Surveys & Tutorials, vol. 15, no. 2, pp. 909-928, 2013.
- [23] D. Bennett, Z. Taylor, P. Tewari, R. Singh, M. Culjat and W. Grundfest, "Terahertz sensing in corneal tissues," *Journal of Biomedical Optics*, vol. 16, no. 5, 2011.
- [24] N. N. Zinov, "Identification of tooth decay using terahertz imaging and spectroscopy," in *Twenty Seventh International Conference on Infrared and Millimeter Waves*, San Diego, USA, 2002.
- [25] J. Holtzman, D. Kohanchi, J. Biren-Fetz, M. Fontana, M. Ramchandani and K. Osann, "Detection and proportion of very early dental caries in independent living older adults.," *Lasers in Surgery and Medicine*, vol. 47, no. 9, pp. 683-691, 2015.
- [26] Y. C. Sim, J. Y. Park, K. M. Ahn, C. Park and J. Son, "Terahertz imaging of excised oral cancer at frozen temperature," *Biomedical Optics Express*, vol. 4, pp. 1413-1421, 2013.
- [27] Y. C. Sim, K. Ahn, J. Y. Park and J. Son, "Temperature-dependent terahertz imaging of excised malignant oral melonoma," *IEEE Biomedical Health nformation*, vol. 17, pp. 779-784, 2013.
- [28] C. Joseph, R. Patel, V. Neel, R. Giles and A. Yaroslavsky, "Imaging of ex vivo non-melanoma skin cancers in the optical and terahertz spectral regions," *Journal of Biophotonics*, vol. 7, no. 5, pp. 295-303, 2014.
- [29] B. S. Peter, "Development and testing of a single frequency terahertz imaging system for breast cancer detection," *IEEE Journal of Biomedical and Health Informatics*, vol. 17, no. 4, pp. 785-797, 2013.

- [30] S. J. Oh, S. H. Kim, K. Jeong, Y. Park, Y. Huh, J. Son and J. Suh, "Measurment depth enhancement in terahertz imaging of biological tissues," *Optics Express*, vol. 21, pp. 21299-21305, 2013.
- [31] C. Martin, C. Ganzales, V. Kilinko and J. Lovberg, "Rapid passive mmw security screening portal," in SPIE Defense and Security Symposium, Orlando, USA, 2008.
- [32] M. Kemp, A. Glauser and C. Baker, "Multi-spectral terahertz imaging using reflected and scattered radiation," *International Journal of High Speed Electronics and Systems*, vol. 17, no. 2, pp. 403-414, 2007.
- [33] T. May, G. Zieger, S. Anders, V. Zakosarenko, H. Meyer, M. Schubert and M. Starkloff, "Safe VISITOR: visible, infrared, and terahertz object recognition for security screening application," in SPIE Defense, Security, and Sensing, Orlando, USA, 2009.
- [34] M. Asada and S. Suzuki, "Room-temperature resonant-tunneling-diode terahertz oscillators," in 2016 41st International Conference on Infrared, Millimeter, and Terahertz waves (IRMMW-THz), Copenhagen, Denmark, 2016.
- [35] R. Izumi, S. Suzuki and M. Asada, "1.98 THz resonant tunneling diode oscillator with reduced conduction loss by thick antenna electrode," in 42nd International Conference on Infrared, Millimeter, and Terahertz Waves (IRMMW-THz), Cancun, Mexico, 2017.
- [36] C. O'Sullivan and J. A. Murphy, Field guide to terahertz sources, detectors and optics, Bellingham, Washington USA: SPIE Press, 2012.
- [37] D. Ko, T. Baek, S. Lee, S. Choi, M. Han and J. Rhee, "Transceiver module using GaAs gunn diode and schottky diode mixer for W-band FMCW radar sensor application," in 8th European Radar Conference, Manchester, UK, 2011.
- [38] A. S. Hajo, O. Yilmazoglu, F. Kiippers and A. Dadgar, "Reliability improvement of high-power THz GaN Gunn sources for active imaging systems," in 43rd International Conference on Infrared, Millimeter, and Terahertz Waves (IRMMW-THz), Nagoya, Japan, 2018.

- [39] Y. Fukasawa, K. Kawaguchi, T. Yoshida, T. Sugiyama and A. Nakagawa, "76-GHz planar Gunn VCO with low oscillation-frequency drift of 1.5 MHz/°C," in 2007 Asia-Pacific Microwave Conference, Bangkok, 2007.
- [40] H. Eisele and G. I. Haddad, "Efficient power combining with D-band (110-170 GHz) InP Gunn devices in fundamental-mode operation," *IEEE Microwave and Guided Wave Letters*, vol. 5, no. 1, pp. 24-26, 1998.
- [41] H. Eisele and R. Kamoua, "High-performance oscillators and power combiners with InP Gunn devices at 260-330 GHz," *IEEE Microwave and Wireless Components Letters*, vol. 16, no. 5, pp. 284-286, 2006.
- [42] K. Chang, W. Thrower and G. Hayashibara, "Millimeter-wave silicon IMPATT sources and combiners for the 110-260 GHz range," *IEEE Transactions on Microwave Theory and Techniques*, vol. 29, pp. 1278-1284, 1981.
- [43] W. Zhang, E. Kasper and J. Schulze, "An 82-GHz 14.6-mW output power silicon impact ionization avalanche transit time transmitter with monolithically integrated coplanar waveguide patch antenna," *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 1, pp. 308-317, 2019.
- [44] J. Nishizawa, P. Plotka, H. Makabe and T. Kurabayashi, "GaAs TUNNETT diodes oscillating at 430-655 GHz in CW fundamental model," *IEEE Microwave* and Wireless Components Letters, vol. 15, pp. 597-599, 2005.
- [45] J. Nishizawa, P. Potka, T. Kurabayashi and H. Makabe, "706-GHz GaAs CW fundamental-mode TUNNETT diodes fabricated with molecular layer epitaxy," *Physica Status Solidi(c)*, vol. 5, no. 9, pp. 2802-2804, 2008.
- [46] F. Maiwald, S. Martin, J. Bruston, A. Mestrini, T. Crawford and P. Siegel, "2.7 THz waveguide tripler using monolithic membrane diodes," *IEEE MTT-S International Microwave Symposium Digest*, vol. 3, pp. 1637-1640, 2001.
- [47] N. M. Burford and M. O. El-Shenawee, "Review of terahertz photoconductive antenna technology," *Optical Engineering*, vol. 56, no. 1, 2017.
- [48] I. C. Mayorga, A. Schmitz, T. Klein, C. Leinz and R. Gusten, "First in-field application of a full photonic local oscillator to terahertz astronomy," *IEEE*

*Transactions on Terahertz Science and Technology*, vol. 2, no. 4, pp. 393-399, 2012.

- [49] K. Han, T. K. Nguyen, I. Park and H. Han, "Terahertz yagi-uda antenna for high input resistance," *Journal of Infrared, Millimeter, and Terahertz Waves*, vol. 18, pp. 441-454, 2010.
- [50] R. Singh, C. Rockstuhl, C. Menzel, T. P. Meyrath, H. Giessen, M. He, F. Lederer and Zhang W., "Spiral-type terahertz antennas and the manifestation of the Mushiake principle," *Optics Express*, vol. 17, pp. 9971-9980, 2009.
- [51] K. Topalli, G. C. Trichopoulos and K. Sertel, "An indirect impedance characterization method for monolithic THz antennas using coplanar probe measurements," *IEEE Antennas Wireless Propagation Letters*, vol. 17, pp. 3-5, 2012.
- [52] K. Button, Infrared and Millimeter Waves V16: Electromagnetic Waves in Matter, Orlando, Florida: Academic Press, INC., 1986.
- [53] L. Xu, R. Lees, E. C. C. Vasconcellos, S. Zerbetto, L. Zink and K. Evenson, "Methanol and the optically pumped far-infrared laser," *IEEE Journal of Quantum Electronics*, vol. 32, no. 3, pp. 392-399, 1996.
- [54] S. Bierdon and G. P. Gallerano, "Overview of terahertz radiation sources," in 26th International Free-Electron Laser Conference, Trieste, Italy, 2004.
- [55] J. Faist, F. Capasso, D. L. Sivco, C. Sirtori, A. L. Hutchinson and A. Y. Cho, "Quantum cascade laser," *Science*, vol. 264, no. 5158, pp. 553-556, 1994.
- [56] B. Williams, S. Kumar, Q. Hu and J. Reno, "Resonant-phonon terahertz quantum-cascade laser operating at 2.1 THz," *Electronics Letters*, vol. 40, no. 7, pp. 431-433, 2004.
- [57] L. Esaki, "Solid-state physics in electronics and telecommunications," in Proceedings of the International Conference on Solid State Physics, Brussels, Academic Press (1960), 1958, p. 514.
- [58] L. Esaki, "Long journey into tunneling," *Reviews of Modern Physics*, vol. 46, p. 237, 1974.

- [59] L. Esaki, "New phenomenon in narrow germanium p-n junctions," *Physical Review Letters*, vol. 109, pp. 603-604, 1958.
- [60] P. Thompson, "The place of the tunnel diode in solid-state circuits," in 1962 IEEE International Solid-State Circuits Conference. Digest of Technical Papers,, Philadelphia, USA, 1962.
- [61] R. Tsu and L. Esaki, "Tunneling in a finite superlattice," *Applied Physics Letters*, vol. 22, no. 11, p. 562, 1973.
- [62] L. L. Chang, L. Esaki and R. Tsu, "Resonant tunneling in semiconductor double barriers," *Applied Physics Letters*, vol. 24, pp. 593-595, 1974.
- [63] C. Kim and A. Brandli, "High-frequency high-power operation of tunnel diodes," *IRE Transactions on Circuit Theory*, vol. 8, no. 4, pp. 416-425, 1961.
- [64] H. Mizuta and T. Tanaoue, The Physics and Applications of Resonant Tunneling Diodes, Cambridge: Cambridge University Press, 1995.
- [65] C. I. Huang, M. J. Paulus, C. A. Bozada, S. C. Dudley, K. R. Evans, C. E. Stutz, R. L. Jones and M. E. Cheney, "AlGaAs/GaAs double barrier diodes with high peak-to-valley current ratio," *Applied Physics Letters*, vol. 51, no. 2, pp. 121-123, 1987.
- [66] T. J. Shewchuk, P. C. Chapin, P. D. Coleman, W. Kopp, R. Fisher and H. Morko, "Resonant tunneling oscillations in a GaAs/AlxGa1-xAs heterostructure at room temperature," *Applied Physics Letters*, vol. 46, no. 5, pp. 508-510, 1985.
- [67] J. Wang, L. Wang, C. Li, B. Romeira and E. Wasige, "28 GHz MMIC resonant tunneling diode oscillator of around 1mW output power," *Electronics Letters*, vol. 49, pp. 816-818, 2013.
- [68] T. Inata, S. Muto, Y. Nakata, T. Fujiii, H. Ohnishi and S. Hiyamizu, "Excellent negative differential resistance of InAlAs/InGaAs resonant tunneling barrier structures grown by MBE," *Japanse Journal of Applied Physics*, vol. 25, no. 12, 1986.
- [69] J. M. L. Figueiredo, Optoelectronic properties of resonant tunnelling diodes, University of Glasgow: Ph.D. dissertation, 2000.

- [70] T. J. Schewchuk, P. C. Chapin, P. D. Coleman, W. Kopp, R. Fischer and H. Morkoc, "Resonant tunneling oscillations in a GaAs-AlGaAs heterostructure at room temperature," *Applied Physics Letters*, vol. 46, no. 5, pp. 508-510, 1985.
- [71] E. R. Brown and C. D. Parker, "Resonant tunnel diodes as submillimetre-wave sources," *Philosophical Transactions of the Royal Society of London. Series A: Mathematical, Physical and Engineering Sciences*, vol. 354, no. 1717, pp. 2365-2381, 1996.
- [72] C. A. Mead and W. G. Spitzer, "Fermi level position at metal-semiconductor interfaces," *Physical Review*, vol. 134, no. 3A, pp. 713-716, 1964.
- [73] M. O. Manasreh, Antimonide-Related Strained-Layer Heterostructures (Optoelectronic Properties of Semiconductors & Superlattices), Boca Raton, USA: CRC Press, 1997.
- [74] T. P. E. Broekaert, W. Lee and C. G. Fonstad, "Pseudomorphic In0.53Ga0.47As/AlAs/InAs resonant tunneling diodes with peak-to-valley current ratios of 30 at room temperature," *Applied Physics Letters*, vol. 53, no. 16, pp. 1545-1547, 1988.
- [75] K. Ismail, B. S. Meyerson and P. J. Wang, "Electron resonant tunneling in Si/SiGe double barrier diodes," *Applied Physics Letters*, vol. 59, no. 8, pp. 973-975, 1991.
- [76] D. Paul, P. See, I. Zozoulenko, K. Berggren, B. Hollander, S. Mantl, N. Griffin,
  B. Coonan, G. Redmond and G. Crean, "N-type Si/SiGe resonant tunneling diodes," *Materials Science and Engineering: B*, vol. 89, no. 1, pp. 26-29, 2002.
- [77] S. Wei and H. C. Lin, "Multiple peak resonant tunneling diode for multi-valued memory," in *Proceedings of the Twenty-First International Symposium on Multiple-Valued Logic*, Victoria, Canada, 1991.
- [78] D. S. Liang, K. J. Gan, J. J. Lu, C. C. Tai, C. S. Tsai, G. H. Lan and Y. H. Chen, "Multiple-valued memory design by standard BiCMOS technique," in WRI World Congress on Computer Science and Information Engineering, Los Angeles, USA, 2009.

- [79] Y. C. Kao, A. C. Seabaugh and H. T. Yuan, "Vertical integration of structured resonant tunneling diodes on InP for multi-valued memory applications," in LEOS 1992 Summer Topical Meeting Digest on Broadband Analog and Digital Optoelectronics, Optical Multiple Access Networks, Integrated Optoelectronics, and Smart Pixels, Newport, USA, 1992.
- [80] T. C. Sollner, W. D. Goodhue, P. Tannenwald and D. Peck, "Quantum well oscillators," *Applied Physics Letters*, vol. 45, no. 12, pp. 1319-1321, 1984.
- [81] E. R. Brown, T. C. Sollner, C. D. Parker, W. D. Goodhue and C. L. Chen, "Oscillations up to 420 GHz in GaAs/AlAs resonant tunneling diodes," *Applied Physics Letters*, vol. 55, no. 17, p. 17, 1989.
- [82] E. R. Brown, C. D. Parker, K. M. Molvar and K. D. Stephan, "A quasioptically stabilized resonant tunneling diode for the millimetre and submillimetre-wave regions," *IEEE Transactions on Microwave Theory and Techniques*, vol. 40, no. 5, pp. 846-850, 1992.
- [83] E. R. Brown, C. D. Parker, A. R. Calawa, M. J. Manfra and K. M. Molvar, "A quasioptical resonant-tunneling-diode oscillator operating above 200 GHz," *IEEE Transactions on Microwave Theory and Techniques*, vol. 41, no. 4, pp. 720-722, 1993.
- [84] E. R. Brown, J. R. Soderstrom, C. D. Parker, L. J. Mahoney, K. M. Molvar and T. C. McGill, "Oscillations up to 712 GHz in InAs/AlSb resonant tunneling diodes," *Applied Physics Letters*, vol. 58, no. 20, pp. 2291-2293, 1991.
- [85] R. P. Smith, S. T. Alien, M. Reddy, S. C. Martin, J. Liu, R. E. Muller and M. J. Rodwell, "0.1 um Schottky-collector AlAs/GaAs resonant tunneling diodes," *IEEE Electron Device Letters*, vol. 15, no. 8, pp. 295-297, 1994.
- [86] M. Reddy, S. C. Martin, A. C. Molnar, R. E. Muller, R. P. Smith, P. H. Siegel, Mondry M. J., M. J. Rodwell, H. Kroemer and S. Allen, "Monolithic Schottkycollector resonant tunnel diode oscillator arrays to 650 GHz," *IEEE Electron Device Letters*, vol. 18, no. 5, pp. 218-221, 1997.

- [87] N. Orihashi, S. Suzuki and M. Asada, "One THz harmonic oscillation of resonant tunneling diodes," *Applied Physics Letters*, vol. 87, no. 23, p. 233501, 2005.
- [88] S. Suzuki, M. Asada, A. Teranishi, H. Sugiyama and H. Yokoyama, "Fundamental oscillation of up to 831 GHz in GaInAs/AlAs resonant tunneling diode," *Applied Physics Express*, vol. 2, no. 054501, 2009.
- [89] S. Suzuki, M. Asada, A. Teranishi, H. Sugiyama and H. Yokoyama, "Fundamental oscillation of resonant tunneling diodes above 1 THz at room temperature,"," *Applied Physics Letters*, vol. 97, no. 242102, pp. 1-3, 2010.
- [90] M. Feiginov, C. Sydlo, O. Cojocari and P. Meissner, "Resonant tnneling diode oscillators operating at frequencies above 1.1 THz," *Applied Physics Letters*, vol. 99, no. 23, p. 233506, 2011.
- [91] S. Suzuki and M. Asada, "Coherent power combination in highly integrated resonant tunneling oscillators with slot antennas," *Journal of Applied Physics*, vol. 46, no. 46, pp. 1108-1110, 2007.
- [92] A. Al-Khalidi, J. Wang and E. Wasige, "Compact J-band oscillators with 1 mW RF output power and over 110 GHz modulation bandwidth," in 3rd International Conference on Infrared, Milimeter and Terahertz Waves (IRMWW-THz 2018), Nagaya. Japan, 2018.
- [93] J. Wang, A. Al-khalidi, K. Alharbi, A. Ofiare, E. Wasige and J. Figueiredo, "High performance resonant tunneling diode oscillators as terahertz sources," in 2016 46th European Microwave Conference (EuMC), London, UK, 2016.
- [94] S. S, M. Shiraishi, H. Shibayama and M. Asada, "High-power operation of terahertz oscillators with resonant tunneling diodes using impedance-matched antennas and array configuration," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 19, no. 1, p. 8500108, 2013.

# Chapter 2. RTD MMIC Fabrication Technology

# **2.1 Introduction**

This chapter outlines the main fabrication processes required to realize an RTD-based circuit using monolithic microwave/millimetre-wave integrated circuit (MMIC) technology. The fabrication was done at the James Watt Nanofabrication Centre (JWNC) at the University of Glasgow. Cleanrooms have a classification based on the number of particles permitted per volume of air, as such the JWNC has rooms ranging from class 10,000 down to class 10, that greatly reduce the possibility of failure due to dust particles. The RTD layer structure is specified in Section 2.2 followed by an explanation to the different MMIC fabrication processes used in this project that include:

- Lithography
- Wet / Dry Etching
- Metal Deposition
- Lift-off process

RTD devices with mesa sizes ranging from  $3 \times 3 \mu m^2$  up to  $30 \times 30 \mu m^2$  were fabricated during this project by optical lithography due to the achievable resolution that meets the requirements, low cost, ease of access and high throughput. In total there are 9 lithography steps required to fabricate an RTD oscillator, almost all of them requiring accurate positioning relative to each other to ensure the proper alignment. For a  $3 \times 3$  $\mu m^2$  RTD device the vertical interconnect access (VIA) size is  $1.5 \times 1.5 \mu m^2$  which results in an alignment tolerance of 0.75  $\mu m$ . Misalignment will result in a failing device and a failed fabrication run. Wet etching process is used to pattern the III-V semiconductor material, while dry etching is used to open a contact VIA over a polymer. Finally, the lift-off process ensures that the pattern placed in lithography comes out, leaving a clean metallisation on the sample.

## 2.2 RTD Device Fabrication Flow

### 2.2.1 Layer Structure

The project involved multiple epilayer structure designs, one such design is detailed below in Table 2.1. The wafer was grown by molecular beam epitaxy (MBE) by IQE Ltd on a semi-insulating InP substrate. The RTD device consists of (arranged from the top layer to the bottom layer): highly n-type doped contact layers (In<sub>0.53</sub>Ga<sub>0.47</sub>As) used as an interface layer for titanium/palladium/gold (Ti/Pd/Au) metallisation in the realization of ohmic contacts, 50 nm lightly doped spacer layer (In<sub>0.53</sub>Ga<sub>0.47</sub>As), 2 nm un-doped spacer layer (In<sub>0.53</sub>Ga<sub>0.47</sub>As), 1.1 nm wide band gap material barrier layer (AlAs;  $E_g = 2.15 \text{ eV}$ ), 1.1 nm un-doped narrow band gap material quantum well layer  $(In_{0.53}Ga_{0.47}As; E_g = 0.71 \text{ eV}), 1.4 \text{ nm}$  un-doped narrow band gap sub-well layer (InAs;  $E_g = 0.36 \text{ eV}$ ) used to lower the resonant energy level inside the quantum well [1] [2], that help reduce the peak voltage, 1.1 nm un-doped narrow band gap material quantum well (In<sub>0.53</sub>Ga<sub>0.47</sub>As), 1.1 nm un-doped wide band gap material barrier (AlAs), 2 nm un-doped spacer layer (In<sub>0.53</sub>Ga<sub>0.47</sub>As), 50 nm lightly doped spacer layer, a 20 nm highly doped etch stop layer (InP), a highly doped contact layer (In<sub>0.53</sub>Ga<sub>0.47</sub>As). The lightly doped spacer layers are used to increase the depletion region width, thereby decreasing the diode's capacitance, making the device suitable for high-frequency operation. It also has been reported that the peak-to-valley current ratio (PVCR) increases with increasing spacer layer thickness, while the peak current density decreases [3]. The un doped spacer layers help reduce the diffusion of dopant impurities to the double quantum well structure. Finally, the InP etch stop layer is used to control the etch depth during wet etching as a different chemical composition is needed to etch the InP layer.

The complete epitaxial layer structure was adapted from [4]. It provides a large peak to valley current ratio (PVCR) and low peak voltage, which are essential device characteristics for high performance oscillator design.

Lover	Layer Structure #1 Wafer China Wafer					
Layer – No.	Thickness (A)	Composition	Doping	Description		
1	450	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Collector		
2	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E18: Si	Sub-Collector		
3	500	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer		
4	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer		
5	11	AlAs	Un-doped	Barrier		
6	11	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Well		
7	14	InAs	Un-doped	Sub-Well		
8	11	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Well		
9	11	AlAs	Un-doped	Barrier		
10	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer		
11	500	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer		
12	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E18: Si	Sub-Emitter		
13	200	InP	1E19: Si	Etch-Stop		
14	4000	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Emitter		
15	2000	InP	Un-doped	Buffer		
		SI: InP		Substrate		

 Table 2.1: RTD epi-layer design structure used in this project.

## 2.2.2 Sample Preparation

The first step is sample preparation and cleaning, where the 3-inch wafer is cleaved into smaller samples of approximately 12 mm x 12 mm dimensions. Before the cleave process the wafer is covered with a layer of photoresist to protect from scratches and contamination. The cleaning process follows with an ultrasonic bath in acetone ((CH<sub>3</sub>)<sub>2</sub>CO), methanol (CH<sub>3</sub>OH) followed by isopropyl alcohol (C<sub>3</sub>H<sub>8</sub>O). Acetone is a good polar solvent and the primary chemical used to remove organic residue; however, the high evaporation rate of the chemical can cause re-deposition of the remnants. Therefore methanol, another polar solvent is used to remove the contaminants related to acetone residue. Isopropanol is the last cleaning agent used and is most suitable for dissolving non-polar contaminants that remain on the sample.

Another important step before starting the fabrication process is to determine the correct side of the InP sample, since both sides are polished and are visually hard to distinguish. Current-voltage (I-V) measurements are therefore carried out on both sides, the measured resistance of the semi-insulating substrate is in the order of

giga-ohms, while that of the highly doped contact layer is usually in the tens of ohms range. An *I-V* curve of the InGaAs layer and InP layer is shown in Fig. 2.1.



*Figure 2.1:* Measured current for the top InGaAs layer (left axis) and substrate InP layer (right axis) for the RTD-EPI layer structure presented in Table 2.1.

#### 2.2.3 Fabrication Process Flow

The fabrication flow diagram needed to realize an RTD device is shown in Figure 2.2. Fabrication starts with the deposition of the top metal contacts on the emitter / collector layer as shown in Fig. 2.2 (a). A number of reviews of ohmic contacts on III - V compound semiconductors are available from [5]-[7]. The metal scheme used is Ti/Pd/Au (20/30/150 nm), which has a reported specific contact resistance  $\rho_c$  of  $0.73 \pm 0.44\Omega \,\mu m^2$  [5]. After metal deposition the sample is placed in a 1:1 solution of acetone and photoresist stripper 1605. The next step, as shown in Fig. 2.2 (b), is to etch the epitaxial layers down to the bottom contact / emitter layer using an acid solvent (H<sub>3</sub>PO<sub>4</sub>:H<sub>2</sub>O<sub>2</sub>:H<sub>2</sub>O 1:1:38) which has an etching rate of about 100 nm/min. The etch stop layer is removed using a different acid solvent (HCl: $H_2O_2 = 1:4$ ) for about 30 seconds. After wet etching another contact metal is deposited on the bottom collector / emitter layer as shown in Fig. 2.2 (c). A second wet etch process follows, illustrated in Fig. 2.2 (d), that exposes the InP substrate and isolates the individual RTD devices. Again, a combination of acid solutions as described previously are used to etch the different layer compositions. Thereafter, the next step is to deposit the passivation layer (polyimide PI-2545) and a VIA is opened by dry etching over the polymer once it has cured as shown in Fig. 2.2 (e). The final metal contacts (Ti/Au 20/380 nm) are deposited to provide a landing surface large enough for the probes and completes the RTD device fabrication as shown in Fig.2.2 (f).



**Figure 2.2:** RTD device fabrication steps: (a) top contact metal deposition, (b) wet etch to the bottom emitter/collector layer, (c) bottom contact metal deposition, (d) wet etch to substrate, (e) PI-2545 deposition and via opening using dry etch, (f) bond pad metal deposition

For oscillator circuits additional passive components such as a thin-film resistor and a metal-insulator-metal (MIM) capacitor are required and extra fabrication steps are needed in order to accomplish this. These will be described in Chapter 3.

# **2.3 Fabrication Processes**

# 2.3.1 Lithography

Lithography is the process used to transfer patterns from a mask onto the sample surface. The most common lithography techniques are photolithography using ultraviolet (UV) light for exposure, electron beam (e-beam) lithography, x-ray lithography and extreme ultraviolet lithography. After being cleaned, the sample is heated on a hot-plate at 120°C for 5 minutes, in order to remove any moisture present on the wafer surface. The sample is then attached to a spinner inside a laminar air flow (LAF) cabinet by means of a vacuum and covered in resist. It is then rotated at high speed in order to spread the resist by means of centrifugal force. The thickness of the resist film is dependent on the rotational speed and fluid viscosity. After spinning, the resistcoated sample is then pre-baked typically at 115°C for 1 min (S1805) and is ready for patterning. The lithography tools used in this project are the Karl Suss MA6 mask aligner (photolithography) and the Vistec VB6-UHR-EWF (e-beam lithography). The pattern is then developed using a solvent to selectively remove the exposed areas of the thin film. The next step is to remove any residual resist using a barrel asher, that uses an oxygen powered plasma. The oxygen ions etch the residual resist at a 10nm/min rate using 120W RF power, removing any residue resist. The sample is ready for deposition or etching. Fig. 2.3 shows an illustration of the photolithography process.



Figure 2.3: Illustration of the photolithography process. The substrate is first coated with a thin layer of positive photoresist. Using the Karl Suss MA6 mask aligner, the photomask is placed in contact with the sample. After exposure the sample is developed, whereby the resist that was not shielded by the photomask is removed leaving an accurate reproduction of the mask pattern.

#### **Electron-beam Lithography**

In electron-beam lithography a focused beam of electrons is directed at the sample surface covered with suitable resist (PMMA, ZEP), sensitive to the electron beam. Due to the bombardment of electrons, the molecular structure and solubility of the resist film are altered. The Vistec VB6-UHR-EWF tool available at the JWNC has a minimum resolution of 0.5 nm and a minimum spot size of 4nm.

In this project e-beam was used in the manufacturing of the optical masks (by Compugraphics) for the photolithography process, and for the manufacturing of single step jobs such as antennas or correcting photolithography mask errors. The desired patterns are first designed using CAD software tools such as L-Edit. An un-processed photomask, consisting of a glass plate coated on one side with a thin layer of chrome, is coated with e-beam sensitive photoresist and placed inside the e-beam tool for patterning. After patterning the mask is immersed into a developing solution where the areas exposed to the electron beam are dissolved, leaving the desired pattern. A chrome etch follows that removes the zones not protected by the resist.

Compared to photolithography, e-beam lithography has a much higher resolution and registration (how accurately successive patterns on the mask can be aligned), but the lower throughput and high cost make it unsuitable for micron-sized RTDs.

#### Photolithography

Photolithography makes use of an ultra-violet (UV) light source of wavelength between  $\lambda = 200 - 400$  nm to expose the photoresist, in a process that is similar to the e-beam lithography process described previously. When compared to e-beam, photolithography is a low cost and high throughput technique, with a resolution capability of 0.5 µm, that meets all the requirements of the manufactured micron-sized RTDs [8]. The sample and hard mask are loaded onto the mask-aligner. After alignment, the mask is placed in contact with the sample, where it blocks light in the areas covered by the chromium pattern. Once developed the patterns define the required features which will be used in the different fabrication steps such as metal/dielectric deposition and etching.

#### Photoresist

Photoresists are essential compounds used in microfabrication and consists of binders, sensitizers and solvents. The role of the sensitizer is to define the photosensitivity of the resist. When exposed to UV light, the chemical structure of the resist becomes insoluble or soluble, depending on the resist type in the developer. The solvents define the viscosity of the resist, which allows the resist to be spun and to form thin layers on the wafer surface. The binders control the thermal characteristics of the resist. There are two types of photoresists, positive and negative resist. In the positive resist, the exposed area dissolves during the developing process, while the un-exposed region remains unchanged. The same principles work in reverse for the negative resist. Both negative and positive photoresist have advantages and disadvantages. The positive photoresists present excellent resolution and can be developed in aqueous developers, however they suffer from bad adhesion and lower resistance to etching when compared to their counterpart. The negative resists offer better adhesion and resistance against etch processes at the cost of lower resolution and can only be developed in toxic organic developers.

Spin coating is the standard resist application method, where a few drops of resist are applied at the centre of a static or a slowly rotating wafer. After application the wafer is accelerated, in most cases to 4000 rpm, where the resist spreads over the wafer, leaving a uniform layer. The soft bake or pre-bake process (115° C for 1 min) that

follows evaporates the remaining solvent, hardening the resist. Spin-film definition at the edge of the sample is often poor [9], as the resist always flows over the edge, it forms a non-uniform coating. This is known as *edge bead* [10]. As such steps are carried out in order to remove the excess resist present at the edge of the sample. A cotton bud containing acetone is carefully used to clean the edge of the sample and/or the underside of the sample if resist is present there.

In this project, two positive photoresists have been used: S1805 and S1818 from MicroChem. The difference between the two photoresists are their respective spin-thickness and is shown in Figure 2.4 [11]. At 4000 rpm spin speed, the thickness measured were  $1.8 \mu m$  and  $0.5 \mu m$  for the S1818 and S1805, respectively.



Figure 2.4: S1800 series photoresist spin speed vs thickness. Adapted from [11].

The thickness of the resist must be considered during the lift-off process, used in metallisation patterning (to be described in Section 2.3.2). Thin resists layers are preferred for better resolution, however there is a minimum resist-film ratio that needs to be satisfied for a successful lift-off. As such for thin metal films (around ~200 nm) a single layer of S1805 resist was used (2.5:1 resist-film ratio). Positive photoresists usually have a slightly positive slope [9] that needs to be corrected in order to prevent side-wall deposition during metallisation. In this project, the sample was soaked in either chlorobenzene ( $C_6H_5CI$ ) or toluene ( $C_7H_8$ ) for a couple of minutes before

exposure to reduce the dissolving rate of the resist's top surface. As a result, the profile of the soaked resist will have an undercut characteristic, making the lift-off process easier. Chlorobenzene was initially used but was discontinued from the JWNC due to health concerns. An illustration of the single-layer metal lift-off process is given in Fig. 2.5.



*Figure 2.5:* Illustration of the lift-off process using toluene. (a) clean, spin and pre-bake the photoresist (b) dip in toluene for about 1-minute (c) expose and develop the photoresist, (d) metallization, and (e) metal lift-off.

When thicker metal layers are required (~400 nm) a bi-layer lift-off process was used. A lift-off resist, usually LOR-10A, is first spun onto the sample and pre-baked at 150°C for 3 mins followed by S1800 series resist deposition. After exposure and development, an overhang profile is formed due to the different dissolution rates between the LOR and S1800 resist. The undercut formed is larger than that of the toluene treated single layer lift-off and ensures an easier lift-off due to no possible side-wall deposition. An illustration of the bilayer lift-off is shown in Fig. 2.6.



*Figure 2.6:* Illustration of the bilayer lift-off process (a) clean, spin and pre-bake the LOR layer (b) spin and deposit S1800 resist (c) expose and develop the photoresist, (d) metallization, and (e) lift-off.

### 2.3.2 Metallisation

Depositing metal on the semiconductor wafer can be accomplished in several different ways, such as physical vapour deposition (PVD), sputtering or electroplating. The technique used to deposit thin metal films in this project was electron-beam physical vapor deposition (EBPVD), a variation of PVD. A Plassys MEB 550S (Plassys II and Plassys IV) system was used to deposit the titanium (Ti), palladium (Pd), molybdenum (Mo), gold (Au) and nickel – chromium (NiCr) metals used in the fabrication. Once the sample is loaded into the evaporator, the main chamber is pumped to a base pressure of  $2 \times 10^{-6}$  Torr and the desired metal scheme is selected via control software. Following the desired metal scheme, the metal crucibles inside the chamber are heated over the melting point by a beam of electrons. The heated metals have high vapor pressures and in the high vacuum, the evaporated atoms will be transported to the substrate. A mechanical shutter is used to turn on the deposition, once the required evaporation rate has been achieved (0.5 nm/s for Au).

Another important feature of the Plassys tool is the available argon gun, that is used to clean the surface of the sample by bombardment with ions. This technique de-oxidises the surface of the sample and lowers the ohmic contact resistance. The specific contact resistance was measured by transmission line model (TLM) technique and is detailed in Chapter 3.

In order to improve the quality of the contacts and their adhesion to the substrate a Rapid Thermal Annealing (RTA) process is used. The sample is loaded in a chamber pumped with nitrogen, where it is rapidly heated to 275° C for 60 s. A degradation can be observed on the metal contacts for longer anneal times and is attributed to indium out-diffusion [12] from the InGaAs layers under the metal film.

## 2.3.3 Dielectric Layer Deposition

In the fabrication process for RTD oscillators, it is necessary to deposit thin films of dielectric materials for the creation of metal-insulator-metal (MIM) capacitors. The two most common passivation materials are silicon dioxide ( $SiO_2$ ) and silicon nitride ( $Si_3N_4$ ), which was used in this project. Inductively Coupled Plasma Chemical Vapour
Deposition (ICP-CVD) is the process used to deposit the thin dielectric film, using the System 100 ICP-CVD from Oxford Instruments. The high-density plasma allows the deposition at low temperatures of the silicon nitride film with minimal hydrogen present [13]. The low deposition temperature is an essential requirement for the S1800 mask. A bi-layer lift-off process was employed as it offers better lift-off when compared to a single layer lift-off process, mainly due to the side-wall deposition of the Si<sub>3</sub>N<sub>4</sub>.

#### 2.3.4 Etching

Etching is referred to as the physical and/or chemical process that removes layers from the surface of the substrate in the areas that are not protected by hard (metal) or soft (resist) masks. This process is accomplished by either wet or dry etching, according to the requirements of the application:

- *Uniformity* is defined as the deviation in etch rate across the sample.
- *Selectivity* is the ratio between the etch rates of different materials.
- *Isotropy* is the uniformity of the etch in all directions

Knowing the etch rates of the materials that are exposed to the dry etch / wet etch solution is crucial in the manufacturing process. High selectivity is desired between the film and substrate. In wet etching the resist selectivity is high (> 50:1), however during dry etch the selectivity is low (~ 1.5:1), and care must be taken not to over-etch and remove the protective resist film. Another important requirement is the degree of isotropy, and is defined as:

$$Isotropy = 1 - \frac{R_l}{R_v} \tag{2.1}$$

where  $R_l$  represents the lateral etch rate and  $R_v$  is the vertical etch rate. An etch process that has a degree of isotropy equal to 0, meaning that there is no lateral etching is called *perfectly isotropic*, while a degree of isotropy equal to 1 is called *perfectly anisotropic*. The wet etching process is shown in Fig. 2.7.



*Figure 2.7:* Etching process. The substrate is first coated with a layer of photoresist which is used as an etch mask to protect the desired areas. A wet or dry etching is carried out. The resist is then removed.

## Wet etching

The most common techniques used in the wet etch process is immersion and spray methods. In the immersion method, the wafers are submerged in a container containing the etching solution for the required time. The advantages of using a wet etch process are:

- Low cost and fast process that does not involve any specialized equipment
- Very high selectivity, making possible the use of the etch stop layers and does not damage the resist mask.
- Low surface damage

Disadvantages include:

- Isotropic process resulting in undercut
- Highly toxic and corrosive chemicals are used
- Non-uniformity across the wafer

For the wet etching of the InGaAs, AlAs, InAs layers a solution of orthophosphoric acid ( $H_3PO_4$ ) and hydrogen peroxide ( $H_2O_2$ ), diluted in water was used ( $H_3PO_4$ : $H_2O_2$ : $H_2O$  1:1:38). The oxidising agent ( $H_2O_2$ ) creates an oxide layer at the material surface and then the acid is used to remove the oxide in a reduction reaction [9]. The etching rate is around 100 nm/min and does not etch the InP substrate. For InP a solution of hydrochloric acid (HCl) and hydrogen peroxide is used (HCl: $H_2O_2$  1:4)

and has an etch rate of 600 nm/min. Because of the isotropic nature of the process, the etch profile is rounded. Undercutting is similar to vertical etched depth and will become severe when the aspect ratio of the feature is high. This undercut is undesirable since it will cause a reduction in the designed device size and lowers the expected performance. Undercutting can be compensated by making the initial mask feature slightly larger than the desired width. Figure 2.8 shows the undercut profile of a  $3 \,\mu m \, x \, 3 \,\mu m$  fabricated device, taken with a scanning electron microscope (SEM).



*Figure 2.8: SEM picture of an RTD device showing the undercut profile caused by wet etching. The effective area of the RTD is reduced due to lateral etching.* 

#### **Dry etching**

The dry etch process is a combination of a physical process and chemical reactivity, whereby the reactive gasses are excited and ionized by RF-fields inside the chamber and bombard the material surface, where the chemical reactivity of the ion removes the material at the surface. The low-pressure in the chamber prevents the particles that are etched away from redepositing on the surface. The process provides high uniformity, etch-rate control and has high levels of anisotropy.

The dry etch process was used to open the VIA in the polyimide (PI-2545) and make a connection between the top contact of the RTD device and the bond pads. In this projected the Oxford Instruments RIE80+ tool was used for dry etching. In order to etch the polyamide layer a mixture of tetrafluoromethane/oxygen ( $CF_4/O_2$ ) was used, with an RF power of 200 W and a mass flow rate of 5/20 sccm. The measured dry etch rate is about 220 nm/min for PI-2545 and 150 nm/min for S1805 photoresist. A laser interferometer was used as an end-point detection technique [14]. The interferometer creates a coherent beam of light (constant phase and frequency) that is split by a beam splitter into two identical beams, one aimed at the sample. These travel different routes and recombine before arriving at the detector. During the etching process, the path length of the beam aimed at the sample varies causing constructive and destructive interference at the detector. Waves which are not completely in phase or out of phase will have a different intensity pattern. When the PI-2545 is completely etched, the laser will have constant intensity. Figure 2.9 shows the interferogram example obtained during the dry etch process.



*Figure 2.9:* Interferogram obtained during the dry etch process of the polyimide layer. The etch duration is approximately 5 min and 30 seconds, total etch depth is  $1.2 \mu m$  for the polyimide layer.

Another purpose of the polyamide layer was to prevent the breaking of the metal when deposited on two areas of different heights. The polyamide cushion allowed for the deposition of thin metal layers, as shown in Fig. 2.10. Without this layer, thicker electrodes would have been needed to prevent the metal from breaking.



*Figure 2.10: SEM* picture of an RTD device showing the polyamide cushion that isolates the top emitter and bottom collector metals and prevents the metal from breaking.

# 2.4 Summary

In this chapter the main aspects of the RTD MMIC fabrication process have been discussed. Photolithography was used to transfer the patterns from the mask onto the resist covered substrate. The wet etching process was used to remove the InGaAs layers as defined by the mask, while dry etching was used to create a VIA opening in the polyimide layer. Single and bi-layer lift-off process have been explained and have been successfully applied to the metallization and dielectric layer deposition procedures. The fabrication process described in this chapter is detailed in Appendix A.

# References

- F. Chevoir and B. Vinter, "Scattering-assisted tunneling in double-barrier diodes: Scattering rates and valley current," *Phys. Rev. B*, vol. 47, no. 7260, 1993.
- [2] P. Roblin, R. C. Potter and A. Fathimulla, "Interface roughness scattering in AlAs/InGaAs resonant tunneling diodes with an InAs subwell," *Jurnal of Applied Physics*, vol. 79, no. 5, pp. 2502-2508, 1996.
- [3] I. Mehdi, R. K. Mains and G. I. Haddad, "Effect of spacer layer thickness on the static characteristics of resonant," *Applied Physics Letters*, vol. 57, no. 9, pp. 899-901, 1990.
- [4] T. Broekaert, W. Lee and C. Fonstad, "Pseudomorphic In<sub>0.53</sub>Ga<sub>0.47</sub>As/AlAs/InAs resonant tunneling diodes with peak to valley current ratios of 30 at room temperature," *Applied Physics Letters*, vol. 53, no. 16, pp. 1545-1547, 1988.
- [5] A. M. Crook, E. Lind, Z. Griffith and M. J. W. Rodwell, "Low resistance, nonalloyed Ohmic contacts to InGaAs," *Applied Physics Letters*, vol. 91, no. 19, p. 192114, 2007.
- [6] A. G. Baca, F. Ren, J. C. Zolper, R. D. Briggs and S. J. Pearton, "A survey of ohmic contacts to III-V compound semiconductors," *Thin Solid Films*, Vols. 308-309, no. 1, pp. 599-606, 1997.
- [7] V. L. Rideout, "A review of the theory and technology for ohmic contacts to group III-V compound semiconductors," *Solid-State Electronics*, vol. 18, no. 6, pp. 541-550, 1975.
- [8] G. S. May and S. M. Sze, Fundamentals of Semiconductor Fabrication, New York: John Wiley & Sons, 2004.
- [9] S. Franssila, Introduction to microfabrication, New york: John Wiley & Sons, 2010.

- [10] Micro Chemicals, "Basics of microstructuring," [Online]. Available: https://www.microchemicals.com/technical\_information/spin\_coating\_photore sist.pdf.
- [11] Shipley, "Microposit S1800 series broadband resists," 2019. [Online]. Available: https://kayakuam.com/wp-content/uploads/2019/09/S1800.pdf.
- [12] I. Krylov, R. Winter, D. Ritter and M. Eizenberg, "Indium out-diffusion in Al<sub>2</sub>O<sub>3</sub>/InGaAs stacks during anneal at different ambient conditions," *Applied Physics Letters*, vol. 104, no. 24, p. 243504, 2014.
- [13] J. Yota, J. Hander and A. Saleh, "A comparative study on inductively-coupled plasma high-density plasma, plasmaenhanced, and low pressure chemical vapor deposition silicon nitride films," *Journal of Vacuum Science & Technology A*, vol. 18, no. 2, pp. 372-376, 2000.
- [14] W. Wei, L. Zhongwen, W. Wu and G. Yungui, "Optical interferometry endpoint detection for plasma etching," in 8th International Conference on Electronic Measurement and Instruments, Xi'an, China, 2007.

# **Chapter 3. Design and Characterization of Passive Components**

# **3.1 Introduction**

The realization of RTD oscillators requires several additional passive components according to the specific design. These include the metal-insulator-metal (MIM) capacitors, the coplanar waveguide (CPW) transmission line, shorted CPW inductors and resistors. In this chapter, the design procedures for these different components are described. Theoretical values of these components are calculated and validated with experimental results. The realization and characterization of Ohmic contacts is also investigated as it plays an important role in the performance of an RTD oscillator.

# **3.2 Design of Passive Components**

## 3.2.1Thin-film and Semiconductor Resistors

In this project thin-film or semiconductor resistors were used to suppresses the bias oscillations present in the NDR region of RTDs. Several factors are considered when designing integrated resistors [1]:

- Sheet resistance
- Power density handling capability
- Accuracy and reproducibility
- Temperature coefficient of resistance (TCR)
- Frequency response

Two different materials were used in this project nickel – chromium (NiCr) and semiconductor (InGaAs) resistors. The advantages of using NiCr resistors are that they show a low TCR (77 ppm / °C) [2]-[4], and are fairly easy to fabricate, however several manufacturing runs suffered from low yield, due to poor adhesion of the NiCr to the InP substrate. A more reliable process was developed using the bottom highly doped

InGaAs layer of the RTD epitaxial structure as a semiconductor resistor. The geometry of a fabricated thin-film NiCr resistors test structure is shown in Fig. 3.1.



*Figure 3.1:* NiCr thin-film resistor design test structure: (a) top view, (b) cross section view.

The resistance value *R* can be calculated using Eq. 3.1 [5], where  $\rho$  is the material bulk resistivity (1.65 × 10<sup>-6</sup>  $\Omega$  m), *l* is the length of the resistor in the direction of current flow, and *w* and *t* are the width and thickness of the film materials respectively.

$$R = \rho \frac{l}{wt} \tag{3.1}$$

With thin film resistors, sheet resistance  $R_{sh}$  is a common property used when designing thin film resistors of uniform thickness. The unit of measure for sheet resistance is  $\Omega/\Box$ . For a 33 nm thick nichrome film, the sheet resistance of NiCr is around 50  $\Omega/\Box$  [6][7]. For the InGaAs semiconductor resistor with high doping concentration (n = 3×10<sup>19</sup> cm<sup>-3</sup>) and 400 nm thickness the sheet resistance is 5  $\Omega/\Box$ , obtained through experimental measurements. The total resistance can be calculated using Eq. 3.2, where the sheet resistance  $R_{sh} = \frac{\rho}{t}$ .

$$R = R_{sh} \frac{l}{w} \tag{3.2}$$

#### **3.2.2 Metal-Insulator-Metal Capacitors**

Planar metal-insulator-metal capacitors (MIM) are a key component in RF circuits due to their low parasitic capacitance and low resistivity electrodes [8]. The MIM capacitor consists of a thin dielectric layer placed in-between two metal plates. The layout of a MIM capacitor is show in Fig. 4.2.



*Figure 3.2: Metal-insulator-metal capacitor design: (a) top view, (b) cross section view.* 

For this project silicon nitride (Si<sub>3</sub>N<sub>4</sub>) was used as the dielectric material. The capacitance value can be calculated using Eq 3.3, where  $\varepsilon_0$  is the vacuum permittivity ( $\varepsilon_0 = 8.85 \times 10^{-12}$  F/m),  $\varepsilon_r$  is the dielectric constant of the Si<sub>3</sub>N<sub>4</sub> ( $\varepsilon_r = 6.8$ ), *A* is the area, and *d* the thickness of the dielectric material. The thickness used in this project was 75nm, which corresponds to 0.8 fF/µm<sup>2</sup> capacitance.

$$C = \frac{\varepsilon_0 \varepsilon_r A}{d} \tag{3.3}$$

During this project a parallel and a series placed capacitor were designed and used in the oscillator circuits. The parallel capacitor ( $C_E$ ) is used as an RF short circuit to prevent RF power being dissipated across the stabilising resistor ( $R_E$ ), and for the shorted transmission line structure used in realizing the resonating inductors (to be described in Section 3.2.3). A series capacitor was used as a DC block, to protect the spectrum analyser from the DC bias current applied during characterization. The impedance value for the parallel and series capacitor was chosen to be less than 0.1  $\Omega$ at the oscillation frequency, i.e. ( $2\pi fC$ )<sup>-1</sup> < 0.1  $\Omega$ .

### **3.2.3 Coplanar Waveguide**

The coplanar waveguide (CPW) was first proposed by C.P. Wen in 1969 and consists of a dielectric substrate with conductors placed on the top surface [9]. The conductors form a center strip line separated by a narrow gap, followed by ground planes on either side. The dimensions of the center strip line, the gap width, and the characteristics of the dielectric substrate determine the effective dielectric constant and characteristic impedance ( $Z_0$ ). The uniplanar construction of the CPW line, in which all conductors are placed on the same side of the substrate simplifies the manufacturing process, allows for shunt / series surface mounting of components and permits fast on-wafer characterization of the devices [10]. This structure demonstrates better dispersion characteristics than microstrip and does not require any vias for grounding. Packaging can be a problem for this type of structure, as the dielectric substrate may come into close proximity to other materials, altering the transmission line characteristics [11]. A CPW structure is shown in Fig. 3.3, where *w*, *s*, *g*, *l*, and *t* are the CPW signal line width, the gap between the center conductor line and the ground plane, the ground plane width, the CPW length and the thickness of the CPW lines, respectively.



**Figure 3.3:** Coplanar waveguide structure on a substrate with dielectric constant  $\varepsilon_r$  and thickness h. The centre conductor line width is denoted as w, the width of the ground planes is g, the gap between the signal line and ground plane is s, the thickness of the conductor is t and the lengths are denoted with l. (a) top view, (b) cross section view.

In this project the LineCalc tool available within Agilent's Advanced Design System software was used to calculate the dimensions (*w* and *s*) for a 50  $\Omega$  transmission line

on InP substrate with the dielectric constant  $\varepsilon_r = 12.5$  [12] and thickness  $h = 630 \,\mu\text{m}$ . The required signal line width and gap spacing was 60  $\mu\text{m}$  and 40  $\mu\text{m}$ , respectively.

In order to reduce the attenuation of the signal, the thickness of the conductor was chosen to be three times the skin depth  $\delta$  [10]. Equation 3.4 is used to calculate the skin depth for a given frequency, where  $\rho$  is the conductor resistivity (2.44 × 10<sup>-8</sup>  $\Omega$  m for gold),  $\mu_0$  is the vacuum permeability (4 $\pi$ ×10<sup>-7</sup> H/m),  $\mu_r$  the relative permeability ( $\mu_r = 1$ ) and f the lowest frequency of interest. For the targeted 30 GHz oscillation frequency the thickness required is 1.3 µm.

$$\delta = \sqrt{\frac{\rho}{\mu_0 \mu_r \pi f}} \tag{3.4}$$

For a transmission line of length *l*, the input impedance  $Z_{in}$  is calculated by Eq. 4.5 [10], where  $Z_0$  is the characteristic impedance,  $Z_L$  denotes the load impedance and  $\beta$  represents the phase constant. The effective dielectric constant ( $\varepsilon_{eff}$ ) for a CPW line can be calculated using Eq 3.7 [13].

$$Z_{in} = Z_0 \frac{Z_L + jZ_0 tan(\beta l)}{Z_0 + jZ_L tan(\beta l)}$$
(3.5)

$$\beta = \frac{2\pi f \sqrt{\varepsilon_{eff}}}{c_0} \tag{3.6}$$

$$\varepsilon_{eff} = \frac{\varepsilon_r + 1}{2} \tag{3.7}$$

The CPW structures were also used to realize the different resonating inductor values required in the oscillator circuits. Fig. 3.4 (a) illustrates a shorted CPW line. In this configuration, the current flows through the metallisation at the end of the slots, storing magnetic energy behind the termination. The result is an inductive reactance when the electrical length of the line is less than 90 degrees [14].



*Figure 3.4:* Schematic of a coplanar waveguide stub: (a) simple short circuit stub, (b) MIM-short stub 3D model.

For a short circuit CPW line structure (i.e.  $Z_L = 0$ ) Eq 3.5 can be rewritten as

$$j\omega L = jZ_0 tan(\beta l) \tag{3.8}$$

where *L* is the line inductance and  $\omega$  represents the angular frequency. Equation 3.8 can be rewritten in terms of the CPW line length *l* as shown in Eq. 3.9.

$$l = \frac{1}{\beta} atan(\frac{\omega L}{Z_0})$$
(3.9)

The simple short configuration shown in Fig. 3.4 (a) has the disadvantage that it does not allow for different bias potentials on the centre and ground conductors. This restriction can be circumvented by placing a large MIM capacitor placed between the centre strip and the ground conductor. The electrical behaviour of this configuration can be described by an inductance connected in series with the MIM capacitance [15]. As stated in the previous section the value for this MIM capacitor needs to be large enough as to not deteriorate the short-circuit characteristics of the stub at the oscillation frequency .

# **3.3 Characterization of Passive Components**

This section covers the characterization of the individual passive components described in the previous section. The CPW and MIM capacitors were characterized

with the help of scattering parameter (S-parameter) measurements using a calibrated vector network analyzer (VNA).

The VNA used in this project is the N5250C VNA from Agilent technologies, which has an operational frequency range from 10 MHz to 110 GHz. The VNA was calibrated using precise reference impedance standards in order to correct the systematic errors found in the system, i.e. frequency response error caused by reflection and transmission tracking with the device under test (DUT), directivity and crosstalk related to signal leakage, source mis-match, and load mis-match [16]. The calibration routine used is Short-Open-Load-Through (SOLT) method, which uses a short circuit, an open circuit, a precisely defined 50  $\Omega$  load and a through connection to enhance the accuracy of the measurement.

### 3.3.1 Coplanar Waveguide

In order to characterize the designed CPW lines, test structures were fabricated and characterized by 2-port S-parameter measurements. The CPW line was designed for a characteristic impedance  $Z_0 = 50 \ \Omega$  on a semi-insulating InP substrate with dielectric constant  $\varepsilon_r = 12.5$ . The width of the centre conductor line is 60 µm and the gap is 40 µm. The ground plane was designed to be at least 3 times the centre strip line width , in order to minimize the radiation losses [17] [18]. A micrograph of the fabricated CPW test structure is shown in Fig. 3.5.



**Figure 3.5:** Micrograph of CPW test structure. The centre strip width is 60  $\mu$ m, the gap width is 40  $\mu$ m, while the ground plane width is 200  $\mu$ m. The CPW test structure length is 500  $\mu$ m.

The measured return loss ( $S_{11}$  and  $S_{22}$ ) are plotted in Fig. 3.6 while the measured insertion loss ( $S_{12}$  and  $S_{21}$ ) are plotted in Fig. 3.7. The return loss is 25 dB at 30 GHz, which indicates that very little power of the transmitted signal is reflected back to the same port. The insertion loss is 0.1 dB, which indicates that the CPW line transmits the signal from one port to the other without losses. Noise is introduced around 67 GHz by the change from coaxial source (1.85 mm) to the waveguide source (WR-10) in the extenders, which provide the extended broadband range of the equipment from 10 MHz to 110 GHz.



*Figure 3.6:* Measured S-parameters  $(S_{11}, S_{22})$  of the CPW line test structure. The return loss indicates that most of the power transmitted is not reflected back to the same port.



*Figure 3.7: Measured S-parameters*  $(S_{12}, S_{21})$  *of the CPW line test structure. The insertion loss indicates that most of the power is transmitted to the second port.* 

#### **3.3.2 MIM Capacitor**

Metal-insulator-metal (MIM) capacitors are widely used in RF circuits due to their low leakage-current, high capacitance value and ease of integration. For this project two types of MIM capacitors, for in-series and in-parallel configurations were used. The series capacitor is used for DC blocking or RF decoupling circuits, while the parallel capacitor has been used for stabilisation and as an RF short in the CPW stub configuration. The performance of the capacitors was investigated using 2 port Sparameter measurements. The capacitors were designed to operate around 30 GHz oscillation frequency. The measured return loss  $(S_{11})$  and insertion loss  $(S_{12})$  for the parallel capacitor and the series capacitor are shown in Fig. 3.8 and Fig. 3.9, respectively. Fig. 3.10 shows the Smith chart for both capacitors. The S-parameters of the parallel capacitor show that it acts as a short circuit where the insertion loss is below -20 dB. The input impedance for the parallel capacitor at 30 GHz is 1 + j0.15 $\Omega$ . The series capacitor acts as a short circuit for frequency ranges from 10 GHz to 110 GHz with an insertion loss of about -0.1 dB. The input impedance for the series capacitor at 30 GHz is  $50.15 + j0.65 \Omega$ . At higher frequencies the parasitic elements (inductance and resistance) become more dominant and start to degrade the capacitor performance.



*Figure 3.8: Measured S-parameter*  $(S_{11}, S_{12})$  *of the parallel capacitor test structure.* 



*Figure 3.9:* Measured S-parameter  $(S_{11}, S_{22})$  of the series capacitor test structure.



*Figure 3.10:* Measured S-parameters  $(S_{11})$  of the series and parallel capacitor test structures plotted on a Smith chart.

## 3.3.3 Thin-film and Semiconductor Resistors

As discussed in the previous section, on-chip resistors were used in the RTD devices and oscillators for stabilisation. NiCr thin-film resistors and InGaAs semiconductor resistors were both used in this project. The measured sheet resistance of a thin film 33 nm thick NiCr resistor is 50  $\Omega/\Box$ , while the InGaAs semiconductor resistor has a sheet resistance of 5  $\Omega/\Box$  when the thickness is 400 nm. A micrograph of a fabricated NiCr resistor can be seen in Fig. 3.11, while the designed and measured values of the resistors can be seen in Table 3.1. Small discrepancy was observed between the theoretical and measured resistance values, this was attributed to the non-uniform deposition thickness and contaminated interface between the material and contact pads.



*Figure 3.11: SEM* picture of a fabricated NiCr resistor test structure. Two resistors are placed in parallel to achieve the designed value.

Туре	${ m R}_{ m sh}$ $(\Omega / \Box)$	Designed Value ( $\Omega$ )	Measured Value $(\Omega)$	Divergence (%)
NiCr	50	10	11.3	13
10101		20	21.3	6.5
InGaAs	5	12	11.4	5
IIIGaAs	3	15	13.9	7.33

## **3.4 Ohmic Contacts on InGaAs**

Very high frequency devices, including the RTD, require a low metal-semiconductor contact resistance in order to reduce the resistive losses, minimize the self-heating effects and the resistive capacitive delay, that imposes a limit on the maximum oscillation frequency of the device [19][20]. Linear transfer length measurements (LTLM) test structures were fabricated in order to accurately measure the contact performance [19][21]. The TLM measurement structures consists of a series of metal-semiconductor contact pads separated by increasing distances as shown in Fig. 3.12. The resistivity is measured by 4-point probe measurement technique [22][23], where a constant current is applied between two probes (named force) and the voltage is measured across the two other probes (sense). The sense probes have a high resistivity, and thus almost no current flows through the sense cable, allowing for an accurate measurement of the voltage drop.



*Figure 3.12:* LTLM structure with various separation distances. Four probes are placed on two adjacent contact pads, where a constant current is applied, and the voltage drop measured.

The LTLM test-structures were fabricated on an InGaAs wafer with a heavily doped 45 nm contact layer (n =  $3 \times 10^{19}$  cm<sup>-3</sup>). The test structures were fabricated by patterning the sample using photolithography, followed by Ti/Pd/Au (20/30/150 nm) metal deposition and lift-off process. Each pattern was then electrically isolated by photolithography and wet etching to the semi-insulating InP substrate. The nominal pad dimensions are 30 µm × 40 µm and the nominal gap widths between the contacts are between 1 to 15 µm. The fabrication of low ohmic contact onto heavily n-doped

InGaAs layers has been extensively researched in the literature [24]-[27]. The main focus has been on removing the native-oxide that forms on the surface of the contact layer before deposition. In order to remove the native-oxide, an argon gun treatment for 1 min was applied before the metal deposition. The argon gun tool is available in Plassys IV and is carried out under vacuum. Deposition happens immediately after, as such, the sample is not exposed to the atmosphere which prevents the formation of the oxidation layer. After lift-off annealing was done at a temperature of 275° C for 1 minute.

The total resistance ( $R_t$ ) between two pads is composed of the two contact resistances ( $R_c$ ), assuming that the contact resistances are the same (i.e.  $R_{C1} = R_{C2}$ ), and the sheet resistance in-between the contacts ( $R_{sh}$ ) multiplied by the ratio between the length and width of the gap as shown in Eq. 3.10.

$$R_t = 2R_C + R_{sh}\frac{l}{w} \tag{3.10}$$

The measurement is carried out for each contact pair that is separated by various distances. A linear plot of distance vs resistance is then obtained as shown in Fig. 3.13. From Eq. 3.10 it can be seen that the slope of the line represents the semiconductor sheet resistance over the width of the contact pad (Eq. 3.11). The intersection between the line and the resistance-axis is equal to double the contact resistance (i.e.  $2 \times R_C$ ). The intersection of the line with the distance axis (when  $R_t = 0$ ) equals to double the transfer length ( $-2 \times L_T$ ). The transfer length is the average distance that an electron/hole travels in the semiconductor beneath the contact pad, before it flows up into the contact [19]. As such the effective area of contact can be regarded as  $L_T \times W$ .

$$Slope = \frac{R_{Sh}}{w}$$
(3.11)



**Figure 3.13:** Example of LTLM resistance as a function of gap distance. The intersection between the line and the resistance axis is double the contact resistance, while the intersection with the distance axis represents double the transfer length.

The specific contact resistivity of the metal contact is calculated using Eq 3.12.

$$\rho_c = L_T^2 R_{sh} \tag{3.12}$$

A fabricated LTLM test structure with a gap between the two metal contacts of 2  $\mu$ m is shown in Fig. 3.14. The measured resistance values are plotted in Fig. 3.15 and tabulated in Table 3.2. The extracted contact resistance ( $R_c$ ), the transfer length ( $L_T$ ), the sheet resistance ( $R_{sh}$ ) and the specific contact resistance are listed in Table 3.3. The achieved specific contact resistance is 10.15  $\Omega$   $\mu$ m<sup>2</sup>. For 4  $\mu$ m × 4  $\mu$ m and 5  $\mu$ m × 5  $\mu$ m devices the contact resistance is equal to 0.6  $\Omega$  and 0.4  $\Omega$ , respectively. The result is one order of magnitude higher when compared to the lowest reported contact resistance in [24].



*Figure 3.14:* SEM of a fabricated LTML test structure with a gap spacing of 2  $\mu$ m. The position of the current probes is marked with  $I_1$  and  $I_2$ , while the sense probes are marked  $V_{sense.}$ 



*Figure 3.15: Measured plot of resistance as a function of contact separation distance for TLM structure.* 

Gap spacing (µm)	Resistance (Ω)	Gap spacing (µm)	Resistance (Ω)
1	2.09	9	9.55
2	2.81	11	11.23
3	3.99	13	12.82
5	6.13	15	14.28
7	7.62		

 Table 3.2
 TLM measurement data

Table 3.3 Extracted TLM parameters

$\mathbf{R}_{\mathrm{C}}\left(\Omega\right)$	L <sub>T</sub> (μm)	${ m R}_{ m SH}~(\Omega/\Box)$	$ρ_C(Ω μm^2)$
0.67	0.60	26.52	10.15

# 3.5 Summary

In this chapter, the fabrication steps required for manufacturing of the passive components (thin-film and semiconductor resistors, MIM capacitors, coplanar waveguide) were detailed. The characteristics of the passive components were analyzed using S-parameter measurements that showed that at the frequency of interest (30 GHz) their behavior is as expected. The metal-semiconductor contact resistance was also investigated by LTLM test structures. The obtained resistance is one order of magnitude higher than the lowest reported in literature and is expected to be reduced by optimizing the surface treatment process before metallization.

## References

- J. Wang, Monolithic Microwave/Millimetrewave Integrated Circuit Resonant Tunneling Diode Sources with around a Milliwatt Output Power, PhD Thesis: University of Glasgow, 2014.
- [2] E. Bloch, D. Mistele, R. Brener, C. Cytermann, A. Gavrilov and D. Ritter, "NiCr thin film resistor integration with InP technology," *Semiconductor Science and technology*, vol. 26, no. 10, p. 105004, 2011.
- [3] H. Shen, J. Arreaga, R. Ramanathan, J. Sawyer and T. Shiban, "Fabrication and characterization of thin film resistors for GaAs-based power amplifiers," in *International Conference on Compound Semiconductor*, San Diego, USA, 2003.
- [4] R. Driad, M. Krieg, N. Geldmacher, J. Rster and F. Benkhelfifa, "Investigation of NiCr thin film resistors for InP-based monolithic microwave integrated circuits (MMICs)," *Journal of the Electrochemical Society*, vol. 158, no. 5, pp. 561-564, 2011.
- [5] A. Goswami, Thin Film Fundamentals, Delhi, India: New Age International, 2006.
- [6] C. Li, Design and characterisation of millimetre wave planar Gunn diodes and integrated circuits, PhD Thesis: University of Glasgow, 2012.
- [7] K. I. Elgaid, A Ka-band GaAs MESFET monolithic downconverter, PhD Thesis: University of Glasgow, 1998.
- [8] B. Razavi, RF Microelectronics, New Jersey, USA: Prentice Hall, 2011.
- [9] C. Wen, "Coplanar waveguide: a surface strip transmission line suitable for nonreciprocal gyromagnetic device applications," *IEEE Transactions on Microwave Theory and Techniques*, vol. 17, no. 12, pp. 1087-1090, 1969.
- [10] R. N. Simons, Coplanar waveguide circuits, componenets, and systems, New Jersey, USA: John Wiley & Sons, 2004.

- [11] M. Golio, RF and microwave handbook, Florida, USA: CRC Press, 2008.
- [12] K. Seeger, "Temperature dependence of the dielectric constants of semiinsulating III-V compounds," *Applied Physics Letters*, vol. 54, no. 13, pp. 1268-1269, 1989.
- [13] S. Gevorgian, T. Martinsson, A. Deleniv, E. Kollberg and I. Vendik, "Simple and accurate dispersion expression for the effective dielectric constant of coplanar waveguides," *IEE Proceedings - Microwaves, Antennas and Propagation*, vol. 144, no. 2, pp. 145-148, 1997.
- [14] K. Beilenhoff, H. Klingbeil, W. Heinrich and H. Hartnagel, "Open and short circuits in coplanar MMIC's," *IEEE Transactions on Microwave Theory and Techniques*, vol. 41, no. 9, pp. 1534-1537, 1993.
- [15] R. Masood and S. Mohsin, "Characterization of a coplanar waveguide open stub-based discontinuity for MMICs and filter applications," *International Journal of Antennas and Propagation*, vol. 2012, no. 1, p. 423706, 2012.
- [16] Keysight, "Understanding VNA calibrations," 2019. [Online]. Available: http://na.support.keysight.com/plts/help/WebHelp/VNACalAndMeas/Underst anding\_VNA\_Calibrations.html.
- [17] G. Ponchak and E. Tentzeris, "Finite ground coplanar waveguide (FGC) low loss, low coupling 90-degree crossover junctions," *IEEE Transactions on Advances Packaging*, vol. 25, no. 3, pp. 385-392, 2002.
- [18] G. E. Ponchak, L. P. Katehi and E. M. Tentzeris, "Finite ground coplanar (FGC) waveguide: its characteristics and advantages for use in RF and wireless communication circuits," in *3rd International Wireless Communications Conference (WCC '98)*, San Diego, USA, 1998.
- [19] D. Schroder, Semiconductor material and device characterization, New York, USA: Wiley-IEEE Press, 1998.
- [20] J. D. Plummer and P. B. Griffin, "Material and process limits in silicon VLSI technology," *Proceedings of the IEEE*, vol. 89, no. 3, pp. 240-258, 2001.

- [21] G. S. Marlow and M. B. Das, "The effects of contact size and non-zero metal resistance on the determination of specific contact resistance," *Solit-State Electronics*, vol. 25, no. 2, pp. 91-94, 1982.
- [22] F. M. Smits, "Measurement of sheet resistivities with the four-point probe," *The Bell System Technical Journal*, vol. 37, no. 3, pp. 711-718, 1958.
- [23] L. B. Valdes, "Resistivity measurements on germanium for transistors," *Proceedings of the IRE*, vol. 42, no. 2, pp. 420-427, 1954.
- [24] A. Crook, E. Lind, Z. Griffith and M. Rodwell, "Low resistance, nonalloyed ohmic contacts to InGaAs," *Applied Physics Letters*, vol. 91, no. 19, p. 192114, 2007.
- [25] A. Baraskar, M. Wistey, Y. J. Lee, B. Thibeault, A. Gossard and M. Rodwell,
   "Ex situ ohmic contacts to n-InGaAs," *Journal of Vacuum Science & Technology B*, vol. 28, no. 4, pp. 517-519, 2010.
- [26] J. C. Lin, S. Y. Yu and S. E. Mohney, "Characterization of low-resistance ohmic contacts to n- and p-type InGaAs," *Journal of Applied Physics*, vol. 114, 2013.
- [27] R. Dormaier and S. Mohney, "Factors controlling the resistance of ohmic contacts to n-InGaAs," *Journal of Vacuum Science & Technology*, vol. 30, no. 3, p. 031209, 2012.

# **Chapter 4. Resonant Tunneling Diode Bias Circuit Instabilities**

# 4.1 Introduction

Many of the proposed applications of Resonant Tunneling Diodes (RTDs) such as transceivers [1]-[3] and digital logic circuits [4]-[6] make use of the negative differential resistance region (NDR) exhibited by the device. Therefore, accurate DC and high frequency characterisation of this critical region is essential for device modelling in computer simulations [7]. However, instability in the NDR can negatively affect the diode's performance, in terms of output power, when used as an oscillator [1] and reduces the dynamic input range when operated as a receiver [2]. In this chapter, the parasitic oscillations as a result of the biasing network are studied. The basic differential equations for a typical RTD circuit are deduced and the stability criteria is derived. This chapter presents a systematic study on the influence of bias circuit oscillations on the measured DC characteristics of an RTD.

The typical measured current-voltage (I-V) characteristics of a single RTD device is shown in Fig. 4.1. Due to the presence of parasitic oscillations, the *I*-V curve shows a plateau-like feature, which has been extensively observed in literature [8]-[12], where it has been attributed to both intrinsic and extrinsic factors. In [9] the authors have explained how the coupling between the energy levels in the emitter well and in the main quantum well can result in a plateau-like structure that is seen in DC measurements, an intrinsic behaviour of their epi-layer structure design. By applying a magnetic field perpendicular to the tunnel barriers, the in-plane motion of electrons is quantized into discrete energy levels, known as Landau levels. The experimental measurements showed that the width of the plateau region decreased with respect to an increasing magnetic field, and finally disappeared at a magnetic field strength of around 8 Tesla (T) [9]. Extrinsic factors, more relevant to the current work, are attributed to the resonance between the RTD self-capacitance and the parasitic resistance and inductance present due to the biasing cables. Since reducing and/or eliminating the parasitics beyond a certain point is not feasible, various techniques need to be employed in order to maintain DC stability [12].



*Figure 4.1:* DC measurement of a  $4x4 \ \mu m^2$  RTD device. The negative differential resistance (NDR) region is located between 0.6 V and 1.2 V. The diode shows a signature plateau-region within the NDR due to parasitic oscillations.

This chapter is organized as follows. Section 4.2 presents the two RTD epi-layer structure designs used in this project as well as their DC characteristics with no bias stabilisation network. In Section 4.3, the main RTD models referenced in literature are analysed and the small-signal circuit parameter extraction discussed. Section 4.4 focuses on the stability criteria, based on the models discussed in the previous section and a new method to overcome the DC-instability is proposed. Numerical calculations, simulations and experimental data are presented in Section 4.5

# 4.2 RTD Layer Structure

Semiconductor film growth has benefited from modern epitaxial layer growth techniques such as metal organic chemical vapor deposition (MOCVD) and molecular beam epitaxy (MBE). Both techniques are able to grow high quality semiconductor materials with precise control over composition and thickness [13]. The wafers in this project were grown using MBE by IQE Ltd on a semi-insulating InP substrate. The two RTD layer structures that were used in this project are presented in Table 4.1.

Laman	Layer Structure #1 Wafer China Wafer			
Layer No.	Thickness (A)	Composition	Doping	Description
1	450	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Collector
2	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E18: Si	Sub-Collector
3	500	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer
4	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer
5	11	AlAs	Un-doped	Barrier
6	11	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Well
7	14	InAs	Un-doped	Sub-Well
8	11	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Well
9	11	AlAs	Un-doped	Barrier
10	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer
11	500	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer
12	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E18: Si	Sub-Emitter
13	200	InP	1E19: Si	Etch-Stop
14	4000	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Emitter
15	2000	InP	Un-doped	Buffer
		SI: InP		Substrate

Lover	Layer Structure #2 Wafer Baseline2				
Layer – No.	Thickness (A)	Composition	Doping	Description	
1	400	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Collector	
2	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	2E18: Si	Sub-Collector	
3	1200	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer	
4	100	In <sub>0.53</sub> Ga <sub>0.47</sub> As	2E16: Si	Spacer	
5	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer	
6	25	AlAs	Un-doped	Barrier	
7	47	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Well	
8	25	AlAs	Un-doped	Barrier	
9	20	In <sub>0.53</sub> Ga <sub>0.47</sub> As	Un-doped	Spacer	
10	100	In <sub>0.53</sub> Ga <sub>0.47</sub> As	2E16: Si	Spacer	
11	100	In <sub>0.53</sub> Ga <sub>0.47</sub> As	5E16: Si	Spacer	
12	800	In <sub>0.53</sub> Ga <sub>0.47</sub> As	2E18: Si	Sub-Emitter	
13	4000	In <sub>0.53</sub> Ga <sub>0.47</sub> As	3E19: Si	Emitter	
14	2000	InP	Un-doped	Buffer	
		SI: InP		Substrate	

Starting from the top for layer structure #1, a highly n-type doped contact layer is used as an interface between the metallization and the RTD structure, followed by a collector/emitter layer (depending on the bias polarity); graded spacer layers are added (with 5E16 doping to un-doped) and are used to increase the depletion region width, and thereby decreasing the diode's capacitance and increasing the cut-off frequency of the diode. The undoped spacer layers help reduce the diffusion of dopant impurities to the double quantum well structure. The quantum well is formed by an un-doped narrow band gap material ( $In_{0.53}Ga_{0.47}As$ ;  $E_g = 0.71 \text{ eV}$ ) sandwiched between two un-doped wide band gap barriers (AlAs;  $E_g = 2.14 \text{ eV}$ ). A 1.4 nm sub-well (layer 7) was introduced in the epilayer structure to lower the resonant energy levels inside the quantum well [14], in order to reduce the peak voltage point ( $V_p$ ). On the emitter side a 20 nm etch-stop layer was introduced in order to control the etch depth during wet/dry etching. The RTD *I-V* characteristics for the two different layer structures is shown in Fig. 4.2.



*Figure 4.2: RTD I-V characteristic of the two wafer structures used in this project: (a) layer structure #1, (b) layer structure #2* 

## **4.3 Resonant Tunneling Diode Models**

Various physical, analytical and semi-analytical models have been proposed to describe the RTD operation. An overview of the equivalent circuit models found in literature [15]-[21] indicate that most deal with the small-signal operation of the RTD. Several large-signal models of an RTD [22]-[25] derived from their small-signal parameters have been implemented into simulation programs with integrated circuit emphasis (SPICE) environments and are used in simulations. The theoretical identification of an accurate equivalent circuit for the RTD is important for:

- i. The suppression of bias circuitry oscillations [12]
- ii. Extraction of device characteristics from experimental measurements [7][26]
- iii. Maximizing the power, frequency and efficiency of RTD oscillators [1]
- iv. Understanding structural changes to the RTD which will lead to improved device characteristics [27]

### 4.3.1 RC Circuit Model

The simplest model is the small-signal equivalent RC circuit, based on the Esaki tunnel diode equivalent circuit [22], where the RTD is modelled as a resistor, with negative conductance  $-G_n$  and capacitance  $C_n$  connected in parallel as shown in Fig. 4.3. The conductance  $G_n$  of the model is the differential conductance of the device. This conductance, which has a negative sign in the NDR region is obtained by differentiating the stable DC *I-V* curve of the device, as shown in Eq 4.1.

$$G_n = \frac{\Delta I}{\Delta V} \tag{4.1}$$

The capacitance  $C_n$  represents the parallel-plate capacitance formed due to the formation of depletion layers on the collector side. This geometrical capacitance can be estimated using Eq. 4.2, where *A* is the device size,  $L_W$ ,  $L_B$  and  $L_D$  is the width of quantum well, barrier and depletion region respectively and  $\mathcal{E}_W$ ,  $\mathcal{E}_B$ , and  $\mathcal{E}_D$  are their respective dielectric constants .

$$C_n = \frac{A}{\frac{L_W}{\varepsilon_W} + \frac{2L_B}{\epsilon_B} + \frac{L_D}{\epsilon_D}}$$
(4.2)



**Figure 4.3:** RC small-signal circuit model for the RTD. The capacitance  $C_n$  represents the geometrical capacitance of the device. The conductance  $G_n$  is obtained by differentiating the stable I-V curve of the device.

## 4.3.2 RLC Circuit Model

A more accurate small-signal equivalent circuit model of the RTD is based upon the parallel-inductance theoretical model, which was proposed by Brown et. al. [17]. The physical origin of the inductance is explained as follows:

At the interface between two different semiconductor materials a heterojunction is formed, band bending will then result in the formation of a triangular well. The formation of the triangular well creates a 2d electron gas (2DEG) population [28], in which the electrons occupy quasibound states. It was observed that, when the applied bias voltage is changed, the current inside the RTD takes time to reach its new value, implying that the RTD should have an inductive character.

In his letter, Brown derived that the inductive character is related to the quasibound state lifetime ( $\Gamma_d$ ). The small-signal RLC model is shown in Fig. 4.4 with  $L_{qw}$  denoting the quantum well inductance.



*Figure 4.4: RLC small-signal equivalent circuit model for RTD. The capacitance*  $C_n$  represents the geometrical capacitance of the device. The conductance  $G_n$  is obtained by differentiating the stable I-V curve of the device.

The RTD's capacitance ( $C_n$ ) is composed of the device geometrical capacitance ( $C_0$ ), which arises from the depletion layers, and the quantum well capacitance  $C_{qw}$ , originating from the change in charge in the quantum well [21].

$$C_n = C_0 + C_{qw} \tag{4.3}$$

The geometrical capacitance  $C_0$  is the parallel plate capacitance formed by the undoped regions confined in the highly doped collector / emitter layers and is calculated using Eq. 4.2. The quantum well capacitance  $C_{qw}$  arises from the electron density change in the quantum well as a function of applied bias.

The charge variation has been derived by [21][29], where  $\Delta Q_c$  is the charge variation seen in the collector as a result of the quantum well charge ( $\Delta Q_c = -\Delta Q_w$ ) and is given by Eq. 4.4. The quantum well capacitance can also be expressed in terms of the RTDs differential conductance  $G_n$  and electron escape rate  $v_c$  (s<sup>-1</sup>).

$$C_{qw} = A \frac{\Delta Q_c}{\Delta V} = -\frac{G_n}{\nu_c} \tag{4.4}$$

The quantum inductance  $L_{qw}$  arises from the charging and discharging effect of the quantum well and is directly proportional to the electron dwell time ( $\tau_{dwell}$ ) in the quantum well during resonant tunneling [17][21]. This quantity changes exponentially with the thickness of the barrier and the electron effective mass in the barrier. The

electron dwell time is calculated using Eq. 4.5, where  $\Gamma$  is the energy full-width through the first resonant level. The frequency independent inductance can be thus estimated using Eq 4.6.

$$\tau_{dwell} = \frac{\hbar}{2\pi\Gamma} \tag{4.5}$$

$$L_{qw} = \frac{\tau_{dwell}}{G_n} \tag{4.6}$$

## **4.4 Bias Circuit Instabilities**

A typical RTD device measurement setup is presented in Fig. 4.5 (a), where  $V_{bias}$  is the bias voltage,  $R_B$  and  $L_B$  are the parasitic resistance and inductance introduced by the biasing cable,  $L_d$  is the inductance of the on-chip coplanar waveguide (CPW) pad and  $R_d$  represents the device contact resistance.



*Figure 4.5:* (a) *DC* measurement setup of an *RTD* device including the bias circuit and device parasitics, (b) simplified equivalent circuit.

The circuit present in Fig. 4.5 (a) can be simplified to the one in Fig. 4.5 (b), since all the parasitic elements are connected in series. Biasing the RTD in the NDR region using the measurement setup described in Fig. 4.5 (b) usually leads to unwanted bias oscillation. This is because the net resistance of the circuit is negative, assuming that the negative differential conductance is much greater than the parasitic elements

introduced. Increasing the series resistance  $R_B$  to be numerically larger than  $G_n$  will result in DC bistability [12][30].

### 4.4.1 Shunt Resistor Stabilisation

The main approach to stabilise an RTD found in literature is to connect a stabilising resistor  $R_{shunt}$  across the device as shown in Fig. 4.6 [31]-[33], which ensures that the combined resistance of the circuit remains positive when biased in the NDR region.



**Figure 4.6:** Circuit for stabilising the RTD using a parallel resistance ( $R_{shunt}$ ) across the device. L denotes the parasitic inductance.  $C_n$  is the diode capacitance and  $-G_n$  is the negative differential conductance. The device contact resistance has been omitted.

Looking into the circuit the real part of admittance is given by Eq. 4.7

$$Re(Y_{in}) = \frac{1}{R_{shunt}} - G_n \frac{1}{(1 - \omega^2 L C_n)^2 + (\omega L C_n)^2}$$
(4.7)

At low frequencies ( $\omega \approx 0$ ) the equation can be simplified as shown in Eq. 4.8. In order to suppress low frequency oscillations, the real part of the admittance must remain positive [31][32].

$$Re(Y_{in}) = \frac{1}{R_{shunt}} - G_n > 0 \tag{4.8}$$

$$R_{shunt} < \frac{1}{G_n} \tag{4.9}$$

The shunt resistor stabilisation method is the most widely used to date, however, there are significant drawbacks when used in circuits.

- i. the dc characteristics of the diode must be indirectly determined by firstly de-embedding the resistor from measurements [31].
- ii. the DC-to-RF conversion efficiency is greatly reduced due to a dc path to ground through the usually low value resistance [1].
- iii. the sensitivity of the diode is affected when operated as a receiver [34].

#### 4.4.2 Shunt Capacitor and Resistor Stabilisation

The RTD bias stabilisation network approach proposed in this work employs the addition of a shunt capacitor  $C_B$  connected in series with the stabilising resistance, to eliminate the dc path to ground [1]. The circuit is shown in Fig. 4.7, where  $V_{bias}$  is the bias voltage to set the device in the NDR region,  $R_S$  and  $L_S$  are the resistance and inductance introduced by the biasing cable,  $R_B$  and  $C_B$  are the shunt resistor and capacitor, respectively, and L is the series inductance of the contact pads of the device.



**Figure 4.7:** Circuit for stabilising the RTD using a parallel resistance  $(R_B)$  and capacitance  $(C_B)$  across the device.

For the analysis, the circuit presented in Fig. 4.7 is split into the low frequency and high frequency equivalents. At low frequencies (MHz range), the circuit in Fig. 4.7 can be simplified to the equivalent circuit in Fig. 4.8 (a), where the RTD is represented by its negative differential conductance ( $-G_n$ ) and self-capacitance ( $C_n$ ).

In this case, the inductance L is considered a short circuit, the shunt resistance  $R_B$  is ignored, as the impedance of capacitor  $C_B$  becomes dominant at low frequencies, and
the device capacitance  $C_n$  (typically tens of femtofarads) is considered negligible when compared with  $C_B$  (typically tens of picofarads).

On the other hand, at high frequencies (GHz range), the circuit in Fig. 4.7 can be simplified to the equivalent circuit in Fig. 4.8 (b), where the inductance  $L_B$  is considered an open circuit and the capacitance  $C_B$  is a short circuit. Note that the circuits in Fig. 4.8 are identical, only with different element values representing their low and high frequency variants.



*Figure 4.8:* (a) Simplified low frequency equivalent circuit, (b) simplified high frequency equivalent circuit.

We analyse the circuit in Fig. 4.8 (a) using nodal analysis by applying Kirchhoff's current law to give:

$$\frac{V}{R_S + sL_S} + sC_B V - G_n V = 0 (4.10)$$

where *V* is the voltage across the parallel circuit and *s* the complex frequency. From (4.10), we can obtain the following characteristic equation:

$$C_B L_s s^2 + (C_B R_s - L_s G_n)s + 1 - G_n R_s = 0$$
(4.11)

The solutions to (4.11) are given by:

$$s_{1,2} = \frac{(L_s G_n - C_B R_s) \pm \sqrt{(C_B R_s - L_s G_n)^2 - 4C_B L_s (1 - G_n R_s)}}{2C_B L_s}$$
(4.13)

The solutions of Eq. 4.13 can be classified as real or complex. Case 1: the solutions are complex and therefore:

$$(C_B R_S - G_n L_s)^2 - 4C_B L_s (1 - G_n R_s) < 0$$
 (4.14)

For the circuit to be stable, the solutions of (4.14) must fall on the left half of the complex frequency plane. As a result, the circuit is stable if:

$$C_B > \frac{L_S G_n}{R_S} \tag{4.15}$$

Case 2: the solutions are real and so:

$$(C_B R_S - G_n L_S)^2 - 4C_B L_S (1 - G_n R_S) > 0$$
(4.16)

For these solutions to fall in the left half of the complex frequency plane the magnitude of the term under the square root sign of (4.13) must be smaller than the magnitude of the first term, so:

$$R_s < \frac{1}{G_n} \tag{4.17}$$

Combining the conditions derived from case 1 and 2, the condition to achieve low frequency circuit stability is:

$$\frac{L_S G_n}{C_B} < R_S < \frac{1}{G_n} \tag{4.18}$$

Since the circuits of Fig. 4.8 (a) and Fig. 4.8 (b) are identical, equation (4.18) can be re-written with the corresponding elements to provide the condition for stability at high frequencies as:

$$\frac{LG_n}{C_n} < R_B < \frac{1}{G_n} \tag{4.19}$$

# **4.5 Measurement Results and Simulations**

To determine the efficacy of the proposed bias stabilisation approach, devices on the two different layer structures were fabricated using optical lithography techniques. The diodes where then characterised using DC and RF measurements in order to extract the parameters,  $G_n$ ,  $L_{qw}$  and  $C_n$ , as a function of the bias voltage. An accurate large-signal model of the RTD's equivalent circuit could then be established and the

stabilising networks tested. RTD's with integrated stabilising networks were then designed, fabricated and measured. The fabrication process details are described in chapters 3 and 4.

## **4.5.1 DC Measurements**

A diagram of the DC measurement set up is shown in Fig. 4.9 and it consists of the Keysight technology B1500 to supply the bias voltage, the bias tee, the DUT and the spectrum analyser/oscilloscope to measure any parasitic oscillations.



*Figure 4.9: DC* measurement set-up for an RTD device. A spectrum analyser/oscilloscope is used to measure the parasitic oscillations.

The measured *I-V* characteristic of a fabricated device is shown in Fig. 4.10. The *I-V* presents the typical plateau-like feature in the NDR region due to the presence of parasitic oscillations during measurement. Without a working stabilising network to extract the true characteristics of the device, the data points extracted from the DC measurements in the NDR are inaccurate and must be discarded. In order to calculate the differential conductance, a large-signal analytical model [25] is used to reconstruct the I-V characteristic, with the NDR approximated with a smooth trace.

From the described model in [25], the RTD I-V is given by:

$$I(V) = S \times \left( J_{exp}(V) + J_{Gaussian}(V) + J_{mod}(V) \right) \quad (4.20)$$

where

$$J_{exp}(V) = A(\exp(B \times V) - \exp(-B \times V))$$
(4.21)

$$J_{Gaussian}(V) = C1(\exp(-D1 \times (V - E1)^2 - \exp(-D1 \times (V + E1)^2))$$
(4.22)

$$J_{mod}(V) = C2(tan^{-1}(D2 \times (V - E2)) + tan^{-1}(D2 \times (V + E2)))$$
(4.23)

The parameters of the equation A, B, C, D, and E are used to achieve the different RTD I-V characteristics and S represents the scaling factor. The parameters are extracted using a curve fitting tool in MATLAB. The resulting spline is continuous and using (4.1) the negative differential conductance is computed and plotted on the right-axis of Fig. 4.10.



*Figure 4.10: Measured I-V characteristic of an RTD device (black), large-signal model of the I-V characteristic with the NDR approximated with a smooth trace (blue), calculated value of conductance of model I-V (red).* 

The extracted model parameters are:

A = 0.9623 A/m<sup>2</sup>, B = 0.0945 V<sup>-1</sup>, C1= 0.1362 A/m<sup>2</sup>, C2 = -0.2847 A/m<sup>2</sup>, D1 = 6.351 V<sup>-2</sup>, D2 = 0.3915 V<sup>-1</sup>, E1 = 0.6744 V, E2 = -0.2612 V

The large-signal model provides a good fit to the measured data, able to reproduce the RTD's PDR regions. The NDR is an approximate and should be sufficiently accurate to calculate  $G_n$ . Using the same setup as described in Fig. 4.9, the parasitic oscillations were measured by a spectrum analyser and are presented in Fig. 4.11. The frequency of oscillation is around 3.25 MHz.



*Figure 4.11:* Low frequency oscillations of the RTD measured by a spectrum analyser.

## 4.5.2 RF Measurements

For the purpose of accurate parameter extraction, S-parameter measurements were carried out at different bias points in the positive differential regions using a Keysight E8361A vector analyser (VNA) with a port power of -20 dBm together with a Keysight 1500A device parameter analyser. The signal from the VNA and the bias voltage were both fed using a 100  $\mu$ m co-planar ground-signal-ground (GSG) probe using a bias tee. The measurement setup is similar to the one presented in Fig. 4.8, with the VNA replacing the oscilloscope.

An initial short-open-load (SOLT) calibration was performed to eliminate systematic errors and move the measurement reference point to the tips of the probe. The Cascade Microtech standard substrate was used for the short, thru and 50  $\Omega$  precision load standard. The open standard was done by lifting the probe tips in the air [35].

The effects of the parasitic elements originating from the pad structures of the RTD have been de-embedded after measurements by including open and short dummy pad test structures on the wafer substrate. The full details on the de-embedding procedure is detailed in [7] and a micrograph of the dummy pad structure is shown in Fig. 4.12.



*Figure 4.12: Micrograph of open and short dummy pad structures used for deembedding of the parasitic elements originating from the CPW pads.* 

Fig. 4.13 shows the measured and fitted values for the real and imaginary part of S11. The frequency span for the measured data was 10 MHz to 110 GHz, however around 20 GHz the system had a resonance and no data could be extracted. A schematic model of the small signal parameters  $R_{s}$ ,  $L_{qw}$ ,  $G_n$  and  $C_n$  was created in Keysight Advanced Design System (ADS) software, where, the circuit elements are extracted by adjusting the simulated model circuit to the measured results. The contact resistance  $R_s$  was extracted using transmission line method (TLM) measurements and the process was described in Chapter 3.



*Figure 4.13:* Measured and simulated real and imaginary parts of S11 after deembedding the parasitics at 0.1 V bias voltage. No data could be extracted above 15 GHz as the RTD and the pads were in resonance.

The process was carried out in the microwave frequencies range at several bias points in the first and second PDR regions. No data could be extracted in the NDR due to not having a device bias stabilization method. Using the extracted values for the inductance at several bias points the experimental value of the time constant  $\tau$  is obtained from the extracted parameters and is estimated at 2.65 ps. This estimation is reasonable, with the experimental value being matched when considering 1.2 nm barrier thickness in the calculation. Using (4.6) the model for the device inductance is presented in Fig. 4.14.



*Figure 4.14: Extracted values for the device inductance (red), model of the device inductance obtained (blue)* 

The device capacitance  $C_n$  is composed of the geometrical capacitance and the quantum well capacitance. The geometrical capacitance estimated using Eq. 4.2 is about 21 fF with a device area of  $5x5 \ \mu\text{m}^2$ . The quantum well capacitance can then be calculated by subtracting the geometrical capacitance from the measured data. Using Eq 4.3 the electron escape rate from the quantum well to the collector is calculated at 0.1 ps and the capacitance variation with bias is plotted in Fig. 4.15.



*Figure 4.15: Extracted values for the device capacitance (red), model of the device capacitance obtained (blue)* 

#### **4.5.3 Simulation Results**

Based on the extracted parameter values, a MATLAB large-signal model of the RTD was realized and can be seen in Fig. 4.16. The lumped elements  $L_{qw}$ ,  $C_n$  and  $G_n$  represent the quantum inductance (RLC) model.  $R_{contact}$  represents the device contact resistance and was set to be 5  $\Omega$ ,  $G_n$  was modelled as a variable resistance,  $L_{qw}$  and  $C_n$  as custom blocks with the values based on the model inductance and capacitance described previously. The lumped elements  $R_s$  (0.1 $\Omega$ ) and  $L_s$  (56 nH) represent the bias cable resistance and inductance respectively.



*Figure 4.16:* MATLAB Simulink large-signal model of RTD measurement setup. The negative differential resistance has been modelled as a variable resistance with the values extracted from the DC large-signal model of the RTD (Fig. 4.10). The inductance and capacitance have been modelled using custom block with their values extracted from Fig. 4.14 and Fig. 4.15, respectively.

The input voltage was increased in the time-domain simulation at fixed intervals and current measurements were recorded in-between voltage changes. The RMS current value at a specific voltage level could then be calculated. When biased in the NDR the circuit oscillated, and a plateau-like region was observed when calculating the RMS current value at each biasing condition. The model and the measurement data are presented below with good agreement, however the model's NDR region extends 100mV more than the measured RTD device and is attributed to inaccuracies

introduced by the large signal model. As a result, the parasitic oscillations are sustained until 1.3V.



*Figure 4.17: Measured I-V characteristic of an RTD device (black), Simulated MATLAB model of DC measurement setup with added extrinsic parasitic elements (red).* 

The MATLAB model was then modified to include the proposed stabilisation network. Based on the model the values for the stabilising network required are  $R_B = 10 \Omega$  and  $C_B = 144$  pF. The requirements on  $L_S$  and  $C_B$  could be relaxed by using an additional small series resistance  $R_s$ , which can then be de-embedded from the measurements. The simulations show that with the addition of the stabilisation network no parasitic oscillations are present in the device, and the *I-V* characteristic matches the large-signal model *I-V*.

#### **4.5.4 Experimental Results**

To demonstrate the feasibility of the stabilisation network, RTDs with monolithically integrated bias resistors and capacitors were fabricated on the two different layer structures. A micrograph of a fabricated 4x4  $\mu$ m<sup>2</sup> RTD with integrated stabilisation network is shown in Fig. 4.18, where the capacitor *C*<sub>B</sub> (144 pF) and resistor *R*<sub>B</sub> (10  $\Omega$ ) are each realized from two parts placed in parallel with the device. The value for the stabilising network is chosen to satisfy both Eq. (4.18) and (4.19).



**Figure 4.18:** Micrograph of a fabricated 4x4 RTD device with integrated bias stabilisation. The capacitor  $C_B$  and resistor  $R_B$  are comprised of two parts connected in parallel with the RTD device.

Using the same measurement setup as described in Fig. 4.9 the frequency spectrum of the RTDs in the NDR region was analysed and shown in Fig. 4.19. For the R and RC stabilisation network the 3 MHz bias oscillation has been removed, and no other high frequency oscillations were observed beyond 10MHz.



Figure 4.19: Frequency spectrum of RTDs with different stabilisation networks.

The RTD *I-V* characteristics for the two different RTD structures is presented in Fig. 4.20. For the RTD in Fig. 4.20 (a) the plateau region can be seen in both the device with no stabilisation and the R stabilisation method. The RC stabilisation I-V still presents a small plateau-region; however, it is believed that it's the result of the coupling between energy levels in the emitter well and in the main quantum well [9]. For the RTD shown in Fig. 4.20 (b) both stabilisation networks have resulted in similar

*I-V* characteristics with the small distortions of the R stabilisation network attributed to small errors in the de-embedding process of the resistor.



**Figure 4.20:** RTD I-V characteristics: (a)  $4x4 \ \mu m^2$  RTD I-V with and without bias stabilisation based on wafer #1,(b)  $4x4 \ \mu m^2$  RTD I-V with and without bias stabilisation based on wafer #2

The experimental results show that both the bias circuits eliminate parasitic oscillations associated with the biasing network. Compared with the shunt resistor stabilisation network, the RC method would eliminate the need of de-embedding the shunt resistance and improve the DC-to-RF conversion efficiency of the RTD oscillators by removing the dc path to ground through the usually low value resistance, typically ~10  $\Omega$ .

# 4.6 Summary

In this chapter the origin of the resonant tunneling diodes plateau region has been investigated. Design equations for an RTD biasing circuit that prevents parasitic oscillations in the NDR region has been derived, based on the discussed models. The design approach used in this work eliminates both low-frequency bias oscillations and high-frequency oscillations, which would hinder oscillator performance and increase the DC-to-RF conversion efficiency. The measured frequency spectrum of the stabilised RTDs confirms the validity of the design criteria.

# References

- [1] A. C. Cornescu, R. Morariu, A. Ofiare, A. Al-Khalidi, J. Wang, J. M. L. Figueiredo and E. Wasige, "High efficiency bias stabilisation for resonant tunneling diode oscillators," *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 8, pp. 3449-3454, 2019.
- [2] T. Miyamoto, A. Yamaguchi and T. Mukai, "Terahertz imaging system with resonant tunneling diodes," *Japanese Journal of Applied Physics*, vol. 55, no. 3, p. 032201, 2016.
- [3] J. Wang, A. Al-khalidi, K. Alharbi, A. Ofiare, E. Wasige and J. Figueiredo,
   "High performance resonant tunneling diode oscillators as terahertz sources," in
   46th European Microwave Conference (EuMC), London, UK, 2016.
- [4] P. Mazumder, S. Kulkarni, M. Bhattacharya, J. P. Sun and G. I. Haddad, "Digital circuit applications of resonant tunneling devices," *Proceedings of the IEEE*, vol. 86, no. 4, pp. 664-686, 1998.
- [5] S. Saha, K. Biswas and M. R. Hossain, "Analysis of digital inverter using single and multiple GaAs/AlGaAs based double barrier resonant tunneling diode," in *International Conference on Innovation in Engineering and Technology* (*ICIET*), Dhaka, Bangladesh, 2018.
- [6] S. Choi, Y. Jeong, J. Lee and K. Yang, "A novel high-speed multiplexing IC based on resonant tunneling diodes," *IEEE Transactions on Nanotechnology*, vol. 8, no. 4, pp. 482-486, 2009.
- [7] R. Morariu, J. Wang, A. C. Cornescu, A. Al-Khalidi, A. Ofiare and E. Wasige, "Accurate small-signal equivalent circuit modelling of resonant tunneling diodes to 110 GHz," *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 11, pp. 4332-4340, 2019.
- [8] P. Zhao, "Numerical simulation of resonant tunneling structures: origin of hysteresis and plateau-like structure and dependence of the intrinsic bistability

and structural parameters," *Journal of Applied Physics*, vol. 87, no. 3, pp. 1337-1349, 2000.

- [9] Z. J. Qiu, Y. S. Gui, S. L. Guo and N. Dai, "Experimental verification on the origin of plateau-like current-voltage characteristics of resonant tunneling diodes," *Applied Physics Letters*, vol. 84, no. 11, pp. 1961-1963, 2004.
- [10] V. J. Goldman, D. C. Tsui and J. E. Cunningham, "Observation of intrinsic bistability in resonant tunneling structures," *Physical Review Letter*, vol. 58, no. 12, pp. 1256-1259, 1987.
- [11] B. A. Biegel, "Wigner function simulation of intrinsic oscillations, hysteresis, and bistability in resonant tunneling structures," in *Optoelectronics and High-Power Lasers and Applications*, San Jose, USA, 1998.
- [12] C. Kinder, I. Mehdi, J. R. East and G. I. Haddad, "Power and stability limitations of resonant tunneling diodes," *IEEE Transactions on Microwave Theory and Techniques*, vol. 38, no. 7, pp. 864-872, 1990.
- [13] "Room-temperature resonant-tunneling-diode terahertz oscillator based on precisely controlled semiconductor epitaxial growth technology," NTT Technical Review, vol. 9, no. 10, pp. 1-7, 2011.
- [14] F. Chevoir and B. Vinter, "Scattering-assisted tunneling in double-barrier diodes: scattering rates and valley current," *Physical Review B*, vol. 47, no. 12, pp. 7260-7274, 1993.
- [15] P. Zhao, H. Cui, D. Woolard, K. Jensen and F. Buot, "Equivalent circuit parameters of resonant tunneling diodes extracted from self-consistent Wigner-Poisson simulation," *IEEE Transaction on Electron Devices*, vol. 48, no. 4, pp. 614-627, 2001.
- [16] M. N. Feiginov, "Displacement currents and the real part of high-frequency conductance of the resonant-tunneling diode," *Applied Physics Letters*, vol. 78, no. 21, pp. 3301-3303, 2001.

- [17] E. R. Brown, C. D. Parker and T. C. Sollner, "Effect of quasibound-state lifetime on oscillation power of resonant tunneling diodes," *Applied Physics Letters*, vol. 54, no. 10, pp. 934-936, 1989.
- [18] J. P. Mattia, A. L. McWhorter, R. J. Aggarwal, F. Rana, E. R. Brown and P. Maki, "Comparison of a rate-equation model with experiment for the resonant tunneling diode in the scattering-dominated regime," *Jurnal of Applied Physics*, vol. 84, no. 2, pp. 1140-1148, 1998.
- [19] F. W. Sheard and G. A. Toombs, "Space-charge effects and AC response of resonant tunneling double-barrier diodes," *Solid-State Electronics*, vol. 32, no. 12, pp. 1443-1447, 1989.
- [20] J. Genoe, C. Van Hoof, W. Van Roy, J. Smet, K. Fobelets, R. Mertens and G. Borghs, "Capacitances in double-barrier tunneling structures," *IEEE Transactions on Electron Devices*, vol. 38, no. 9, pp. 2006-2012, 1991.
- [21] Q. Liu, A. Seabaugh, P. Chahal and F. J. Morris, "Unified AC model for the resonant tunneling diode," *IEEE Transactions on Electron Devices*, vol. 51, no. 5, pp. 653-657, 2004.
- [22] M. Lotfi and D. Zohir, "A SPICE behavioral model of tunnel diode: simulation and application," *International Journal of Control and Automation*, vol. 9, no. 4, pp. 39-50, 2016.
- [23] A. Sellai, H. Al-Hadhrami, S. H. Al-Harthi and M. Henini, "PSPICE resonant tunneling diode models and application circuits," *International Journal of Electronics*, vol. 92, no. 3, pp. 131-142, 2005.
- [24] E. R. Brown, O. B. McMahon, L. J. Mahoney and K. M. Molvar, "SPICE model of the resonant-tunnelling diode," *Electronics Letters*, vol. 32, no. 10, pp. 938-940, 1996.
- [25] S. F. Nafea and A. A. Dessouki, "An Accurate Large-signal SPICE Model For Resonant Tunneling Diode," in *International Conference on Microelectronics*, Cairo, 2010.

- [26] S. Diebold, S. Nakai, K. Nishio, J. Kim, K. Tsuruda, T. Mukai, M. Fujita and T. Nagatsuma, "Modeling and simulation of terahertz resonant tunneling diodebased circuits," *IEEE Transactions on Terahertz Science and Technology*, vol. 6, no. 5, pp. 716-723, 2016.
- [27] V. P. Kesan, D. P. Neikirk, P. A. Blakey, B. G. Streetman and T. D. Linton, "The influence of transit-time effects on the optimum design and maximum oscillation frequency of quantum well oscillators," *IEEE Transactions on Electron Devices*, vol. 35, no. 4, pp. 405-413, 1988.
- [28] M. A. B. M. Zawawi, Advanced InGaAs/AlAs resonant tunneling diodes for applications in integrated mm-waves MMIC oscillators, PhD Thesis: University of Manchester, UK, 2015.
- [29] R. Lake and J. Yang, "A physics based model for the RTD quantum capacitance," *IEEE Transactions on Electron Devices*, vol. 50, no. 3, pp. 785-789, 2003.
- [30] M. Reddy, Schottky-collector resonant tunnel diodes for sub-millimeterwave applications, PhD Thesis: University of California, Santa Barbara, 1997.
- [31] M. Bao and K. L. Wang, "Accurately measuring current-voltage characteristics of tunnel diodes," *IEEE Transactions on Electron Devices*, vol. 53, no. 10, pp. 2564-2568, 2006.
- [32] M. Reddy, R. Y. Yu, H. Kroemer, M. J. W. Rodwell, S. C. Martin, R. E. Muller and R. P. Smith, "Bias stabilization for resonant tunnel diode oscillator," *IEEE Microwave and Guided Wave Letters*, vol. 5, no. 7, pp. 219-221, 1995.
- [33] L. A. Davidson, "Optimum stability criterion for tunnel diodes shunted by resistance and capacitance," *Proceedings of the IEEE*, vol. 51, no. 9, p. 1233, 1963.
- [34] S. Diebold, S. Nishio, Y. Nishida, J. -Y. Kim, K. Tsuruda, T. Mukai, M. Fujita and T. Nagatsuma, "High-speed error-free wireless data transmission using a terahertz resonant tunneling diode transmitter and receiver," *Electronics Letters*, vol. 52, no. 24, pp. 1999-2001, 2016.

[35] Cascade Microtech, "Choosing the optimal two-port calibration," 2011.[Online]. Available: http://www.cascademicrotech.com/blog/2011/choosing-the-optimal-two-port-calibration-method.

# Chapter 5. Resonant Tunneling Diode Based Oscillators

# **5.1 Introduction**

As described previously in Chapter 1, devices which exhibit negative differential resistance (NDR) can be used to realize RF sources. This chapter describes and presents RTD based oscillator topologies and adapts the proposed stabilisation method discussed in Chapter 4 to increase the DC-to-RF conversion efficiency of an oscillator circuit. The derivation of the theoretical maximum output power and cut-off frequency for an RTD oscillator is also presented.

The RTD oscillators in this project were designed for on-wafer characterization. The spectrum analyzer used in this project was the E4448A spectrum analyzer from Agilent Technologies, which can measure frequencies up to 50 GHz. For higher frequency measurements an external mixer is used to down-convert the frequency. The measurement setup for RF oscillators is shown in Fig. 5.1.



**Figure 5.1:** Schematic block diagram used for on-wafer frequency measurements using a spectrum analyser. Biasing of the device is done through a ground-signal-ground (GSG) probe, an external DC block capacitor is used when not incorporated on-chip.

This chapter is organized as follows: Section 5.2 describes the various performance metrics of RTD-based oscillator circuits, including the expected radio frequency (RF) power generated. Section 5.3 covers the conventional RTD oscillators as well as the high-efficiency design and compares them based on their respective figure-of-merits (FOM). A comparison between RTD-based voltage-controlled oscillators (VCOs) and other published oscillators in the same frequency range is also presented.

# **5.2 Characteristics of RTD oscillator**

#### 5.2.1 Maximum RF Power



**Figure 5.2:** RF large-signal equivalent circuit of an RTD based oscillator with the RTD represented by a voltage controlled current source I = f(V) and selfcapacitance  $C_n$ . L represents the resonating inductor and  $R_{load}$  is the load resistance.

The RF equivalent circuit of an RTD based oscillator is shown in Fig. 5.2, where the RTD is replaced by its large signal model, consisting of a voltage controlled current source and self-capacitance ( $C_n$ ) [1]. For analytical purposes the voltage controlled current source I = f(V) when biased in the NDR can be represented by a cubic polynomial [2]-[4] as shown in Eq. 5.1, where *a* and *b* are constant positive coefficients associated with the particular device,  $a = (3\Delta I/2\Delta V)$  and  $b = (2\Delta I/\Delta V^3)$ .

$$I(V) = -aV + bV^3 (5.1)$$

By applying Kirchhoff's current, the currents in the circuit can be expressed as  $i_1 + i_2 + i_3 + i = 0$ , or

$$\frac{1}{L}\int v \, dt + C_n \frac{dv}{dt} + \frac{v}{R_{load}} + (-av + bv^3) = 0 \tag{5.2}$$

by differentiating with respect to time Eq. 5.2 can be re-written as

$$\frac{v}{L} + C_n \frac{d^2 v}{dt^2} + \left(\frac{1}{R_{load}} - a + 3bv^2\right) \frac{dv}{dt} = 0$$
(5.3)

And can be re-arranged as shown in Eq. 5.4, where  $G_{load} = 1/R_{load}$ 

$$LC_n \frac{d^2 v}{dt^2} + L(G_{load} - a + 3bv^2) \frac{dv}{dt} + v = 0$$
(5.4)

Assuming that the voltage v is a sinusoidal signal,  $v = V\cos(\omega t)$  where V is the amplitude of the signal and  $\omega$  is the angular frequency given by  $\frac{1}{\sqrt{LC_n}}$ . The instantaneous power dissipated by the load can be described as

$$P_{load} = v^2 G_{load} = (V cos(\omega t))^2 G_{load}$$
(5.5)

by integrating Eq. 5.5 over one period the average power dissipated by the load can be written as

$$P_{avg\_load} = \int_0^{2\pi} V^2 \cos(\omega t)^2 G_{load} dt = G_{load} \frac{V^2}{2}$$
(5.6)

Similarly, the instantaneous power over the RTD device is given by

$$P_{RTD} = -iv = av^2 - bv^4 \tag{5.7}$$

which, when integrating over one period will result the average power generated by the RTD device

$$P_{RTD_{avg}} = \int_0^{2\pi} a(\operatorname{Vcos}(\omega t))^2 - b(\operatorname{Vcos}(\omega t))^4 dt =$$

$$=\frac{uv^{2}}{2} - \frac{3bv^{3}}{8}$$
(5.8)

As the power generated by the RTD device is equal to the power dissipated over the load (5.6) and (5.8) can be combined, as shown in (5.9)

$$\frac{aV^2}{2} - \frac{3bV^4}{8} = G_{load} \frac{V^2}{2}$$
(5.9)

The solution for V in Eq. 5.9 is given by

$$V = 2\sqrt{\frac{G_n - G_{load}}{3b}} \tag{5.10}$$

By substituting (5.10) into (5.6), the average power dissipated over the load is then given by

$$P_{avg\_load} = \frac{2(G_n - G_{load})G_{load}}{3b}$$
(5.11)

And when  $G_{load} = Gn / 2$ , the maximum theoretical output power delivered to the load (obtained by differentiating Eq. 5. 11 and equating to zero) is given by

$$P_{max} = \frac{G_n^2}{6b} = \frac{3}{16} \Delta I \Delta V \tag{5.12}$$

The maximum output power is proportional to the difference in peak-to-valley current  $\Delta I$  and peak-to-valley voltage  $\Delta V$ . A device with a larger area and hence a larger  $\Delta I$  is needed in order to obtain a higher output power. For 4 µm × 4 µm and 5 µm × 5 µm devices fabricated on layer structure #1 (detailed in Chapter 4) the maximum RF power is 3.3 mW and 5.6 mW, respectively.

## 5.2.2 Effect of Parasitics on Maximum RF Power

In the derivation of output power presented in the previous section, the circuit model does not account for the resistive losses due to parasitics. The series contact resistance  $R_s$  and parasitic inductance  $L_s$  have a large impact on power delivery as the oscillation frequency of the circuit increases. The RF equivalent circuit in which the parasitic elements are also included is shown in Fig. 5.3 (a) [5][6]. This circuit can be re-drawn as shown in Fig. 5.3 (b) where the total equivalent impedance of the load  $Z_{total}$  is given by Eq. 5.13.



*Figure 5.3:* (a) *RTD* large-signal model oscillator with parasitic resistance  $R_s$  and parasitic inductance  $L_s$ . (b) *RF* equivalent circuit of *RTD* large-signal model oscillator.

$$Z_{total} = R_s + R'_L + jX \tag{5.13}$$

$$R'_{L} = \frac{\omega^2 L^2 G_L}{1 + \omega^2 L^2 G_L^2} \tag{5.14}$$

$$X = \frac{\omega L}{1 + \omega^2 L^2 G_L^2} + \omega L_S \tag{5.15}$$

The circuit shown in Fig. 5.3 (b) can be re-drawn in terms of admittance  $Y_{total}$  of the series resistance  $R_s$ , inductance L and load  $G_L$  as shown in Fig. 5.4.

$$Y_{total} = \frac{1}{Z_{total}} \tag{5.16}$$



*Figure 5.4: RTD* large-signal model oscillator with equivalent parallel load conductance  $G_{L}^{'}$  and equivalent parallel susceptance *B*.

By combining (5.13) – (5.16), the equivalent conductance  $G'_L$  and equivalent susceptance *B* can be described as

$$G'_{L} = \frac{R_{s} + \frac{\omega^{2}L^{2}G_{L}}{1 + \omega^{2}L^{2}G_{L}^{2}}}{(R_{s} + \frac{\omega^{2}L^{2}G_{L}}{1 + \omega^{2}L^{2}G_{L}^{2}})^{2} + (\frac{\omega L}{1 + \omega^{2}L^{2}G_{L}^{2}} + \omega L_{s})^{2}}$$
(5.17)

$$B = \frac{\frac{\omega L}{1 + \omega^2 L^2 G_L^2} + \omega L_S}{(R_S + \frac{\omega^2 L^2 G_L}{1 + \omega^2 L^2 G_L^2})^2 + (\frac{\omega L}{1 + \omega^2 L^2 G_L^2} + \omega L_S)^2}$$
(5.18)

Since the circuit presented in Fig. 5.4 is identical to that described in Fig. 5.2, the power delivered to the equivalent load is given by

$$P_{load} = \frac{2(G_n - G'_L)G'_L}{3b}$$
(5.19)

Referring back to Fig. 5.3 the resistance  $R'_L$  is the apparent load resistance seen by the diode due to the external circuit components. Therefore, the power delivered to the load can be expressed as

$$P_{load} = \frac{2(G_n - G'_L)G'_L}{3b} \frac{R'_L}{R'_L + R_s}$$
(5.20)

The apparent load  $G_L$  varies with frequency and therefore does not present an ideal load for maximum power, i.e. the output power drops when the oscillation frequency increases. For any given frequency, an optimum value for the RTD oscillator load  $G_L$  may be found.

For a 5  $\mu$ m × 5  $\mu$ m RTD oscillator on layer structure #1 with G<sub>n</sub> = 150 mS, R<sub>s</sub> = 0.6  $\Omega$ , L<sub>s</sub> = 10 fH, L = 150 pH and b = 0.28, the simulated output power vs frequency is presented in Fig. 5.5. The simulations show that for a given RTD device an optimum frequency and load conductance exists. The analysis presented here is an upper limit to the maximum output power as it does not include the effects of the quantum well inductance  $L_{qw}$ , or the variation in conductance  $G_n$  and capacitance  $C_n$  with bias and frequency.



**Figure 5.5:** Simulated output power as a function of frequency for a single 5  $\mu$ m × 5  $\mu$ m RTD oscillator with Gn = 150 mS, Rs = 0.6  $\Omega$ , Ls = 10 fH, L = 150 pH and b = 0.28.

# 5.2.3 Oscillation and Cut-off Frequency



*Figure 5.6: RTD* small-signal equivalent circuit with parasitic series resistance  $R_s$  and parasitic inductance  $L_s$ .

The equivalent circuit model of an RTD including the parasitic series resistance  $R_s$  and inductance  $L_s$  is shown in Fig. 5.6, where the impedance of the system can be described by

$$Z = R_{s} + \frac{-G_{n}}{G_{n}^{2} + \omega C_{n}^{2}} + j\omega (L_{s} - \frac{C_{n}}{G_{n}^{2} + \omega C_{n}^{2}})$$
(5.21)

At the angular frequency  $\omega_r$  given by (5.22) the real part of the impedance becomes zero. This frequency corresponds to  $f_{max}$ , the cut-off frequency for the diode. When the frequency is larger than  $f_{max}$  the real part of Z becomes positive, and so the RTD will no longer functions as an active device.

$$\omega_r = \frac{G_n}{C_n} \sqrt{\frac{1}{G_n R_s} - 1} \tag{5.22}$$

For an RTD oscillator the oscillation frequency can be derived from the susceptance of the circuit. Fig. 5.7 shows the equivalent circuit model for an RTD oscillator including parasitic elements.



Figure 5.7: RTD RF small-signal equivalent circuit with parasitic resistance Rs.

The admittance looking into the RTD circuit is given by

$$Y_{in} = \frac{\omega^2 C_n^2 R_s + G_n^2 R_s + G_n}{G_n^2 R_s^2 + 2G_n R_s + \omega^2 C_n^2 R_s^2 + 1} + j \left(\frac{\omega C_n}{G_n^2 R_s^2 + 2G_n R_s + \omega^2 C_n^2 R_s^2 + 1} - \frac{1}{\omega L}\right)$$
(5.23)

The susceptance B is given by

$$B = \frac{\omega C_n}{G_n^2 R_s^2 + 2G_n R_s + \omega^2 C_n^2 R_s^2 + 1} - \frac{1}{\omega L}$$
(5.24)

And by equating (5.24) to 0 the frequency of oscillation is given by

$$f_{osc} = \frac{G_n R_s + 1}{2\pi \sqrt{LC_n - C_n^2 R_s^2}}$$
(5.25)

Assuming that the parasitic series resistance is negligible, i.e.  $R_s \approx 0$  the above equation can be simplified to

$$f_{osc} = \frac{1}{2\pi\sqrt{LC_n}} \tag{5.26}$$

## 5.2.4 Phase Noise

Phase noise (PN) is a key parameter for many oscillator applications and is a measure of the power spectral density of the phase angle. It is the result of short-term variations or fluctuations that occur in a signal due to time-domain instabilities. Phase noise in the frequency domain is equivalent to jitter in the time domain. An ideal oscillation can be expressed in the frequency domain as [7][8]

$$V(t) = A\sin(\omega t) \tag{5.27}$$

where A is the signal amplitude and  $\omega$  represents the angular frequency given by  $\omega = 2\pi f$ . However, circuit and device noise can alter both the phase and amplitude of the oscillator. The disturbance from an ideal oscillation can be expressed as

$$V(t) = [A + \varepsilon(t)] \sin(\omega t + \varphi(t))$$
(5.28)

where  $\varepsilon(t)$  represents fluctuations present in the amplitude of the signal and  $\varphi(t)$  is the phase fluctuations. Inherently all practical oscillators have an amplitude-limiting mechanism, because the oscillations occur when the loop gain is unity. If the amplitude grows, gain compression caused by the non-linearity of the transfer function occurs and the oscillation amplitude dampens. Likewise, if the amplitude drops, the loop gain goes over unity and its restored due to expansive characteristics. As such the contribution of the amplitude variation on the signal is greatly attenuated and phase noise generally dominates [9].

The spectrum of a pure sinusoid such as the one described in (5.27) is a thin line and can be mathematically described as a Dirac delta function [10]. Noise present in the signal broadens the spectrum of the signal as shown in Fig. 5.8, where the noise spectrum of an RTD oscillator is shown spreading out on either side of the signal.



**Figure 5.8:** Noise spectrum of an RTD oscillator. The ideal oscillator is represented by a Dirac delta function at  $f_0$ . The phase noise is measured as the ratio of offset frequency output power (highlighted in blue) relative to the carrier output power.

The phase noise is characterized by the ratio of the noise power in 1 Hz bandwidth at a frequency offset ( $f_0 + f_c$ ) away from the carrier ( $f_0$ ) to the carrier signal power (5.29) and it is measured in decibels relative to the carrier per Hertz of bandwidth, dBc/Hz.

$$L(f) = \frac{P_{offset/Hz}}{P_{carrier}}$$
(5.29)

where L(f) is the single sideband power referenced to the carrier,  $P_{offset/Hz}$  is the power in a 1Hz bandwidth at a frequency away from the carrier and  $P_{carrier}$  is the power of the carrier signal.

## 5.2.5 Figure of Merit for VCOs

Efficiency in an oscillator is defined as the ration between the useful power output (RF output power) divided by the total power dissipation in the system, i.e. (5.30) where  $P_{RF}$  is the power delivered to the RF load and  $P_{DC}$  is the dc power dissipated by the oscillator. The DC-to-RF conversion efficiency of conventional RTD oscillators is usually less than 1%, a problem caused mainly by the usually low value shunt resistor used for stabilisation (~ 10  $\Omega$ ) in oscillator topologies. For a conventional RTD oscillator design (shown in Fig. 5.9) when the voltage bias  $V_{bias} = 1$  V the power dissipated by a 10  $\Omega$  shunt resistance is 100mW, while the RTD device only consumes around 10 - 20 mW.

$$Efficiency = \frac{P_{RF}}{P_{DC}} * 100$$
(5.30)

As the characteristics of voltage-controlled oscillators (VCO) include oscillation frequency, output power, efficiency and tuning range a widely adopted measure to identify the characteristics of the VCO is the figure-of-merit (FOM) [11].

$$FOM = L(\Delta f_{offset}) - 20\log\left(\frac{f_0}{\Delta f_{offset}}\right) + 10\log\left(\frac{P_{DC}}{1mW}\right) \quad (5.31)$$

$$FOM_T = FOM - 20\log\left(\frac{TR}{10}\right)$$
(5.32)

$$FOM_{TP} = FOM_T - 10\log\left(\frac{P_{RF}}{1mW}\right)$$
(5.33)

where  $L(\Delta f_{offset})$  represents the phase noise at the offset frequency,  $f_0$  is the oscillation frequency,  $P_{DC}$  is the dc power dissipated by the oscillator, TR represents the tuning range (%) and  $P_{RF}$  is the measured output power of the oscillator. The performance of the VCO is regarded to be better with a higher absolute value of FOM.

# **5.3 RTD Oscillator Topologies**

#### **5.3.1 Single RTD Oscillators**

The schematic circuit diagram of a single RTD oscillator is shown in Fig. 5.9.  $V_{bias}$  represents the bias voltage applied to the diode with  $R_S$  and  $L_S$  representing the parasitic resistance and inductance introduced by the biasing network.  $R_B$  is the stabilising resistor connected in parallel with the RTD in order to suppress the low frequency bias oscillations. The value for  $R_B$  is chosen as to satisfy Eq. 5.34.

$$R_B < \frac{1}{G_n} \tag{5.34}$$

The decoupling capacitor  $C_E$  is used to short-circuit the RF signal to ground in order to avoid RF power been dissipated across the stabilising resistor  $R_B$  [12]. The capacitor has a width and height of 100 µm and 350 µm respectively, the thickness of the Si<sub>3</sub>N<sub>4</sub> dielectric is 75 nm ( $\varepsilon_r = 6.8$ ) which corresponds to a capacitance value of  $C_E = 30$ pF.  $C_E$  will act as a short circuit at the desired oscillation frequency if

$$\frac{1}{2\pi f_0 C_E} < 0.1 \,\Omega \tag{5.35}$$

The RTD is connected through a CPW line with length  $l_r$  to the decoupling capacitor  $C_E$ . Since the RF signal sees the decoupling capacitor as a short, the connection of  $l_r$  and  $C_E$  can be approximated by an inductor L for the RF signal and its inductance value is determined by  $l_r$ . The inductor L is designed to resonated with the RTD self-capacitance  $C_n$  to obtain the desired frequency of oscillation. The frequency of oscillation is given by (5.26).

An on-chip or external capacitor  $C_{block}$  is introduced in the circuit to prevent any DC from reaching the load resistance  $R_L$  and is chosen as to satisfy (5.35).



**Figure 5.9:** Conventional single RTD device oscillator topology where  $V_{bias}$  is the biasing voltage,  $R_s$  and  $L_s$  represent the bias cable inductance and resistance, respectively. Shunt resistance  $R_B$  is used to suppress the low frequency bias oscillations.  $C_E$  is the decoupling capacitor and L the resonating inductor.

Fig. 5.10 shows the small signal RF equivalent circuit of the single RTD device oscillator where the RTD is represented by its self-capacitance  $C_n$  and negative differential conductance  $-G_n$ . From the de-embedding procedure discussed previously in chapter 2 the RTD self-capacitance is in the 0.8 - 1.5 fF/ $\mu$ m<sup>2</sup> range. The 3D illustration of the design described in Fig. 5.9 is shown in Fig. 5.11.



*Figure 5.10:* Small-signal equivalent circuit where the RTD is represented by the self-capacitance  $C_n$  in parallel with the negative conductance  $-G_n$ .



*Figure 5.11:* 3D illustration of a single RTD device oscillator. The shunt resistance  $R_B$  is fabricated from two parts placed on either side of the DC line.

Single RTD oscillators with two different device sizes 4  $\mu$ m × 4  $\mu$ m, and 5  $\mu$ m ×  $\mu$ m were designed and fabricated using the single oscillator topology, but with different CPW lengths. The 50  $\Omega$  characteristic impedance CPW line structure consists of a 60  $\mu$ m wide signal line and 40  $\mu$ m gap to the ground planes. The device sizing has been done according to [6].

For a single 5 µm × 5 µm device with a CPW length of 80 µm, the measured oscillation frequency was 27.84 GHz with a measured oscillation power of -2.8 dBm (0.52 mW) when the bias voltage was  $V_{bias} = 0.84$  V and  $I_{bias} = 129$  mA. For this measurement setup, the insertion loss of the probe, cables and dc block were measured using a reference 30 GHz signal source with the output power set at 0 dBm. To measure the insertion loss, the signal from the reference source was first measured using the spectrum analyzer, then again through the biasing lines and probes placed on a 50  $\Omega$ characteristic impedance CPW line. The insertion loss at 30 GHz was measured to be 5.4 dB and the corrected result for the oscillator output power is 2.6 dBm (1.8 mW). The stabilising resistor  $R_B$  ( $R_B = 8\Omega$ ) has a total  $I_{Rb} = 105$  mA current flowing across with 93 mW power dissipated. The current flowing through the diode was  $I_{bias} - I_{Rb} =$ 24 mA. The total DC power consumed by the oscillator is 108 mW, and the DC-to-RF conversion efficiency is 1.66 %. The measured single side-band phase noise of a 30 GHz RTD oscillator from 100 kHz to 100 MHz offset frequency is shown in Fig. 5.12. The phase noise is -92 dBc/Hz at 100 kHz offset, -100 dBc/Hz at 1 MHz offset and -137 dBc/Hz at 10 MHz offset from the carrier. The calculated FOM of the fabricated oscillator is -168.7 dBc/Hz.



*Figure 5.12: Measured single side-band phase noise of 30 GHz single RTD oscillator at 10 kHz offset from carrier to 100 MHz offset from carrier.* 

Using the parameters extracted for a 5  $\mu$ m × 5  $\mu$ m RTD device from layer structure #1 discussed previously in chapter 2, a model of the conventional RTD oscillator (Fig. 5.9) was simulated in MATLAB software. The measured and simulated spectrum of the RTD oscillator is presented in Fig. 5.13 at a bias level of 0.84 V.



*Figure 5.13:* Simulated and measured spectrum of RTD oscillator. The output power of the simulated RTD is about 3 mW with an oscillation frequency of 30.9 *GHz.* 

The calculated frequency of oscillation using (5.25) was 30.6 GHz, the simulated frequency is 30.9 GHz and the measured frequency 27.8 GHz. The output power of the simulated RTD oscillator (3 mW) is about half of to the theoretical maximum oscillation power (5.6 mW) given by (5.12). The model described in Chapter 4 is able to reproduce both the DC and RF characteristics of the fabricated devices.

The RF characteristics of the fabricated of the 4  $\mu$ m × 4  $\mu$ m and 5  $\mu$ m ×  $\mu$ m devices are summarized in Table 5.1. The average frequency tuning range ( $\Delta f$ ) of the fabricated RTD oscillators is about 6 GHz, with tuning sensitivity  $\Delta f / \Delta V_{bias} = 10$  GHz / V. The output power fluctuates by about 5 dB to 7 dB over the tuning range.

RTD size (µm)	Bias Voltage (V)	CPW length (µm)	Measured frequency (GHz)	Corrected output power (mW)	DC-to-RF efficiency (%)
4×4 μm <sup>2</sup>	1.0	300	23.4	0.71	0.47
	0.94	180	25	0.74	0.55
	0.8	150	27.5	0.70	0.70
	1.2	90	34.16	1.3	0.62
5×5 μm <sup>2</sup>	0.90	240	17.4	1.54	1.25
	1.05	180	21.3	1.73	1.06
	0.93	120	24.0	2.34	1.79
	0.84	80	27.8	1.80	1.66
Simulated	0.84	N/A	30.9	3.0	2.7

Table 5.1: Measurement results of the single RTD oscillators

# 5.3.2 High-Efficiency RTD Oscillators

As the efficiency of the standard single RTD oscillator topology described in the previous section is limited, mostly by the high power dissipated by the shunt resistance  $R_B$ , the oscillator circuit topology utilizing the RC stabilisation network was proposed as shown in Fig. 5.14 [13]. The value of the stabilising resistance  $R_B$  is calculated the same as in the previous section using (5.32). The value of capacitor  $C_B = 144$  pH is calculated using Eq. 5.36, where the parasitic inductance  $L_S = 0.9$  nH and parasitic resistance  $R_S = 2 \Omega$  where assumed to be dominated by the bias-tee connector. For a 4  $\mu$ m RTD the maximum negative differential conductance was measured at  $G_n = -70$  mS.

$$C_B > \frac{L_S G_n}{R_S} \tag{5.36}$$

At low parasitic oscillation frequencies (low MHz range) the resistor  $R_B$  can be ignored, and the capacitance  $C_E$  adds to the capacitance  $C_B$ . At higher parasitic oscillation frequencies  $C_B$  will appear as a short for the parasitic oscillations and  $R_B$  is used to stabilise the device. No RF power generated by the RTD is dissipated across the stabilising network as the capacitance  $C_E$  is designed to be a short at the designed oscillation frequency.



*Figure 5.14: Proposed RTD device oscillator topology with RC stabilisation network.* 

Oscillators with the RC stabilisation network topology where fabricated with a device size of 4  $\mu$ m × 4  $\mu$ m. The value for the inductance *L* was 140 pH for ~30 GHz oscillators. The choice of the oscillation frequency and device size was to facilitate easier circuit realisation and device characterisation. A micrograph of the fabricated oscillator circuit is shown in Fig. 5.15. The overall size of the oscillator circuit was around 1000×700  $\mu$ m<sup>2</sup>.


**Figure 5.15:** Micrograph of the fabricated RTD oscillator with an integrated stabilising network. For measurement, a GSG probe is used. The capacitor  $C_B$  and resistance  $R_B$  are split in two and placed in parallel with the RTD. The capacitor  $C_E$  acts as a short to ground for the RF signal.

For the high efficiency RTD oscillator the measured frequency of oscillation was 34.1 GHz at  $V_{bias} = 0.94$  V and  $I_{bias} = 18$  mA. The corrected output power was 3.95 dBm (2.48 mW), which compares well with the theoretical 5.18 dBm (3.3 mW) estimated from the 4  $\mu$ m × 4  $\mu$ m dc characteristics of the device. The measured oscillator spectrum is shown in Fig. 5.16 and Fig. 5.17 shows the measured oscillator output power and frequency as a function of bias voltage.



*Figure 5.16: Measured high efficiency RTD spectrum at*  $V_{bias} = 0.94$  V and  $I_{bias} = 18$  mA. The frequency of oscillation is 34.1GHz with an output power of 3.95 dBm.



**Figure 5.17:** Measured oscillator output power and frequency as a function of  $V_{bias}$ .

The RTD oscillator has a tuning range of about 7 GHz, with a tuning sensitivity of  $\Delta f$  /  $\Delta V_{bias} = 11.7$  GHz / V and high output power (1 to 2 dBm) across most of the operating voltage range. The calculated DC-to-RF conversion efficiency at  $V_{bias} = 0.94$  V is 14.7%, over a 10-fold improvement when compared to the conventional design and close to the theoretical 19.5% maximum efficiency for this epi-layer structure design and size. The measured phase noise of the high efficiency RTD oscillator is shown in Fig. 5.18. The phase noise is -89 dBc/Hz at 100 kHz offset, -137 dBc/Hz at 1 MHz offset and -133 dBc/Hz at 10 MHz offset from the carrier. The calculated FOM for the high efficiency RTD oscillator is -215 dBc/Hz.



*Figure 5.18: Measured phase noise for the high efficiency RTD oscillator. The phase noise is -89 dBc/Hz at 100 kHz offset, -137 dBc/Hz at 1 MHz offset and -133 dBc/Hz at 10 MHz offset from the carrier.* 

#### 5.3.3 Discussion

RTD oscillators with two different circuit topologies that produce a high oscillation power in the Ka-band (26.5 GHz – 40 GHz) have been described in this chapter. The oscillators were fabricated using photolithography process and employed different CPW structures in order to realize the MIM capacitors and inductance values.

Conventional RTD oscillators use the shunt-resistor method for bias oscillation suppression and, as a result, the efficiency of such oscillators is severely degraded. The proposed method does not hinder the oscillator performance and can be adapted to higher PVCR (peak-to-valley current ratio) epitaxial structures, thereby increasing the oscillator output power and conversion efficiency. A comparison between the fabricated oscillators and other published work is summarized in Table 5.2. To the best of author's knowledge, the fabricated high efficiency RTDs have the highest FOM between the different fabrication technologies.

Reference	[16]	[15]	[18]	[14]	[17]	This work	This work
TECH.	0.15 um pHEMT	130 nm SiGe BiCMOS	65-nm CMOS	65-nm CMOS	InGaAs /AlAs RTD	InGaAs /AlAs RTD	InGaAs /AlAs <b>RTD</b>
Fosc (GHz)	28.3	30.9	35.1	28.3	28.7	27.8	34.1
TR (%)	13.4	16.8	24.4	17.8	3.48	10.9	10.7
P <sub>DC</sub> (mW)	80	40.2	4.8	4.1	289.1	108	16.92
P <sub>RF</sub> (mW)	15	2.5	0.06	0.15	0.26	1.8	2.48
DC-to-RF Efficiency (%)	18.7	6.21	1.25	3.65	0.08	1.6	14.6
PN (dBc/Hz) @ 1MHz	-102	-101.4	-106.6	-102.8	-114	-100	-137
FOM (dBc/Hz)	-172.0	-175.1	-187.2	-185.7	-178.3	-168.7	-215
FOM <sub>T</sub> (dBc/Hz)	-174.5	-179.6	-194.9	-190.7	-168.9	-169.4	-215.5
FOM <sub>TP</sub> (dBc/Hz)	-186.3	-183.6	-207.1	-198.9	-163.0	-171.9	-219.4

 Table 5.2:
 Comparison between different VCOs in the Ka-band

## **5.4 Summary**

In this chapter, a new biasing network for RTD oscillators for high DC-to-RF conversion efficiency was presented. Highly efficient mm-wave RTD oscillators with tunable frequency between 29 and 36 GHz were fabricated and measured. The employed bias stabilisation network does not consume dc power resulting in over a 10-fold improvement in the DC-to-RF conversion efficiency when compared with the standard shunt resistance method. Combined with the low phase noise of the oscillator, RTDs provide a simple, low cost solution for high-capacity wireless communication systems and other applications. Higher efficiencies are expected by a reduction in the peak voltage and of the valley current. The design can be scaled for other epitaxial layer structures with higher PVCR, thereby further increasing the output power and the conversion efficiency.

## References

- C. Kim and A. Brandli, "High-frequency high-power operation of tunnel diode," *IRE Transactions on Circuit Theory*, vol. 8, no. 4, pp. 416-425, 1961.
- [2] P. Nahin, The science of radio: with matlab and electronics workbench demonstrations, New York: Springer-Verlag, 2001.
- [3] M. Reddy, Schottky-collector resonant tunnel diodes for sub-millimeterwave, PhD Thesis: University of California Santa Barbara, 1997.
- [4] W. F. Chow, Principles of tunnel diode circuits, Malabar, USA: Krieger Pub Co, 1964.
- [5] L. Wang, "Output Power Analysis and Simulations of Resonant Tunneling Diode Based Oscillators," in *Communications in Computer and Information Science*, Shanghai, China, Springer, 2012, pp. 47-55.
- [6] L. Wang, Reliable design of tunnel diode and resonant, PhD thesis: University of Glasgow, 2012.
- [7] Hewlett-Packard (HP), "Phase noise characterisation of microwave oscillators," HP Product Note 11729C-2, Washington, USA, 1985.
- [8] D. Scherer, "The 'art' of phase noise measurements," in *HP RF Microwave Symposium*, 1984.
- [9] T. H. Lee and A. Hajimiri, "Oscillator phase noise: a tutorial," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 3, pp. 326-336, 2000.
- [10] E. Rubiola, Phase Noise and Frequency Stability in Oscillators, Cambridge: Cambridge University Press, 2010.
- [11] Y.-C. L. Lin, M.-L. Yeh and C.-C. Chang , "A high figure-of-merit low phase noise 15-GHz CMOS VCO," *Journal of Marine Science and Technology*, vol. 21, no. 1, pp. 82-86, 2013.

- [12] J. Wang, L. Wang, C. Li, B. Romeira and E. Wasige, "28 GHZ MMIC resonant tunneling diode oscillator of around 1mW output power," *Electronics Letters*, vol. 49, no. 13, pp. 816-818, 2013.
- [13] A. C. Cornescu, R. Morariu, A. Ofiare, A. Al-Khalidi, J. Wang and E. Wasige, "High-Efficiency Bias Stabilization for Resonant Tunneling Diode Oscillators," *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 8, pp. 3449-3454, 2019.
- [14] T. Ding, X. Fan and D. Zhao, "Ka-band wideband VCO with LC filtering technique in 65-nm CMOS," *Electronics Letters*, vol. 55, no. 10, pp. 581-583, 2019.
- [15] Y. F. Meng, B. M. Frank and A. El-Gabaly, "A novel variable inductor-based differential Colpitts VCO design with 17% frequency tuning range for 30 and 60 GHz applications," in *IEEE MTT-S International Microwave Symposium*, Tampa, FL, 2014.
- [16] B. Piernas, K. Nishikawa, T. Nakagawa and K. Araki, "A compact and lowphase-noise Ka-band pHEMT-based VCO," *IEEE Transactions on Microwave Theory and Techniques*, vol. 51, no. 3, pp. 778-783, 2003.
- [17] J. Wang, L. Wang, C. Li, B. Romeira and E. Wasige, "28 GHz MMIC resonant tunnelling diode oscillator of around 1mW output power," *Electronics Letters*, vol. 49, no. 13, pp. 816-818, 2013.
- [18] J. Baylon, P. Agarwal, L. Renaud, S. N. Ali and D. Heo, "A Ka-band dual-band digitally controlled oscillator with -195.1 dBc/Hz FoMt based on a compact high-Q dual-path phase-switched inductor," *IEEE Transactions on Microwave Theory and Techniques*, vol. 67, no. 7, pp. 2748-2758, 2019.

# Chapter 6. Wireless Data Transmission Link

## 6.1 Introduction

Chip-sized and room-temperature devices that can both emit and detect millimeter/submillimeter wave radiation are key components for future mobile communication applications [1]-[4]. Recently, high power RTD transmitters have been reported operating in the J-band (220 GHz - 325 GHz) frequency range and with approximately 1 mW output power [5][6]. As data transmitters RTDs are easily modulated by a signal superimposed on the bias lines, and also possess a wide modulation bandwidth [6][7]. On the detector side, most commonly used are Schottkybarrier diodes (SBDs), due to their small-form factor and high-speed response at room temperature [8]-[10], however SBDs are strongly temperature dependent due to thermionic emission of charge carriers over the potential barrier [11][12]. RTD based detectors have recently gained research momentum [1][13]-[15] due to their strong non-linear *I-V* characteristics and the possibility of integrating the transmitter and receiver on the same circuit, thereby lowering the implementation cost and accelerating adoption into the market. When compared to SBDs, the RTD detectors were able to match and/or surpass the performance of their commercially available counterparts [13][16].

In this chapter, wireless links experiments were done jointly with another PhD student (Razvan Morariu) to assess the feasibility of future RTD transceiver designs. The RTD detectors where designed and developed as part of Razvan Morariu's PhD research project, while the RTD transmitters were designed and developed as part of this project. The full characteristics of the RTD detector design will be covered in his thesis, which will be published in the future. The (joint) result of the wireless transmission experiments is reported here as a joint result with the consent of Razvan Morariu.

## 6.2 Experimental Setup

The experimental setup for wireless data transmission using two RTDs, one operating as a transmitter (Tx) and one as a receiver (Rx), is shown in Fig. 6.1. A pseudo-random binary sequence (PRBS) generator was connected through a bias-T to the Tx. The RTD was biased in the PDR region, close to the peak voltage, and the signal from the PRBRS was used to modulate the carrier using OOK (on-off keying). The signal was radiated into free space through a 26 GHz to 40 GHz horn antenna (15 dBi gain) connected to the RTD through a CPW-to-waveguide transition. An identical antenna fitted with a waveguide-to-CPW transition is used at the receiver side. The Tx and Rx RTD were separated by a distance of 15 cm for data communications (constrained by the limited space available in the measurement setup). The RTD receiver is biased from a second power supply through a bias-T. The demodulated RF signal is obtained from the DC side of the RTD Rx and is passed through a low noise amplifier (LNA) and into an oscilloscope/spectrum analyzer for measurements.



*Figure 6.1:* Schematic block diagram of wireless data transmission setup using *RTDs as transmitters and receivers.* 

#### 6.2.1 RTD Transmitter

Fig. 6.2 shows how the RTD transmitter is modulated. The RTD Tx is biased within the PDR at  $V_{off}$ , and the baseband (BB) signal is used to shift the operating bias point at  $V_{on}$ . The modulated signal is created by the on-off keying of the BB signal, in which a logic-one biases the RTD in the NDR to create oscillations and a logic-zero returns the RTD at the  $V_{off}$  position and the oscillations stop. The BB signal level has been chosen to move the RTD biasing point to where the power of the oscillator is at its highest. A similar modulation scheme has been tested where the RTD is biased within the NDR region and the BB signal is used to turn off the oscillations by moving the bias in the second positive differential region.



*Figure 6.2:* Operation of the RTD as a transmitter using OOK modulation scheme. The BB signal is used to move the operating point of the RTD (on/off).

The resulting signal changes with respect to the baseband modulating signal, where the amplitude of the high frequency carrier takes on the shape of the lower frequency modulating signal. An ideal carrier and baseband signal can be expressed mathematically as:

$$e_c = E_c \sin(\omega_c t) \tag{6.1}$$

$$e_m = E_m \sin\left(\omega_m t\right) \tag{6.2}$$

where  $E_c$  and  $E_m$  represent the amplitude of the carrier and baseband signal and  $\omega_c$  and  $\omega_m$  the angular frequency of the carrier and baseband signal, respectively. For the analysis, the BB signal was treated as a single sine wave, however the actual BB signal has a square-wave form which can be regarded as a sum of sine waves. The resulting signal can be written in the form:

$$e_{am} = (E_c + E_m \sin(\omega_m t)) \sin(\omega_c t)$$
(6.3)

by normalizing the carrier amplitude to one the equation can be re-written as:

$$e_{am} = (1 + m\sin(\omega_m t))\sin(\omega_c t)$$
(6.4)

where *m* represents the modulation index and is defined as the ratio of the modulation signal amplitude to the carrier amplitude i.e.

$$m = \frac{E_m}{E_c} ; 0 \le m \le 1 \tag{6.5}$$



*Figure 6.3:* Amplitude modulation of a carrier signal by a much lower frequency modulating signal at 0.5 and 1 modulation index value. For OOK modulation the modulation index is 1 (100 %).

by expanding Eq. 6.4 we obtain:

$$e_{am} = \sin \omega_c t + \frac{m}{2} \cos(\omega_c - \omega_m) t - \frac{m}{2} \cos(\omega_c + \omega_m) t \quad (6.6)$$

where  $\sin \omega_c t$  represents the carrier frequency,  $\frac{m}{2}\cos(\omega_c - \omega_m) t$  represents the lower sideband and  $\frac{m}{2}\cos(\omega_c + \omega_m) t$  is the upper sideband. For an OOK modulation scheme the modulation index *m* is 1, as such the sideband amplitudes is half of the carrier amplitude, i.e. a quarter of the power.

The RF spectrum of an RTD oscillator modulated with a 1-0-1-0 pattern at 2 GHz clock frequency is shown in Fig. 6.4. The RTD transmitter has an oscillation frequency of approximately 30 GHz with -1.6 dBm (0.7 mW) corrected output power. The output of the RTD transmitter was fed to a CPW-to-waveguide transition and into the antenna.



*Figure 6.4: RF* spectrum of transmitter *RTD* when biased in the *NDR* and the *BB* signal was used to move the biasing point in the second *PDR* region. The modulating sidebands can be seen at a 1 GHz distance from the carrier (1-0-1-0 pattern @ 2 GHz clock frequency)

### 6.2.2 RTD Receiver

The measured RF power at the receiver antenna with a distance between the antennas set to 15 cm was approximately -30 dBm (1  $\mu$ W). The modulated signal was then fed through a waveguide-to-CPW transition to the RTD receiver, which operated in the square-law region as shown in Fig. 6.5.



*Figure 6.5:* Operating point of an RTD receiver. The non-linear characteristics of the IV are shown in the inset.

The current-voltage relationship of the diode detector when biased in the non-linear region is of the form:

$$i(t) = av^{2}(t) + bv(t)$$
 (6.7)

where the coefficients a and b are constants associated with the detector. The output of the system varies with respect to the squared input voltage. By applying the modulated signal to the squaring device, we get:

$$i(t) = ae_{am}(t)^2 + be_{am}(t)$$
 (6.8)

For the analysis the second term of Eq. 6.8 can be ignored, and by expanding the input signal the resulting equation is shown in (6.9)

$$i(t) = a \left( \sin \omega_c t + \frac{m}{2} \cos(\omega_c - \omega_m) t - \frac{m}{2} \cos(\omega_c + \omega_m) t \right)^2 \quad (6.9)$$

The output of the detector is the square of the amplitude modulated signal and since the input is being multiplied by the  $\sin \omega_c t$  term, two of the resulting terms will be the original modulation signal i.e.

$$e_{am} \times \sin\omega_{c} t = \underbrace{\frac{m}{2} \sin\omega_{m} t}_{\text{original BB signal}} - \underbrace{\frac{1}{2} \cos 2\omega_{c} t + \frac{m}{4} [\sin(2\omega_{c} - \omega_{m})t - \sin(2\omega_{c} + \omega_{m})t]}_{\text{AM signal centred at } 2\omega_{c}}$$
(6.10)

## **6.3 Measurement Results**

The output of the RTD receiver was connected two a 20-dB gain (0 – 3 GHz) LNA in order to enhance the demodulated signal before connecting it to the oscilloscope. The measured eye diagrams for a 1 Gb/s and a 2 Gb/s data transfer speed is shown in Fig. 6.6. The complete measurement setup is shown in Fig. 6.7. For a 1 Gb/s data transfer the measured signal was 7.2 mV  $V_{pp}$  with a signal to noise ratio of approximately 15. For the 2 Gb/s data transfer the signal measured 2.2 mV  $V_{pp}$  and a signal to noise ratio of 5. Error-free transmission measured with a bit error rate tester (BERT) was achieved from 1 Gb/s to 2.4 Gb/s.

Beyond 2.4 Gb/s the BER increased ( $5.8 \times 10^{-6}$  at 2.6 Gb/s), largely due to the 3-dB bandwidth of the RTD baseband circuit. For higher data rates the RC time constant at the input of the RTD transmitter must be taken into consideration [15][17]. The time constant  $\tau$  is the product of the circuit resistance and the circuit capacitance, i.e.  $\tau = RC$ . In the RTD oscillator design, a shunt capacitor (60 pF) is used to create the shorted-CPW inductance. The RC delay time associated with this system is 60 picoseconds ( $C_E = 60$  pF,  $R_S = 1 \Omega$  approximated), thus the data rate limited by this constraint is 2.6 Gb/s.



**Figure 6.6:** Eye diagrams of RTD wireless link (a) 1 Gbit/s data rate,  $7.2mVV_{pp}$ , S/N = 15 (b) 2 Gbit/s data rate,  $2.2 mVV_{pp}$ , S/N = 5.



*Figure 6.7: Picture of the measurement setup used in the wireless data transmission link.* 

## 6.4 Summary

In this chapter a joint experiment of an RTD transmitter and an RTD receiver has been presented including an explanation of the operating principle of the modulation / demodulation process. Wireless, error-free data transmission has been achieved at a distance of 15 cm and 2 Gb/s data transfer rate with a respectable signal-to-noise ration. Further improvements can be made in both transmitter and receiver by increasing the 3dB-bandwith point of the RTD transmitter and by selecting an appropriate RTD which has a higher curvature component  $\left(\frac{d^2I}{dV^2} / \frac{dI}{dV}\right)$  in the dc characteristics [18]. This experiment has shown that RTD transceivers are feasible for next generation low-cost wireless transmission.

## References

- K. Arzi, A. Rennings, D. Erni, N. Weimann, W. Prost, S. Suzuki and M. Asada, "Millimeter-wave signal generation and detection via the same triple barrier RTD and on-chip antenna," in 2018 First International Workshop on Mobile Terahertz Systems (IWMTS), Duisburg, Germany, 2018.
- M. R. Palatella, "Internet of Things in the 5G era: enablers, architecture, and business models," *IEEE Journal on Selected Areas in Communication*, vol. 34, no. 3, pp. 510-527, 2016.
- [3] J. Webber, H. Kamoda, N. Kukutsu and T. Kumagai, "Millimeter-wave wireless communication in a data center cabinet with adaptive control of propagation," in 21st Asia-Pacific Conference on Communications (APCC), Kyoto, Japan, 2015.
- [4] S. Rommel, T. Raddo, U. Johannsen, C. Okonkwo and I. Monroy, "Beyond 5G
   wireless data center connectivity," in *Proceedings Volume 10945, Broadband Access Communication technologies XIII*, San Francisco, USA, 2019.
- [5] J. Wang, A. Al-khalidi, K. Alharbi, A. Ofiare, E. Wasige and J. Figueiredo,
  "High performance resonant tunneling diode oscillators as terahertz sources," in
  46th European Microwave Conference (EuMC), London, UK, 2016.
- [6] A. Al-Khalidi, J. Wang and E. Wasige, "Compact J-band oscillators with 1 mW RF output power and over 110 GHz modulation bandwidth," in 3rd International Conference on Infrared, Milimeter and Terahertz Waves (IRMWW-THz 2018), Nagoya, Japan, 2018.
- [7] S. Suzuki and M. Asada, "Terahertz communications using resonant-tunnelingdiode oscillators," in 11th European Conference on Antennas and Propagation (EUCAP), Paris, France, 2017.
- [8] J. Wang, A. Al-Khalidi, A. Cornescu, R. Morariu, A. Ofiare and E. Wasige, "Design, fabrication and characterisation of RTD terahertz oscillators," in

*European Microwave Conference in Central Europe (EuMCE)*, Prague, Czech Republic, 2019.

- [9] K. Karashima, M. Shiraishi, K. Hinata, S. Suzuki and M. Asada, "Heterodyne detection of output of sub-THz RTD oscillator using InP-SBD detector and RTD local oscillator," in 35th International Conference on Infrared, Millimeter, and Terahertz Waves, Rome, Italy, 2010.
- [10] N. Oshima, K. Hashimoto, S. Suzuki and M. Asada, "Terahertz wireless data transmission with frequency and polarization division multiplexing using resonant-tunneling-diode oscillators," *IEEE Transactions on Terahertz Science* and Technology, vol. 7, no. 5, pp. 593-598, 2017.
- [11] A. K. Kerr and Y. Anand, "Schottky diode MM detectors with improved sensitivity and dynamic range," *Microwave Journal*, vol. 24, pp. 67-71, 1981.
- [12] R. Sharma, "Temperature dependence of I-V characteristics of Au/n-Si Schottky barrier diode," *Journal of Electron Devices*, vol. 8, pp. 286-292, 2010.
- [13] T. Shiode, T. Mukai, M. Kawamura and T. Nagatsuma, "Giga-bit wireless communication at 300 GHz using resonant tunneling diode detector," in *Asia-Pacific Microwave Conference*, Melbourne, Australia, 2011.
- [14] N. Nishigami, Y. Nishida, S. Diebold, J. Kim, M. Fujita and T. Nagatsuma, "Resonant tunneling diode receiver for coherent terahertz wireless communication," in *Asia-Pacific Microwave Conference (APMC)*, Kyoto, Japan, 2018.
- [15] J. Webber, N. Nishigami, J. Kim, M. Fujita and T. Nagatsuma, "Terahertz wireless communications using resonant tunnelling diodes with radio-overfibre," *Electronics Letters*, vol. 55, no. 17, pp. 949-951, 2019.
- [16] Y. Takida, S. Suzuki, M. Asada and H. Minamide, "Sensitivity measurment of resonant tunneling diode terahertz detectors," in 44th International Conference on Infrared, Millimeter, and Terahertz Waves (IRMMW-THz), Paris, France, 2019.

- [17] J. Park, J. Lee, K. Lee and K. Yang, "A 0.71-pJ/b on-off keying Ka-band oscillator using an InP-based resonant tunneling diode," *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 7, pp. 660-662, 2017.
- [18] P. Fay, W. Li, S. Rahman, Z. Jiang and L. Liu, "Tunneling-based heterostructure devices for millimeter-wave and THz sensing," in *IEEE National Aerospace and Electronics Conference (NAECON) and Ohio Innovation Summit (OIS)*, Dayton, USA, 2016.

# Chapter 7. Conclusions and Future Work

## 7.1 Conclusions

Devices that exhibit negative differential resistance (NDR) such as tunnel diodes, IMPATT diodes, Gunn diodes and resonant tunneling diodes can be used as electronic sources for high frequency signals. Among them, the RTD is the fastest operating solid-state electronic device with a reported frequency of oscillation up till now of 1.98 THz [1], and with a theoretical bandwidth of 2.5 THz [2]. However, the biggest limitations of an RTD oscillator are its low output power and low DC-to-RF conversion efficiency, due to the presence of parasitic oscillations when biased in the NDR. The bias oscillations associated with the device makes it difficult to accurately determine the DC characteristics of the device and therefore, results in a lack of accurate models which could help in the optimization process of various applications. The purpose of this PhD project was to address these limitations and focused on the stability criteria of RTDs and their application in high frequency oscillator circuits.

The results achieved in this project are as follows:

- i. Design equations for realizing a stable RTD biasing network were derived. The method enables direct measurement of the RTD's *I-V* characteristics, in contrast with the conventional shunt resistor stabilisation method which the *I-V* characteristics had to be de-embedded after the measurement. The proposed method eliminates both low frequency bias oscillations as well as high frequency oscillations which would otherwise distort the measured characteristics. Furthermore, the method described can be adapted for oscillator designs, thereby improving the DC-to-RF conversion efficiency by reducing the DC current consumption.
- Radio frequency (RF) characterization of an RTD device and DC characterization were used in the realization on an accurate large-signal model of the RTD. The S-parameter measurements were used to determine an

accurate small-signal model of the device, which when combined with the large-signal *I-V* characteristics was able to reproduce the DC measurement results in a non-stable RTD device (plateau region) and the RF characteristics when used in an oscillator circuit. The model is expected to serve in the optimization process of RTD devices for millimeter and submillimeter wave applications.

- iii. MMIC RTD oscillators using the proposed stabilisation method were designed and fabricated. Highly efficient millimeter wave RTD oscillators with over 7 GHz tuning frequency and over a milliwatt output power were presented. The employed bias stabilisation network did not consume any DC power, resulting in over a 10-fold improvement in DC-to-RF conversion efficiency when compared to the conventional designs and close to the theoretical maximum of the employed epi-layer structure. The presented approach could have a major impact with regards to the adoption of RTD technology for portable devices where battery capacity is at a premium.
- iv. A wireless data transmission link using two RTDs operating as transmitter and receiver was demonstrated, proving the feasibility of future RTD transceiver designs, where the same RTD chip can be used in a half-duplex system to both transmit and receive, thereby lowering the implementation cost and accelerating the adoption of this technology into the market.

## 7.2 Future Work

## 7.2.1 RTD Modulation Bandwidth

In this thesis a millimeter-wave communication system using two RTDs has been demonstrated and has the potential to play a major role in the next generation of communication devices. However, the RTD oscillators in this thesis were not optimized for high data transfer rates, as such the bandwidth was limited to approximately 2.6 Gb/s. A large decoupling capacitor in the bias network results in a slow response as the capacitor charges/discharges when a data signal is applied to it. The modulation bandwidth is therefore limited by the biasing network and decoupling capacitor  $C_E$ , which form a low pass filter. An alternative method would be to modulate the signal on the RF side of the oscillator, where a balun / lange coupler paired with

PIN diodes can be used to perform binary phase shift keying modulation (BPSK) [3][4], therefore removing the constraint imposed on the biasing network. A layout design of a CPW balun is shown in Fig. 7.1 and a schematic diagram of a BPSK modulator is shown in Fig. 7.2.



*Figure 7.1:* Layout of CPW balun, the phase difference between the output at port 2 and 3 is 180°. Adapted from [3].



*Figure 7.2:* Schematic block diagram of a BPSK modulator. The switches represent the PIN diodes, which are used to turn on/off the carrier input. Adapted from [4].

In summary, high performance and highly efficient RTD oscillators were demonstrated in this thesis. The oscillators could form compact, room temperature operating building blocks for THz system in different mobile short-range communication and imaging systems.

## References

- [1] R. Izumi, S. Suzuki and M. Asada, "1.98 THz resonant-tunneling-diode oscillator with reduced conduction loss by thick antenna electrode," in 42nd International Conference on Infrared, Millimeter, and Terahertz Waves (IRMMW-THz), Cancun, Mexico, 2017.
- T. C. Sollner, W. D. Goodgue, P. E. Tannenwald, C. D. Parker and D. D. Peck,
   "Resonant tunneling through quantum wells at frequencies up to 2.5 THz,"
   *Applied Physics Letters*, vol. 43, no. 6, pp. 588-590, 1983.
- [3] G.-Y. Chen and J.-S. Sun, "Enhanced bandwidth for a coplanar coupled-line balun design," Microwave and Optical Technology Letters, vol. 49, no. 12, pp. 2902-2905, 2007.
- [4] B. R. Jackson, Y. Zheng and C. E. Saavedra, "A CMOS direct-digital BPSK modulator using an active balun and common-gate switches," in IEEE International Symposium on Circuits and Systems, New Orleans, USA, 2007.

## **Appendix A. Fabrication Process**

#### Sample cleaning

- 1. Ultrasonic bath in acetone for 3 minutes
- 2. Ultrasonic bath in methanol for 3 minutes
- 3. Ultrasonic bath in IPA for 3 minutes
- 4. Rinse with de-ionized water (RO-water)

#### **Collector contact metal**

- 1. Spin S1805 at 4000 RPM for 30 s
- 2. Bake sample on hotplate at  $115^{\circ}$  C for 60 s
- 3. Submerge in chlorobenzene for 5 minutes
- 4. Expose using MA6 for 2.4 s / hard contact / 40  $\mu$ m separation
- Develop sample in solution 1:1 Microposit developer concentrate (MDC): H<sub>2</sub>O for 50 s
- 6. Rinse in RO-water for 60 s
- 7. Blow dry with N<sub>2</sub>
- 8. Plasma ash the sample at 120 W RF power for 2 minutes
- Submerge sample in solution 1:10 ammonium hydroxide (NH<sub>4</sub>OH): H<sub>2</sub>O for 20 s
- 10. Rinse in RO-water for 60s
- 11. Blow dry with  $N_2$
- 12. Argon gun for 30 s (in Plassys 4)
- 13. Deposit Ohmic contact Ti/Pd/Au 20/30/150 nm
- 14. Lift-off after submerging in acetone/1165 resist remover at 50° C for 60 minutes
- 15. Rinse in Ro-water for 60 s
- 16. Blow dry with  $N_2$

#### Etch to the bottom InGaAs contact layer

- 1. Spin S1805 at 4000 RPM for 30 s
- 2. Bake sample on hotplate at  $115^{\circ}$  C for 60 s
- 3. Expose using MA6 for 2.4 s / hard contact / 40  $\mu$ m separation
- 4. Develop sample in solution 1:1 MDC: H2O for 50 s
- 5. Rinse in RO-water for 60 s
- 6. Blow dry with  $N_2$
- 7. Plasma ash the sample at 120 W RF power for 2 minutes
- Etch using solution orthophosphoric acid (H<sub>3</sub>PO<sub>4</sub>): hydrogen peroxide (H<sub>2</sub>O<sub>2</sub>): H<sub>2</sub>O 1:1:38, etch rate 100 nm / min
- 9. Rinse in RO-water for 60 s
- 10. Measure etch depth
- 11. Submerge in acetone/1165 resist remover at 50° C for 30 minutes

#### Emitter contact metal

1. Same as collector contact metal

#### Etch to InP substrate

- 1. Spin S1805 at 4000 RPM for 30 s
- 2. Bake sample on hotplate at 115° C for 60 s
- 3. Expose using MA6 for 2.4 s / hard contact / 40  $\mu$ m separation
- 4. Develop sample in solution 1:1 MDC: H<sub>2</sub>O for 50 s
- 5. Rinse in RO-water for 60 s
- 6. Blow dry with  $N_2$
- 7. Plasma ash the sample at 120 W RF power for 2 minutes
- 8. Etch using solution  $H_3PO_4$ :  $H_2O_2$ :  $H_2O$  1:1:38, etch rate 100 nm / min
- Etch into InP substrate using solution hydrochloric acid (HCl): H<sub>3</sub>PO<sub>4</sub> 1:4 for 30 s
- 10. Rinse in RO-water for 60 s
- 11. Measure etch depth
- 12. Submerge in acetone/1165 resist remover at 50° C for 30 minutes

### Passivation and VIA opening

- 1. Spin polyimide PI-2545 at 8000 RPM for 30 s
- 2. Bake sample on hotplate at 115° C for 2 minutes
- 3. Bake sample in oven at 180° C for a minimum of 6 hours
- 4. Spin S1805 at 1500 RPM for 30 s
- 5. Bake sample on hotplate at 115° C for 2 minutes
- 6. Expose using MA6 for 2.8 s / hard contact / 40  $\mu$ m separation
- 7. Develop sample in solution 1:1 MDC: H<sub>2</sub>O for 50 s
- 8. Bake sample on hotplate at 115° C for 10 minutes
- Dry etch using tetrafluoromethane (CF<sub>4</sub>): O<sub>2</sub> 5: 20 sccm, 200 W RF power until interferometer shows straight line (approximately ~ 6 minutes)
- 10. Submerge in acetone/1165 resist remover at 50° C for 30 minutes

#### Thin-film resistor

- 1. Spin S1805 at 4000 RPM for 30 s
- 2. Bake sample on hotplate at 115° C for 60 s
- 3. Submerge in chlorobenzene for 5 minutes
- 4. Expose using MA6 for 2.4 s / hard contact / 40 µm separation
- Develop sample in solution 1:1 Microposit developer concentrate (MDC): H<sub>2</sub>O for 50 s
- 6. Rinse in RO-water for 60 s
- 7. Blow dry with  $N_2$
- 8. Plasma ash the sample at 120 W RF power for 2 minutes
- 9. Etch into InP substrate using solution HCl: H<sub>3</sub>PO<sub>4</sub> 1:4 for 30 s
- 10. Bake sample on hotplate at  $115^{\circ}$  C for 60 s
- 11. Deposit resistor NiCr 33 nm thickness
- 12. Submerge in acetone/1165 resist remover at 50° C for 30 minutes

## 1st Bond pad

- 1. Spin LOR 10A at 6000 RPM for 30 s
- 2. Bake sample on hotplate at 150° C for 3 minutes
- 3. Spin S1805 at 4000 RPM for 30 s
- 4. Bake sample on hotplate at 115° C for 2 minutes
- 5. Expose using MA6 for 2.4 s / hard contact / 40  $\mu$ m separation
- 6. Develop sample in MF319 for 75 s
- 7. Plasma ash the sample at 120 W RF power for 2 minutes
- 8. Deposit bond pad Ti/Au 20/380 nm
- 9. Submerge in acetone/1165 resist remover at 50° C for 60 minutes

## Metal-insulator-metal capacitor

- 1. Spin LOR 3A at 3000 RPM for 30 s
- 2. Bake sample on hotplate at 150° C for 3 minutes
- 3. Spin S1805 at 4000 RPM for 30 s
- 4. Bake sample on hotplate at 115° C for 2 minutes
- 5. Expose using MA6 for 2.4 s / hard contact / 40  $\mu$ m separation
- 6. Develop sample in MF319 for 75 s
- 7. Plasma ash the sample at 120 W RF power for 2 minutes
- 8. Deposit silicon nitride (Si<sub>3</sub>N<sub>4</sub>) dielectric by ICP-CVD 75 nm
- 9. Submerge in acetone/1165 resist remover at 50° C for 60 minutes

### 2<sup>nd</sup> Bond pad

1. Same as  $1^{st}$  bond pad



*Figure A.1:* Fabrication process: (a) top contact metal deposition, (b) etch to bottom InGaAs layer, (c) bottom contact metal deposition, (d) etch to InP substrate, (e) deposit polyamide and open VIA for bond pad contact



**Figure A.2:** Fabrication process: (a) thin-film resistor and  $1^{st}$  bond pad metal deposition, (b) dielectric silicon nitride deposition (Si<sub>3</sub>N<sub>4</sub>), (c)  $2^{nd}$  bond pad metal deposition